ESSAYS ON THE INSTITUTIONAL DETERMINANTS OF PUBLIC EXPENDITURE

Inaugural-Dissertation zur Erlangung des Grades Doctor oeconomiae publicae (Dr. oec. publ.) an der Ludwig-Maximilians-Universität München

2008

vorgelegt von Andreas Kappeler

Referent: Prof. Dr. Andreas Haufler Korreferent: Prof. Dr. Thiess Büttner Promotionsabschlussberatung: 04.02.2009

Acknowledgements

Many people have directly and indirectly contributed to the completion of this thesis. I want to emphasize my gratitude for all the support and encouragement that I experienced throughout the past three years.

First of all, I would like to thank my supervisor, Professor Andreas Haufler, for his comprehensive supervision and a perfect combination of instructive criticism and encouragement. This thesis owes much to his sophisticated comments and suggestions.

During the past three years, I have also received a lot of support from my colleagues at the Munich Graduate School of Economics and the Economics Department at the University of Munich (LMU). In particular I would like to thank Thiess Büttner, Jin Cao, Andre Ebner, Gerhard Illing, Rainer Lanz, Nadine Riedel, Johannes Rinke, Marco Runkel, Christina Strassmeier, Anno Stolper, and Christian Traxler for many helpful discussions and hints.

Part of my thesis, in particular chapter 2, have been conducted during an internship at the European Investment Bank, Luxembourg. I would like to acknowledge the hospitality and the wonderful atmosphere at the EFS team of the EIB. I am especially grateful to Timo Valila for his supervision and encouragement, as well as to the entire EFS team for many fruitful discussions and insights.

Further, I also would like to thank the School of Economics of Fudan University, Shanghai, which hosted me for one month during my PhD studies. This time and the comprehensive supervision of Professor Chen Zhao were particularly helpful in setting up my research agenda. I could benefit from his suggestions throughout my PhD studies.

Also, financial support from the Deutsche Forschungsgemeinschaft (DFG) is gratefully acknowledged.

Finally, I want to thank my parents and family as well as my wife Lizhe for their incessant support throughout this period. I am forever indebted to you for your understanding, encouragement, and help!

Andreas Kappeler Munich, September 2008

Contents

1	Intr	roduction	1				
	1.1	Public Expenditure and the Institutional Setting of the Public Sector $% \mathcal{A}$.	1				
	1.2	Fiscal Federalism and Public Expenditure	2				
	1.3	Privatization and Public Expenditure	4				
	1.4	Policy Implications	5				
2	Fiscal Externalities in a Three-Tier Structure of Government						
	2.1	Introduction	8				
	2.2	Literature Review	11				
	2.3	Benchmark Case: Two Tiers of Government	12				
	2.4	Analysis with Three Tiers of Government	16				
		2.4.1 Non-Redistributive Funding of Matching Grants	20				
		2.4.2 Redistributive Funding of Matching Grants	22				
		2.4.3 High-Level Government without Tax Autonomy	24				
	2.5	Conclusion	28				
	2.6	Appendix	31				
3	The	e Composition of Public Investment and Fiscal Federalism: Panel					
	Dat	a Evidence From Europe	43				
	3.1	Introduction	43				
	3.2	Modelling the Composition of Public Spending	47				
		3.2.1 Two Forms of Competition for Private Capital	49				
		3.2.2 Composition of Public Investment	50				

	3.3	Compo	osition of Public Investment in Europe: Stylized Facts \ldots .	52
	3.4	Empir	ical Analysis	54
		3.4.1	Model Specification	54
		3.4.2	Sample Properties	55
		3.4.3	Methodology	56
		3.4.4	Basic Results	59
		3.4.5	Measuring Decentralization	62
		3.4.6	Identifying Regional Autonomy	65
		3.4.7	Results for Aggregated Types of Investment	67
		3.4.8	External vs. Internal Competition	68
		3.4.9	Methodological Robustness-Checks	71
	3.5	Conclu	usion	76
	3.6	Appen	dix	78
	3.7	Appen	dix	82
4	Priv	vatizati	on and Public Social Spending	34
4	Priv 4.1	v atizat i Introd	ion and Public Social Spending	34 84
4	Priv 4.1 4.2	v atizat i Introd Literat	ion and Public Social Spending	34 84 87
4	Priv 4.1 4.2 4.3	v atizat i Introd Literat Model	ion and Public Social Spending 8 uction	84 84 87 89
4	Priv 4.1 4.2 4.3	vatizati Introd Literat Model 4.3.1	aon and Public Social Spending 8 uction cure Review Firms	84 84 87 89 90
4	Priv 4.1 4.2 4.3	vatizati Introd Literat Model 4.3.1 4.3.2	ion and Public Social Spending a uction	84 84 87 89 90
4	Priv 4.1 4.2 4.3	vatizati Introd Literat Model 4.3.1 4.3.2 4.3.3	aon and Public Social Spending auction uction auction cure Review auction Firms auction Households auction State Sector auction	 84 84 87 89 90 91 92
4	Priv 4.1 4.2 4.3	vatizati Introd Literat Model 4.3.1 4.3.2 4.3.3 Result	ion and Public Social Spending auction uction	84 87 89 90 91 92 94
4	 Priv 4.1 4.2 4.3 4.4 4.5 	vatizati Introd Literat Model 4.3.1 4.3.2 4.3.3 Result Result	ion and Public Social Spending auction uction	 84 87 89 90 91 92 94 97
4	 Priv 4.1 4.2 4.3 4.4 4.5 4.6 	vatizati Introd Literat Model 4.3.1 4.3.2 4.3.3 Result Result Excess	ion and Public Social Spending auction uction	 84 84 87 89 90 91 92 94 97 00
4	 Priv 4.1 4.2 4.3 4.4 4.5 4.6 4.7 	vatizati Introd Literat Model 4.3.1 4.3.2 4.3.3 Result Result Excess Conclu	ion and Public Social Spending ion uction ion sure Review ion Firms ion Households ion State Sector ion without Lump-Sum Tax ion Burden of Taxation ion 1 1	 84 84 87 89 90 91 92 94 97 00 03
4	 Priv 4.1 4.2 4.3 4.4 4.5 4.6 4.7 4.8 	vatizati Introd Literat Model 4.3.1 4.3.2 4.3.3 Result Result Excess Conclu Appen	ion and Public Social Spending auction uction	 84 87 89 90 91 92 94 97 00 03 06

List of Figures

2.1	EU Structural Funds 2004-2006, Areas Eligible under Objective 1 and 2	9
2.2	Redistribution with Three Tiers of Government	17
2.3	Implemented Level of Investment for a Poor Region Depending on c .	26
4.1	Private Sector Employees and Self-Employed as Share of Total, by Type of Province, 1996-2006	86
4.2	Social Public Expenditure by Category as Share of Total, Province Average, 1996-2006	86
4.3	Sign of Derivative of Overall Redistribution wrt. ρ	96

List of Tables

3.1	Functional Breakdown of Public Investment	52
3.2	Results for One-step GMM by Type of Investment (per Trend GDP) $~$.	59
3.3	Classification of Taxes (in Decreasing Order of Control over Revenue	
	Sources)	63
3.4	Results Using Decentralization Measure tDec1	64
3.5	Results Using Decentralization Measure tDec3	65
3.6	Results for Regional Autonomy (Categories (d) to (e)) $\ldots \ldots \ldots$	66
3.7	Results, Public Inputs vs. Public Consumption Goods	67
3.8	Results Including FDI-Stocks (Stock of Inflows and Outflows, EU15) $$.	69
3.9	Results Including FDI-Index (FDI Flows, In and Out, per GDP, EU15)	70
3.10	Results in Levels, Investment in Infrastructure per Trend GDP	72
3.11	Results in Levels, Investment in Hospitals and Schools per Trend GDP	73
3.12	Results in Levels, Investment in Public Goods per Trend GDP $\ . \ . \ .$	74
3.13	Results in Levels, Investment in Redistribution per Trend GDP \ldots	75
3.14	Summary Statistics	82
3.15	Multi-co-linearity	82
3.16	Unit Root Tests	83
3.17	Q-test of Autocorrelation for Dependent Variable	83

Chapter 1

Introduction

1.1 Public Expenditure and the Institutional Setting of the Public Sector

What is the role of the public sector? According to Richard Musgrave (1959), it includes macroeconomic stabilization, income redistribution, and resource allocation. To realize these objectives, the state levies taxes and provides goods and services for individuals and firms. Oates (1972) already observed that *institutional factors* such as the federal structure of a state play a huge role in this context. E.g. if the federal government, a supranational organization (for instance the EU), or a strong private sector try to influence the decision process at the regional level, this may well determine the amount and quality of the goods and services provided. Thus, the distribution of decision power among different political institutions or the size of the public sector itself is likely to affect the composition of public expenditure. Until now, these links are still not completely understood. It is the aim of this thesis to cast some more light on the impact of the institutional setup of the public sector on its expenditure decisions.

Two institutional features are of particular interest to us: fiscal federalism and privatization. First, it is well known that the distribution of power among different tiers of government, that is *fiscal federalism*, may affect the spending decisions of the government. While there exists a huge literature on fiscal federalism with two tiers of government, the implications of more complex federal structures for the level and composition of public spending is still an open issue. For instance, the EU can be interpreted as a governmental structure with at least three tiers of government: regional and national governments as well as the EU itself. In this case it is no longer clear, how the different tiers of governments will behave and which level of public investment is actually realized. One important aspect here is that the federal structure does not only affect the level, but also the composition of public spending and investment. Second, *privatization* is another important institutional aspect, which might affect public spending: If a public firm is privatized, it is likely to change from a mostly social towards a more profit oriented objective. This may lead to changes in the demand for public services and goods and thereby the spending decisions of the government.

This thesis is organized as follows: The remainder of this chapter discusses the basic findings of the thesis and its policy implications in a broader context. Chapter 2 investigates how redistribution in more complex federal systems affects regional public investment. Chapter 3 provides an empirical analysis of what determines the composition of public investment in Europe, with fiscal decentralization being of special interest. Finally, Chapter 4, provides a theoretical framework to better understand how privatization may affect public spending and in particular redistribution. Note that all four Chapters can be read independently.

1.2 Fiscal Federalism and Public Expenditure

There is a huge literature on the determinants and implications of fiscal federalism. One of its basic questions is how to distribute spending and tax autonomy between the regional and the central governments. Oates (1972) developed a comprehensive framework to analyze the potential advantages and drawbacks of a centralized system. On the one hand, decentralization assures that decisions are made as close as possible to where people actually live. This facilitates the adjustment of the provision of public goods to the heterogeneous preferences of voters. This is what the famous "Decentralization Theorem" dating back to Oates (1972) basically says. On the other hand, externalities of different kinds may affect public expenditure (see e.g. Keen and Marchand (1998)). E.g. regions may compete in tax rates to attract firms or in public goods to attract human capital. In this case, inefficiencies arise due to strategic behavior of regional governments. A strong central government is then needed to set the proper incentives in order to implement the optimal provision of public goods at the local level. In practical terms both issues - response to local preference as well as externalities - are relevant. Most countries therefore opt for kind of a "in-between" solution - that is a federal system where decisions are taken partly by local and partly by federal institutions. However, the degree of fiscal federalism may also affect the level and composition of public spending and in particular of public investment as we will see below.

In Chapter 2, we try to understand the incentive effects generated by (de-)centralization in more complex federal systems. Indeed, federal systems are often more sophisticated than commonly assumed: In many cases, at least three tiers of government are involved in federal decision making. For instance in Germany or the US, there exist central, state, and local governments, implying a rather complex structure of the federal system in place. The EU itself is also organized beyond at least three tiers of government, though the EU as the highest tier disposes of only limited autonomy. This chapter is to better understand how the distribution of power among higher tiers of governments affects the investment decisions taken at the regional level. Based on Dahlby (1996), we set up a model with three tiers of government, to analyze federal redistribution in the presence of fiscal externalities. Our analysis identifies an additional qualitative disincentive effect, particularly for intermediate governments: They behave strategically in order to attract additional redistribution funds from outside, while still employing corrective policies towards their own regions. Our results also suggest that differently from the US, the federal system of the EU may lead to inefficiently low regional investment. This holds, because by construction the EU does not dispose over sufficient autonomy to thwart regional and national under-investment tendencies.

Chapter 3 provides an empirical investigation of the impact of decentralization not only on the level, but also on the composition of public investment. Though, public investment is an important variable in economics, its nature, drivers, and impact are still not completely understood. Most notably, there is often confusion about what it is in the first place. Perhaps the most prominent example of this type of confusion is the customary synonymous use of "public investment" and "infrastructure investment" in much of economic literature. Our data show, however, that there is a great deal of infrastructure investment that is not public, and there is a great deal of public investment that is not infrastructure investment. While it is well-known that many roads and municipal swimming pools are publicly funded and provided, both economic theory and empirical analyses have hardly distinguished between them. Keen and Marchand (1997) are among the first to think about what affects the composition of public spending. Based on their theoretical findings, we concentrate on public investment, only. The analvsis yields some interesting insights, most notably that fiscal decentralization boosts economically productive public investment, notably infrastructure, while economically less productive public investment, such as recreational facilities, remains unchanged. While not readily reconcilable with the traditional theory of fiscal federalism, these findings can be interpreted in terms of the literature on fiscal competition, with not only tax rates but also the quality of public expenditure (in particular infrastructure) weighing in firms' location decisions.

1.3 Privatization and Public Expenditure

Since privatization - by definition - affects the structure of the public sector it will most likely also have an impact on its spending decisions. First of all, privatization shifts the objective of firms from welfare maximization towards more profit orientation. Persons employed in a public firm may then be exposed to a lower risk of becoming unemployed than private employees. Until now, however, little is known about how privatization affects other sectors of government activity (Sheshinski et al. (2003)), in particular as regards the level and composition of public expenditure. For instance, a change in the demand for public goods may lead to an increase of public funds (higher taxation) or an adjustment of the composition of public spending, (e.g. public investment vs. social spending). Thus, in order to properly assess the benefits of privatization, it is essential to understand its implications for public expenditure.

Chapter 4 provides a framework to investigate how privatization might affect public social spending. Based on Kanbur (1981), the model includes two sectors, a private and a public, with the risk of becoming unemployed being higher in the private sector. An exogenously given increase of privatization then leads to a higher expected rate of unemployment as well as higher productivity of workers. Since it is the privatization policy of the government that generates additional risks, it is likely that voters will require it to bear the cost involved. The public sector will increase its social spending in order to satisfy the changing demand for public goods. We investigate how privatization affects the per capita unemployment transfers as well as overall redistribution if transfers are financed only through the profits of the public firms or through public profits and a lump-sum tax. Our results suggest that overall redistribution increase with privatization under rather mild assumptions, while per capita transfers decrease if redistribution is financed only through profits of public firms. On the other hand, if the government disposes over lump-sum taxes as additional tool to finance redistribution, both overall and per capita redistribution increase with privatization. Moreover, if it is costly to raise tax funds, it is no longer clear whether privatization leads to overall efficiency gains: Higher redistribution is accompanied by a higher need to raise costly tax funds, which outbalances at least some of the benefits from privatization.

1.4 Policy Implications

Public spending is an important driver of the overall economic performance. As our results suggest, there is a link between institutional characteristics - with our main focus being on decentralization and privatization - and the level and composition of public expenditure. This allows for interesting policy conclusions:

First, supra-national or federal cooperation may not only lead to efficiency gains, but

also generate additional disincentive effects at different levels of government. As Chapter 2 suggests, political decision processes in more complex federal systems cause additional strategic behavior of political representatives. This is particularly true for intermediate levels of government, which then try to exploit the central (or supranational) government in order to benefit their own jurisdictions. One should therefore bear in mind that supranational cooperation leads to a higher complexity and additional distortions within the system, which stand against its benefits (e.g. efficiency gains from free trade).

Second, in more complex federal structures, it is important to transfer sufficient autonomy towards the highest tier of government. Our findings have wide-ranging implications, especially for the European Union: If too little power is delegated to the highest tier of government, there is no way to evade the additional disincentive effects arising through the more complex federal structure. Thus, the additional externality in our setting requires more centralization. As the example of the EU shows, especially in transition or reform periods, this may be difficult to achieve. One possible solution to the problem is to delegate tax autonomy towards the EU to finance the required investment grants for the regions.

Third, institutional reforms - in particular decentralization - affect not only the level, but also the composition of public investment. This has wide-ranging implications for the economic performance and the provision of public goods at the local level. We find that decentralization leads to externalities, which distort the composition of public investment towards more infrastructure. Thus, against the common intuition, central governments should pay attention that their regions do not over-invest in order to attract private firms. This is an interesting result, in particular since most governments run comprehensive grant programs to boost regional investment.

Fourth, privatization is another important factor to understand the composition of public spending and the role of the state in general. While privatization implies less state intervention in one respect (less influence in firm's decisions) it entails higher government activity in other respects (more redistribution). This shift in government engagement generates additional indirect distortions: On the one hand, the additional need for public funds to finance redistribution may generate a social cost - e.g. through the excess burden of taxation. On the other hand, higher social spending may also be financed through a reduction in other types of public spending (e.g. public investment). In both cases, additional distortions arise, which at least partly outbalance the benefits of privatization. Thus, in order to correctly assess the benefits of a privatization reform, it is crucial to understand its potential indirect effects on the tax and expenditure policies of the government.

Note that one should be careful in deriving specific policy recommendations from these conclusions. Both theoretical and empirical findings are based on assumptions, which may not hold under real conditions. For instance, the three tiered federal structure in Chapter 1 assumes a benevolent government and complete information - assumptions which may not hold in real political decision processes. Furthermore, the empirical analysis in Chapter 3 by construction does not distinguish between the spending decisions made by different tiers of government. Thus, it is not straightforward to see, which implications these results have for a specific regional or federal government. Additional research would be highly valuable in this context.

Chapter 2

Fiscal Externalities in a Three-Tier Structure of Government

2.1 Introduction

In contrast to the common assumption in the literature on fiscal federalism, more than two tiers of government are involved in most federal decision-making. E.g. regions in the US finance expenditure by about 20 percent through federal transfers and by 35 percent through state grants. On the other hand, local governments in the EU receive more than 70 percent of their grants from national governments, while the EU itself only plays a minor role in the provision of regional funds.¹ This implies that the role of the intermediate government is much stronger in Europe than in the US. We argue that such differences in the allocation of power between the two highest tiers of government crucially determine the behavior at each level of government. In this context, the role of the middle-level governments is of particular interest: They employ corrective policies vis-a-vis their regions, while still engaging in strategic interaction with the highest level of government.

In order to understand the complexity of such federal systems, we focus on federal re-

¹Portugal is one important exception. Its regions receive about 19 percent of grants from international organizations. See Ford (1999) and Bergvall (2006) for further details.

distribution with fiscal externalities of public investment. Regional redistribution may generate a need for additional investment grants. This argument holds if federal redistribution exerts fiscal externalities leading to strategic behavior and under-investment by all regions (e.g. see Dahlby (1996)). However, with three tiers of government, not only the level of federal redistribution matters, but also the allocation of redistribution power between the two highest levels of government. This becomes particularly evident within the EU. There, middle-level and high-level authorities may have different objectives for federal redistribution and the use of corrective matching grants. Further, differently from the US, the highest level of government in the EU does not dispose over tax autonomy. For this reason it provides conditional transfers for investment only towards poor regions as shown in Figure (2.1). Thereby, the EU wants to achieve two aims at the same time: On the one hand, it redistributes from rich to poor regions. On the other hand, it still aims at implementing optimal regional investment. All this suggests that the analysis of fiscal externalities in more complex federal structures can be expected to become increasingly important in the future.



Figure 2.1: EU Structural Funds 2004-2006, Areas Eligible under Objective 1 and 2

Source: European Union (2006).

This paper investigates federal redistribution with three tiers of governments when regional investment generates fiscal externalities. Within this setting we try to understand the impact of fiscal externalities on investment targets and corrective policies at different tiers of government. Our findings suggest that a transfer of redistribution power towards the highest level of government generates an additional, qualitative incentive effect for middle level governments: They want their own regions to under-invest in order to attract additional redistribution funds from outside regions.

Within this setting, we also analyze the role of tax autonomy to finance corrective policies at the two highest levels of government. Three cases are distinguished: First, both high level and middle level governments finance corrective matching grants with region type specific taxation. This tax schedule allows concentrating on the pure disincentive effects arising from redistribution. Since the middle level government wants its regions to strategically under-invest in order to attract additional redistribution funds, only the highest level of government provides conditional transfers. Second, we assume a general lump sum tax to finance corrective matching grants. In this case, the tax regime generates an additional, indirect redistribution effect. Thus, middle level governments balance the positive effect from taxation with the negative effect of redistribution resulting in a higher investment target. In this case, both high level and middle level governments provide conditional transfers to implement first best investment at the regional level. Third, corresponding to the federal structure of the EU, we assume that only middle level governments dispose over tax autonomy to finance matching grants. In this case, the high level government can provide investment incentives to its regions only if it imposes investment restrictions on its equalization transfers. Thus, if redistribution funds available at the EU-level are low, this may lead to inefficiently low investment by both rich and poor regions.

The paper proceeds as follows. Section 2.2 gives a short literature overview. Section 2.3 presents the basic model with two tiers of government. Section 2.4 extends this setting to a three-tiered federal system with mutual redistribution by the two highest levels of governments. After characterizing the impact of mutual redistribution on the investment targets at all three tiers of government, Section 2.4.1 determines matching grant policies adopted by the two higher levels of governments with type specific taxation. This setting is generalized in Section 2.4.2 where we allow for general lump sum

taxes to finance matching transfers. Section 2.4.3 investigates the case where only the middle-level government can levy taxes. Section 2.5 concludes.

2.2 Literature Review

The analysis of intergovernmental grants and federal redistribution is well established in economic theory. A good overview is provided by Johnson (1988) proposing a general theory of redistribution. He also refers to problems arising from spillovers and factor mobility. However, he does not explicitly address strategic disincentives generated within the redistribution system itself. More recently, Persson and Tabellini (1996) present a general framework of redistribution and taxation. Oates (1972) provided an analytical framework justifying conditional grants. He proved optimality of matching grants to account for spillover effects of public goods beyond regional borders.

More than one decade later, Inman (1988) doubted whether the widespread use of conditional grants can only be justified by traditional efficiency and spillover arguments. His findings motivated further research to explain conditional grants as a common policy instrument, for instance, by introducing information asymmetries, fiscal externalities, and conditional grants in redistribution systems. E.g. Huber and Runkel (2006) assume information asymmetries among national and regional governments to show that conditional block grants or capped matching grants may be required to implement a second best solution. As Dahlby (1996) shows, federal redistribution generates fiscal externalities leading to under-investment by all regions. Regional governments account for the positive income effect of their investment resulting in less federal transfers in the future. This argument is empirically supported by the study of Matheson (2005). He finds that federal redistribution does discourage regional investment in Russia. Also, Fenge and Wrede (2004) pointed out that redistribution within the EU generates externalities. However, they do not refer to the impact of more sophisticated three-tiered federal systems on the implemented level of regional investment and optimal matching grants by both national and EU bodies. Martinez-Lopez (2005) argues that fiscal externalities may results in either under- or over-investment in a setting with two tiers of government by considering additional tax externalities. Since the empirical impact of this finding needs still to be explored, we abstract from his analysis and concentrate on the disincentive effects generated by fiscal externalities.

Although three-tiered federal structures are rather common, only few papers are concerned with this issue. E.g. see Cremer and Pestieau (1996) for a literature review on the distributive implications of European integration. However, in most cases this literature cuts the perspective of regional behavior and thereby falls back on an analysis with two tiers of government, only. One exception is Cremer and Pestieau (1997) considering income redistribution in a setting with three tiers of government. They identify a trade off between inter- and intra-national redistribution under incomplete information. Differently from their approach, we concentrate on fiscal externalities of redistribution and corrective policies in a setting with three tiers of government and complete information. This allows identifying the structural disincentive effects arising in more complex federal systems.

In a broader sense this paper is also related to the bailout problem prominently examined by Wildasin (1997). He also highlights strategic behavior of regions as means to acquire additional federal aid. However, compared to our analysis, Wildasin investigates a rather extreme case concerning strategic jurisdictional bankruptcy. In the present setting, conditional grant payments may be interpreted as instruments to prevent future bailout.

2.3 Benchmark Case: Two Tiers of Government

As a benchmark, we investigate the disincentives for regional investment generated within a redistribution system with two tiers of government. For this purpose consider a federation consisting of one central government and a large number (N) of regions of type 1 and of type 2, respectively. Production in region *i* is logarithmic and depends on the type specific productivity parameter ρ_i as well as public investment I_i :

$$Y_i = \rho_i \cdot \ln I_i \tag{2.1}$$

The analysis is undertaken from the perspective of a particular region of type 1, but results directly apply to any other region. You may interpret I_1 as a public input such as infrastructure or schooling outlays.² The representative individual in region 1 exclusively enjoys utility from private consumption C_1 . In order to isolate the pure strategic argument in this framework, local public investment is assumed to generate no direct utility or spillover effects. Individuals only benefit from regional public investment through increases in per capita income. The private budget constraint reads

$$C_1 = Y_1 + b \cdot \left(\frac{\sum_{k=1}^{2} Y_k}{2} - Y_1\right) - I_1.$$

The redistribution parameter $b \in [0, 1]$ may be referred to as the rate of income equalization among regions. For b = 0 no federal redistribution occurs, for b = 1 regional income gaps are completely offset.

Regional governments maximize the utility of the representative consumer. Together with the private budget constraint, the regional objective function writes

$$U(C_1) = U\left(Y_1 + b \cdot \left(\frac{\sum_{k=1}^{2} Y_k}{2} - Y_1\right) - I_1\right)$$

The central government maximizes utilitarian welfare over both types of regions.³

$$W = \sum_{i}^{2} U\left(Y_i + b \cdot \left(\frac{\sum_{k}^{2} Y_k}{2} - Y_i\right) - I_i\right)$$
(2.2)

 $^{^{2}}$ Note that usually transfers are determined on the basis of GDP in precedent years and investment generates benefits only in the future. However, introducing dynamics into the model complicates the analysis without providing further insights.

³Note that in this specification the central government wants to choose b such that marginal utilities are equalized over regions. Since we are only interested in how different tiers of government react to a redistribution system, we henceforth assume parameter b to be exogenously fixed. Indeed, in most countries the degree of federal redistribution seems to be rather inflexible at least in the mid-term, as federal structures are mostly established by complex political processes.

Maximizing the objective functions of the central and regional government wrt. I_1 shows that income equalization at rate b leads to under-investment of both rich and poor regions. Their investment target $(I_i = \rho_i \cdot (1 - \frac{b}{2}))$ is below the target of the central planer $(I_i = \rho_i)$. In line with Dahlby (1996), fiscal externalities of investment arise due to federal redistribution. Taking the level of redistribution, b, as granted, each region is aware of the positive income effect of its investment as well as its negative impact on redistribution transfers received from other regions: For a rich region, additional investment leads to higher transfers payable to poor regions. For a poor region, higher investment results in fewer transfers received from rich regions. Both types of regions do not account for the positive effects of their own investment on transfer commitments of other regions. The central planer, however, accounts for all fiscal externalities. Thus, its investment target exceeds the actual investment of both rich and poor regions. Clearly, the regional incentives to under-invest can not be resolved without additional intervention by the center.

Since redistribution generates incentives for regions to under-invest, the center has to achieve two objectives at the same time. First, it has to assure an exogenously given degree of redistribution among regions. Second, it wants to implement first best investment by introducing corrective policies. In line with the standard literature in public finance, corrective matching grants are an efficient instrument to provide incentives for regional expenditure (e.g. see Oates (1972)). This reasoning imposes the following sequence on the model:

- 1. The redistribution parameter, b, is exogenously determined.
- 2. The central government determines the matching rate γ_1 .
- 3. Regional governments invest $I_1(\gamma_1, b)$.

If regional government 1 receives a conditional matching transfer from the central

government, its budget-constraint is given by

$$C_1 = Y_1(I_1) + b \cdot \left(\frac{\sum_{k=1}^{2} Y_k(\cdot)}{2} - Y_1(\cdot)\right) - (1 - \gamma_1) \cdot I_1 - T_1.$$

 γ_i is the matching rate of the center towards region *i*. Matching transfers are financed by overall lump sum tax T_i , defined as

$$T_i = \frac{\sum_k^2 \gamma_k \cdot I_k}{2}.$$

This leads to

Proposition 2.1: Assume regional income equalization at rate b. Then,

i) matching rate γ₁ = 1 - 1-b/((1-b/2+U'/2)/(1-b/2)) implements first best investment by region 1.
ii) matching rates for rich regions are higher than for poor regions.
Proof see Appendix 2.6. ■.

We know that without further intervention by the central planer, regional governments under-invest for a given level of redistribution, b. In order to correct for strategic behavior, the central government provides additional matching grants for regional investment. If the center finances a share γ_1 of investment, regions effectively face a lower cost of investment and are therefore willing to invest more. This mechanism assures overall efficiency: The center can meet the exogenously determined redistribution target and at the same time assure first best investment by regional governments.

Note that the matching rate depends on relative marginal utilities. Regions of type 1 are poor relative to regions of type 2 for $U'_1 > U'_2$. If the difference in income rises, the center is willing to increase its investment target for the rich regions in order to partly close this gap: More production in the rich regions leads to higher transfers to the poor regions and thereby to a reduction in their marginal productivity. Thus, the center increases its matching rate for the rich type in order to boost overall production in the economy and redistribution towards the poor type. Note that for identical marginal utilities, the result simplifies to $\gamma_1 = \frac{b}{2}$, which corresponds to the externality generated by federal income equalization at share b with two types of regions involved.

2.4 Analysis with Three Tiers of Government

Disincentive structures and optimal corrective policies are much more complex in a federal system with three tiers of government. E.g. consider the US or the EU: Two tiers of government (federal and state vs. supranational and national governments) with distinct objectives mutually redistribute and engage in corrective policies towards their regions. However, there are crucial differences between the US- and EU-system. In the US, federal and state governments dispose over tax autonomy to finance matching transfers. In the EU, only national governments can levy taxes to finance conditional grants. As we will see this may have far-reaching implications for the implemented level of regional investment.

For the moment, the central government may either represent a federal or a supranational government. We label this the high-level or H-government. The state or national governments in-between are called middle-level or M-governments. In particular, assume two middle-level governments, A and B, both consisting of N regions of type 1 and 2 or of type 3 and 4, respectively, where N is assumed to be large. The highlevel maximizes over four representative regions, while each middle level government maximizes over two types of regions. This structure is shown in Figure (2.2).

From the perspective of a regional government, redistribution transfers are determined by two institutions at different hierarchical levels. The region is only concerned with the utility of its representative voter. Parameter c represents the share of redistribution exogenously assigned to the high-level government. Redistribution transfers for each region are now provided by one middle-level as well as high-level government: A share (b-c) of redistribution is accomplished by the middle-level government, while a share c of redistribution is transferred from the middle-level government to the high-level government: For example, if 10 percent of the regional income gap is to be covered by





high-level redistribution, the middle-level government reduces its share of redistribution by the same percentage. This assures that parameter c captures a qualitative shift in redistribution power while leaving its absolute level unchanged. Note that for c = 0, we fall back to the benchmark case discussed in Section 2.3.

In the following, preferences are supposed to be linear in consumption. This allows disregarding discrepancies between marginal utilities.⁴ Adopting again the perspective of a particular region of type 1, the regional objective function writes

$$U(C_1) = Y_1 + (b-c) \cdot \left(\frac{\sum_{k=1}^2 Y_k}{2} - Y_1\right) + c \cdot \left(\frac{\sum_{k=1}^4 Y_k}{4} - Y_1\right) - I_1$$
(2.3)

with $b, c \in [0, 1]$ assumed to be exogenously given and $b \ge c$. We define $Y_i(\cdot) = Y_i(I_i) = Y_i$. Transfers from the middle-level are identical to the redistribution grants of the central planer in Section 3, since it also considers two types of regions, only. The high-level government compares regional GDP with average GDP over all four types of regions and equalizes income disparities by a share c.

The middle-level government A considers utilitarian welfare of representative regions

⁴With a nonlinear utility function, results remain qualitatively unchanged. However, differences in marginal utilities among regions are then accounted for in form of weights, complicating calculations notably.

of type 1 and 2:

$$W_A^M = \sum_{i=1}^2 \left(Y_i + (b-c) \cdot \left(\frac{\sum_{k=1}^2 Y_k}{2} - Y_i \right) + c \cdot \left(\frac{\sum_{k=1}^4 Y_k}{4} - Y_i \right) - I_i \right).$$
(2.4)

Finally, the high-level government maximizes utility over all 4 types of regions:

$$W^{H} = \sum_{i=1}^{4} \left(Y_{i} + (b-c) \cdot \left(\frac{\sum_{k=1}^{2} Y_{k}}{2} - Y_{i} \right) + c \cdot \left(\frac{\sum_{k=1}^{4} Y_{k}}{4} - Y_{1} \right) - I_{i} \right).$$
(2.5)

Before going into the analysis of corrective policies by the two upper tiers of government, let us identify the effect of a shift in redistribution power towards the highest level (as measured by parameter c) on investment incentives of regional, middle-level, and high-level governments. Note that regions can realize their own preferred level of investment, as long as M- and H-governments do not dispose over matching transfers as additional policy instrument. The investment target of the middle-level government for region i is labeled I_i^M , the target for the high-level is labeled I_i^H . These targets are determined by maximizing Equation (2.4) and Equation (2.5) with respect to I_i , respectively. This leads to

Proposition 2.2: Assume that redistribution is implemented jointly by a M- and a H-government. Then,

- *i)* the investment incentive of a regional government of type 1 decreases in b.
- ii) the regional investment target of the M-government decreases in c.
- iii) the regional investment target of the H-government is independent of b and c.

iv) the regional investment target of the H-government exceeds the target of the Mgovernment. Thus, $I_1^H > I_1^M > I_1^*$.

Proof see Appendix 2.6. \blacksquare .

Quantitative disincentive effects from redistribution are captured by b. Thus, along the argument in Section 2.3, fiscal externalities at the regional level increase in the redistribution parameter b. For a higher value of b, regions can gain more by exploiting the federal redistribution system. On the other hand, the regional desire to invest is not affected by a shift in redistribution power among high level and middle level government as measured by parameter c. This result holds, since the number of regions of each type is already large: A transfer of more redistribution autonomy to a higher level considering more regions - does not change the strategic considerations of a regional government competing already with a large number of regions.⁵

A middle-level government only considers regions within its own borders. Due to constant marginal utilities and the symmetry in our model, there is no quantitative effect of redistribution (measured by b) at the middle-level. Redistribution among its regions per se, does not generate disincentives, since the M-government does take into account the fiscal externalities arising among its own regions. On the other hand, it knows well that for any c > 0, a lower output in region 1 leads to more redistribution at the central level and thereby potential resource inflows from outside regions. Thus, the middle-level government accounts for the qualitative change in redistribution power by behaving more strategically if parameter c increases. However, its investment target will always exceed the regional target, since it already accounts for strategic behavior among its own regions.

The high-level government maximizes utility over all regions in the federation. It properly accounts for all existing fiscal externalities within the system. Therefore, its investment objective is independent of b. For the same reason a qualitative change of redistribution as measured by a change in parameter c, does not occur at the high level. Its investment target is above the target of both middle level and regional governments and identical to the result for the central planer in Section 2.3.

Without corrective policies, regional under-investment occurs, since H- and M governments do not dispose over any means to affect regional behavior. Therefore, we introduce matching grants from the high and the middle-level government towards their regions. Three settings are of particular interest: Section 2.4.1 and 2.4.2 assume

⁵Note that for small values of N this result does not hold. High level redistribution then implies redistribution over a larger number of (countable) regions resulting in an increase of strategic behavior and thereby additional investment disincentives for each region.

that conditional transfers are financed by region type specific or lump sum taxes, respectively. The latter resembles the federal structure in the US. Section 2.4.3 refers to a structure similar to the EU, where the highest level of government can not impose taxes, but provides redistribution funds conditional on investment.

2.4.1 Non-Redistributive Funding of Matching Grants

In the following, we introduce conditional transfers, which are financed through regiontype specific taxes by both the high level and middle level. The high-level government may here be interpreted as a federal government, whereas intermediate governments correspond to state authorities. Consider b and c again to be exogenously determined and a federal structure as described in Figure (2.2). Middle- as well as high-level governments dispose over independent tax autonomy. The structure of the model evolves as follows:

- 1. Redistribution parameters b and c are exogenously determined.
- 2. Simultaneous Move Game between M- and H-government in γ_i^M and γ_i^H .
- 3. Regional governments invest $I_i(\gamma_i^M, \gamma_i^H, b, c)$.

 γ_i^M and γ_i^H are region specific matching rates provided by the middle-level and highlevel government towards region *i*, respectively. Intuitively, we impose non-negativity constraints $\gamma_i^H \ge 0$ and $\gamma_i^M \ge 0$. Referring again to a particular region of type 1, Equation (2.3) rewrites

$$U_1 = Y_1 + (b - c) \left(\frac{\sum_{k=1}^2 Y_k}{2} - Y_1\right) + c \left(\frac{\sum_{k=1}^4 Y_k}{4} - Y_1\right) - (1 - \gamma_1^M - \gamma_1^H)I_1 - T_1$$

with $T_1 = T_1^M + T_1^H$ defined by $N \cdot T_1^M = \sum_k^N \gamma_k^M \cdot I_k$ and $N \cdot T_1^H = \sum_k^N \gamma_k^H \cdot I_k$. In this setting, matching grants are financed by type specific lump sum taxes. This kind of funding does not generate additional investment distortions and isolates the impact of

direct redistribution on investment targets and corrective policies.⁶ Note also that γ_i^M and γ_i^H are strategic substitutes. The maximization problem of H- and M-governments are set up accordingly by referring to Equation (2.4) and Equation (2.5). Since the number of regions of each type is large, regions do not consider the tax effect in their optimization problem. Solving this game backwards leads to

Proposition 2.3: Assume that redistribution is implemented jointly by a M- and a H-government and matching transfers are financed by type specific taxation. Then, in the unique Nash Equilibrium, $\gamma_1^H > 0$ and $\gamma_1^M = 0$. The first best regional investment target of H, I_1^H , is implemented.

Proof see Appendix 2.6. \blacksquare .

Conditional grants are provided only by the high-level government, whose regional investment objective exceeds the target of the middle-level. The H-government completely takes into account fiscal externalities and therefore aims at implementing first best investment. On the other hand, the M-government intends to attract additional redistribution funds from outside regions for c > 0 (see Proposition 2). Therefore, it wants its regions to invest below first best. It provides a matching rate below the preferred level of the H-government. In response, the high-level increases its matching rate in order to prevent investment below its own target. This leads to over-investment from the perspective of the middle-level government, which in response reduces its matching rate even further. Repeating this line of argument results in a corner solution: The middle-level government does not provide any investment grants, since γ_1^M is bounded from below by $\gamma_1^M \epsilon$ [0, 1]. The high-level government can implement its first best target by choosing its matching rate $\gamma_1^H > 0$ high enough.

 $^{^6 {\}rm Section}~2.4.2$ addresses additional incentives arising from a more plausible funding of matching grants through general lump sum taxes.

2.4.2 Redistributive Funding of Matching Grants

In the following we analyze regional investment targets and corrective policies if the funding of matching grants generates additional incentive effects. Again, assume that c is exogenously determined and preferences are linear in consumption. As before, regions do not consider the tax effect in their optimization problem, since the number of each type of regions is large. Middle as well as high-level governments dispose over tax autonomy to finance conditional investment transfers. However, matching grants are now funded by general lump sum taxes according to $T_i = T^M + T^H$ with $4 \cdot N \cdot T^M = \sum_{k}^{4N} \gamma_k^M \cdot I_k$ and $2 \cdot N \cdot T^H = \sum_{k}^{2N} \gamma_k^H \cdot I_k$. In this sense, the general tax system varies over levels of government. Compared to the high-level government, a middle-level government includes fewer regions in the lump sum tax mechanism to finance matching grants. As we will see, funding matching grants thereby generates an additional, indirect redistribution effect. Solving the simultaneous move game backwards leads to

Proposition 2.4: Assume that federal redistribution is implemented jointly by a *M*and a *H*-government and matching grants are financed by general lump sum taxes. Then, in the unique Nash Equilibrium, $\gamma_1^H = c$ and $\gamma_1^M = b - c$. Investment targets of *H*- and *M*-governments are equal, $I_1^H = I_1^M$, and correspond to first best regional investment.

Proof see Appendix 2.6. \blacksquare .

General lump sum taxes generate positive investment incentives at the middle-level. The lump sum tax to finance matching grants varies over levels of government. Compared to the high-level, the tax schedule of the middle-level government includes fewer regions. This generates an additional, indirect redistribution effect: The M-government takes into account that conditional transfers from the high-level towards its regions are partly financed outside its own borders. In consequence, it is interested in attracting additional matching transfers provided by the high-level. Though, the potential of exploiting the redistribution system through under-investment remains unchanged. After all, this leads to an increase in the investment target of the middle-level government compared to the analysis in Section 2.4.1.

In the unique Nash Equilibrium, the H-government responds to the additional investment incentives at the middle-level by reducing its matching transfers. The high-level accounts for the positive impact of its matching rate on the investment target of the middle-level. It sets γ_1^H such that the target of the middle-level corresponds to the first best. This occurs if the tax effect neutralizes the negative effect of federal redistribution on the investment target of the M-government, which is achieved for $\gamma_1^H = c$. A matching grant γ_1^H above this level is not optimal, since it leads to incentives to over-invest for the M-government: Its willingness to exploit the tax system at the high level would be stronger than the disincentives from redistribution. A matching rate below this level is not optimal, since this implies a too low investment target at the middle-level and we would fall back to a situation as described in Section 2.4.1. Lump-sum taxes to fund matching transfers introduce additional, indirect redistribution among regions. In order to minimize this indirect effect, the high-level government wants γ_1^M to be as high as possible. This results in the Nash Equilibrium described above: Matching rates correspond to shares of redistribution c and (b-c), respectively, equalizing the incentives from the tax system and the redistribution system. Interestingly, the willingness of the M-government to attract more conditional transfers from the H-government effectively leads to a reduction of γ_1^H compared to the corner solution in Section 2.4.1 and to an equilibrium in which the M-government itself has to provide conditional transfers.⁷

This result can be related to some interesting patterns in the federal system of the US. There, federal as well as state governments provide redistribution transfers. Further, they all dispose over independent tax autonomy. Our results imply that in this case state governments want their own regions to strategically under-invest to attract additional funds from outside regions. Indeed, to correct for strategic behavior of regional and state governments, the US-federal government runs 600 grant programs, 550 of them being categorical. In addition, the federal government provides notable

⁷Note that this result does not change qualitatively if the assumption of linear utility is relaxed. The tax and redistribution effects remain active also in this case.

conditional transfers also to the states; about 85 % of the transfers towards state governments are earmarked and destined to be distributed among regions. This suggests that the federal government is indeed aware of the strategic behavior of both state and regional governments. Due to its dominant role it can provide sufficient investment incentives to evade strategic behavior of both levels and assure optimal regional investment (see Ford (1999) for further details).

2.4.3 High-Level Government without Tax Autonomy

Let us now turn to a federal system such as the EU, where the highest level of government is restricted in its fiscal autonomy. Consider again a federal structure as in Figure (2.2). However, we now think in terms of two independent nations constituting a supranational organization. Again, a fixed share c of redistribution authority is shifted towards the highest level (e.g. the EU). As before, assume that the number of regions is large and region 1 is poor compared to region 2. Differently from the analysis in Section 2.4.1 and 2.4.2, only middle-level governments can impose taxes. For the regional budget constraint this implies $T_i = T_i^M$. Since the highest level of government does not dispose over tax autonomy, it looses its preferred instrument to finance matching grants.

The only way for the high-level government to influence regional decisions is then to provide its redistribution transfers conditional on regional investment. Technically speaking, the high-level government then faces the additional constraint

$$0 \le \tau_i + \gamma^H \cdot I_i \le c \cdot \left(\frac{\sum_{k=1}^4 Y_k(\cdot)}{4} - Y_i(\cdot)\right), \qquad (2.6)$$

where τ_1 represents a lump sum transfer with $\tau_1 \geq 0$. As before, $\gamma_1^H \epsilon [0, 1]$. Equation (2.6) can be interpreted as a constraint to finance matching grants through redistribution funds directed towards this particular region. The total amount of matching grants payable to a region can not exceed its eligibility for redistribution funds. If redistribution funds at the high level are available beyond its need to finance matching

grants, it can in addition pay a lump sum transfers $\tau_1 > 0$. On the other hand, this implies that the high level government has no means to provide any funds to rich regions (Equation (2.6) is then bounded from below and $\tau_2 = \gamma^H \cdot I_2 = 0$). For that reason, the maximization problem of a rich and a poor region are structurally different. Rich regions face the following maximization objective:

$$C_2 = Y_2 + (b-c) \cdot \left(\frac{\sum_{k}^2 Y_k}{2} - Y_2\right) + c \cdot \left(\frac{\sum_{k}^4 Y_k}{4} - Y_2\right) - (1 - \gamma_2^M)I_2 - T_2^M.$$

They do not receive matching transfers from the H-government but contribute to finance its redistribution funds towards the poor regions. Using Equation (2.6), the objective function of a poor region writes

$$C_1 = Y_1 + (b-c) \cdot \left(\frac{\sum_k^2 Y_k}{2} - Y_1\right) - (1 - \gamma_1^M - \gamma_1^H)I_1 - T_1^M + \tau_1.$$

Poor regions receive matching grants from H- and M-governments. The high level finances its matching grants through redistribution funds, while the middle level government can impose type specific taxes to finance its investment grants. As before, the high level government maximizes over four types of regions, while the middle level government considers two types of regions. Solving this game backwards leads to

Proposition 2.5: Assume that the H-government can finance its matching transfers through its redistribution funds, only. Then,

i) for a rich region, $\gamma_2^M > 0$ and $\gamma_2^H = 0$ implement $I_2 = I_2^M$. ii) for a poor region three cases can be distinguished: If $c > \overline{c}$, then $\gamma_1^H > 0$ and $\gamma_1^M = 0$ implement $I_1 = I_1^H$. If $\underline{c} < c < \overline{c}$, then $\gamma_1^H > 0$ and $\gamma_1^M = 0$ implement $I_1^M < I_1 < I_1^H$. If $c < \underline{c}$, then $\gamma_1^H > 0$ and $\gamma_1^M > 0$ implement $I_1 = I_1^M$. This holds for $\overline{c} = \frac{\frac{b}{2}\rho_1}{\frac{\sum_k^4 Y_k}{4} - Y_1}$ and $\underline{c} = \frac{\frac{b}{2} \cdot \rho_1}{\frac{\sum_k^4 Y_k}{4} - Y_1 + \frac{\rho_1}{2}}$. Proof see Appendix 2.6. \blacksquare .

For rich regions, the intuition is straightforward. The high-level government does not dispose over tax autonomy. Since rich regions contribute to the redistribution system, the high level does not have means to provide any funds towards rich regions. On the other hand, the middle-level government finances matching transfers by type specific taxation as before. It is therefore able to finance matching grants also for the rich. Because middle-level governments want all their regions to strategically under-invest (desire to attract redistribution funds from outside regions), they provide too little matching grants. Accordingly investment in rich regions is below the first best.

Figure (2.3) shows that three cases can be distinguished for poor regions.





In range 3, c is large. The constraint in Equation (2.6) is not binding. The Hgovernment disposes over sufficient redistribution funds paid by rich regions in order to implement first best investment, I_1^H , in poor regions. This is achieved by providing redistribution funds towards the poor conditional on investment (see Equation (2.6)). Redistribution funds which are not needed to provide investment incentives can then be paid in lump sum form, that is $\tau_1 > 0$. This scenario is similar to the corner solution in Section 2.4.1 with $\gamma_1^H > 0$ and $\gamma_1^M = 0$. The high level government can implement its desired level of investment in poor regions by providing a sufficient amount of its

redistribution transfers conditional on investment.

In range 2, the value of c is too low in order to assure that the H-government can implement its desired target for regional investment. The constraint in Equation (2.6) is binding. However, c is large enough to allow the H-government to implement a level of regional investment, which is above the target of the middle-level if it provides all its redistribution funds to poor regions conditional on investment. Although first best investment can not be implemented, regions invest above the target of the Mgovernment, that is $I_1^M < I_1 < I_1^H$. Thus, the latter does not provide any matching grants towards the poor.

Finally, in range 1, c is small. Redistribution funds disposable at the high-level are too low to provide sufficient incentives to poor regions to even implement the target of the M-government, I_1^M - even if the H-government provides all its equalization transfers conditional on regional investment. Since the H-government wants to get as close as possible to its own investment target, it provides all its available funds conditional on investment. Therefore, $\gamma_1^H > 0$ and $\tau_1 = 0$. The middle-level then provides additional matching grants until its own target I_1^M is implemented, thus $\gamma_1^M > 0$.

The results in this section can be directly related to the federal system of the EU. We know from Proposition 2.3 that middle level governments also behave strategically in a federation with three tiers of government. In order to implement optimal regional investment, the highest level of government has to provide conditional transfers. However, since the tax autonomy of the EU is restricted, it has no means to influence decisions in rich regions through matching grants (see Figure (2.1)). Since national governments do not provide sufficient matching transfers, rich regions in Europe are likely to under-invest. Also for poor regions, the EU does not seem to dispose over enough redistribution funds to provide sufficient investment incentives: For low values of c, corresponding to a limited redistribution autonomy, the EU provides all its redistribution funds conditional on regional investment. Indeed, this is confirmed by Figure (2.1). Therefore, in terms of our model, the EU-system does most likely correspond to range 1 or 2 in Figure (2.3): Funds from the EU are exclusively conditional on in-

vestment suggesting that the EU is bounded in its desire to boost regional investment. This suggests that funds available at the EU level are not high enough (low value of c) and that national governments then restrict their conditional transfers to let their poor regions under-invest. Thus, our analysis suggests that both rich and poor regions in Europe invest below first best.

2.5 Conclusion

Federal systems are often organized beyond more than two tiers of government. This paper has proposed a framework to analyze strategic behavior in a three-tiered federal system. In particular, we investigate federal redistribution accomplished by two different tiers of governments when regional investment generates fiscal externalities. Our model reveals that for a given level of income equalization, a shift in redistribution power towards the highest level of government reduces the investment targets of the middle-level government. This may be interpreted as an additional, qualitative effect of redistribution arising in a federal system with three tiers of government.

Further, we characterize regional investment when the two higher levels of government engage in corrective policies. In this context we find that the design of the tax systems to fund corrective matching grants is one key feature in understanding government behavior, particularly at the intermediate level. Three cases are distinguished. First, we imagine mutual federal redistribution when corrective matching transfers are financed by type specific taxation. This tax schedule does not generate additional incentives and allows concentrating on the pure disincentive effects from redistribution. The middle level government wants its own regions to strategically under-invest in order to attract redistribution funds from outside regions. However, since the highest level government accounts for all existing disincentives in the system, in aims at implementing first best investment. Therefore, in the unique Nash Equilibrium, only the highest level of government provides conditional transfers.

Second, we generalize this setting by allowing for general lump sum taxes to finance

corrective matching grants. This tax rule, however, introduces an additional, indirect redistribution effect at the middle level. Since matching grants from the highest level are equally financed by all regions, the middle level government wants its own regions to invest more in order to attract additional conditional transfers from the upper level. Thus, middle level governments now balance a positive indirect income effect of the tax system with the negative effect of redistribution. As a result, both high level and middle level governments provide conditional transfers to implement first best investment in poor regions.

Third, corresponding to the federal structure of the EU, we assume that only middle level governments dispose over tax autonomy to finance matching grants. In this case, the high level government can provide investment incentives towards its poor regions, only if it imposes investment restrictions on its redistribution funds. Therefore, investment of poor regions crucially depends on the redistribution autonomy delegated towards the highest level of government. Only if the upper level disposes over sufficient redistribution funds, it can provide enough transfers conditional on investment to the poor regions to assure first best investment. On the other hand, the high level government has no means to provide any funds to rich regions (they contribute to federal redistribution) resulting in under-investment by the rich.

The implications of our model can be used to explain several features of federalism in Europe and the US. Results from Section 3.2 directly apply to the federal system in the US. They suggest that it is mainly the federal government that should provide investment incentives in order to implement optimal regional investment. This prediction is in line with the increasingly dominant role of the federal government in the US as well as its increased use of categorical grants in recent years. On the other hand, our findings in Section 2.4.3 can be related to the federal system in Europe: The EU does not dispose over tax autonomy. All it can do is to provide its redistribution funds conditional on investment in order to provide investment incentives at least for poor regions. While the desired level of redistribution can still be realized, this allows to implement high regional investment. This holds only if the EU disposes over sufficiently high redistribution funds. The fact that the EU provides all its funds conditional on investment is a strong signal for under-investment in both rich and poor regions in Europe. Thus, our model predicts that national governments are too dominant relative to EU institutions.

The proposed setting is but a first step in the analysis of more complex federal systems. There is plenty of scope for future research in this increasingly important field. In particular, the theoretical foundations for studying complex federal systems remain weak, especially as regards the behavior of the intermediate government in more complex situations. One example is to extend our framework to incorporate debt policy and to study the problem of bailouts, which may help to better understand strategic behavior of regional and state governments. Also it may be interesting to empirically examine the size of the qualitative effect of mutual redistribution as described in our framework. This could be particularly insightful for the EU where basically all member states dispose of at least two tiers of government.
2.6 Appendix

Proof of Proposition 2.1:

i) Maximizing the objective function of region 1 wrt. I_1 yields $I_1^* = \frac{\rho_1(1-\frac{b}{2})}{1-\gamma_1}$.

The maximization problem of the central government wrt. γ_1 leads to its target for the marginal productivity of region 1, $\frac{\partial Y_1(I_1^*)}{\partial I_1^*} = \frac{1}{1 - \frac{b}{2} + \frac{U_2'}{U_1'} \cdot \frac{b}{2}}$. These two results determine the optimal matching rate,

$$\gamma_1 = 1 - \frac{1 - \frac{b}{2}}{(1 - \frac{b}{2} + \frac{U_2'}{U_1'} \cdot \frac{b}{2})}.$$

This matching rate implements first best investment.

ii) If region 1 is poor compared to a region of type 2, then $U'_1 > U'_2$. Substituting this in γ_1 yields the result.

Proof of Proposition 2.2:

i) Deriving the objective function of the regional government wrt. I_1 yields the FOC of the regional government:

$$\frac{U(C_1)}{\partial I_1} = \left(\frac{\partial Y_1(I_1)}{\partial I_1} \left(1 - (b - c) - c\right) - 1\right) = 0$$

Transformation leads to

$$I_1^* = \rho_1 \cdot (1-b)$$
.

Derivation wrt. b yields the result.

ii) Deriving Equation (2.4) wrt. I_1 yields the FOC of the M-government:

$$\frac{\partial W^M}{\partial I_1} = \left(\frac{\partial Y_1(I_1)}{\partial I_1} \cdot (1 + (b - c)(\frac{1}{2} - 1) + c(\frac{1}{4} - 1)) - 1\right) \\ + \left(\frac{\partial Y_1(I_1)}{\partial I_1} \cdot (b - c)(\frac{1}{2}) + c(\frac{1}{4}) \cdot \frac{\partial Y_1(I_1)}{\partial I_1}\right) = 0.$$

This simplifies to

$$I^M = \rho_1 \cdot \left(1 - \frac{c}{2}\right).$$

Derivation wrt. c yields the result.

iii) Deriving Equation (2.5) wrt. I_1 yields the FOC of the H-government:

$$\begin{aligned} &\frac{\partial W^M}{\partial I_1} = \left(\frac{\partial Y_1(I_1)}{\partial I_1} \cdot \left(1 + (b-c)\left(\frac{1}{2} - 1\right) + c\left(\frac{1}{4} - 1\right)\right) - 1\right) \\ &+ \left(\frac{\partial Y_1(I_1)}{\partial I_1} \cdot (b-c)\left(\frac{1}{2}\right) + c\left(\frac{1}{4}\right) \cdot \frac{\partial Y_1(I_1)}{\partial I_1}\right) \\ &+ \left(c\left(\frac{1}{4}\right) \cdot \frac{\partial Y_1(I_1)}{\partial I_1}\right) + \left(c\left(\frac{1}{4}\right) \cdot \frac{\partial Y_1(I_1)}{\partial I_1}\right) = 0.\end{aligned}$$

This simplifies to

$$I_1^H = \rho_1.$$

Derivation wrt. b and c yields the result.

iv) Comparing the preferred marginal productivity levels of the M- and the regional

government yields

$$\frac{1}{(1-b)} > \frac{1}{(1-\frac{c}{2})}$$

for $b, c \in (0, 1]$ and $b \ge c$. Comparing the preferred marginal productivity level of the H- and the M-government yields

$$1 > \frac{1}{\left(1 - \frac{c}{2}\right)}.$$

for $c \in (0, 1]$. It follows directly that $I_1^* < I_1^M < I_1^H$.

Proof of Proposition 2.3:

The proof consists of four steps:

1. The maximization problem of a region of type 1 yields FOC

$$\frac{U(C_1)}{\partial I_1} = \left(\frac{\partial Y_1(I_1)}{\partial I_1} \left(1 - b\right) - \left(1 - \gamma_i^M - \gamma_i^H\right)\right) = 0.$$

This simplifies to

$$I_1^* = \frac{\rho_1(1-b)}{1-\gamma_1^M - \gamma_1^H}.$$

2. The M-government chooses γ_1^M by maximizing its objective function wrt. γ_1^M . Its FOC writes

$$\frac{\partial W^M}{\partial \gamma_1^M} = \left(\frac{\partial Y_1(I_1^*)}{\partial I_1^*} \frac{\partial I_1^*}{\partial \gamma_1^M} \cdot \left(1 + (b-c)(\frac{1}{2}-1) + c(\frac{1}{4}-1)\right) - \frac{\partial I_1^*}{\partial \gamma_1^M}\right) \\ + \left(\frac{\partial Y_1(I_1^*)}{\partial I_1^*} \frac{\partial I_1^*}{\partial \gamma_1^M} \cdot (b-c)(\frac{1}{2}) + c \cdot (\frac{1}{4}) \cdot \frac{\partial Y_1(I_1^*)}{\partial I_1^*} \frac{\partial I_1^*}{\partial \gamma_1^M}\right) = 0.$$

This yields the reaction function of the M-government

$$\gamma_i^M = 1 - \gamma_i^H - \frac{1 - b}{(1 - \frac{c}{2})}.$$

3. The H-government chooses γ_1^H by maximizing its objective function wrt. γ_1^H . Its FOC writes

$$\frac{\partial W^H}{\partial \gamma_1^H} = \left(\frac{\partial Y_1(I_1^*)}{\partial I_1^*} \frac{\partial I_1^*}{\partial \gamma_1^H} \cdot \left(1 + (b-c)(\frac{1}{2}-1) + c(\frac{1}{4}-1) \right) - \frac{\partial I_1^*}{\partial \gamma_1^H} \right) \\
+ \left(\frac{\partial Y_1(I_1^*)}{\partial I_1^*} \frac{\partial I_1^*}{\partial \gamma_1^H} \cdot (b-c) \cdot (\frac{1}{2}) + c(\frac{1}{4}) \cdot \frac{\partial Y_1(I_1^*)}{\partial I_1^*} \frac{\partial I_1^*}{\partial \gamma_1^H} \right)$$

$$+c \cdot \left(\frac{1}{4}\right) \cdot \frac{\partial Y_1(I_1^*)}{\partial I_1^*} \frac{\partial I_1^*}{\partial \gamma_1^H} + c \cdot \left(\frac{1}{4}\right) \cdot \frac{\partial Y_1(I_1^*)}{\partial I_1^*} \frac{\partial I_1^*}{\partial \gamma_1^H} = 0.$$

This yields the reaction function of the H-government,

$$\gamma_1^H = b - \gamma_1^M.$$

4. To determine the Simultaneous Move Nash Equilibrium, substitute γ_1^H in the reaction function of the M-government. This leads to

$$\gamma_1^M = 1 - b + \gamma_1^M - \frac{1 - b}{(1 - \frac{c}{2})}.$$

There is no interior solution for this problem.

Two potential corner solutions need to be verified given the parameter restrictions $\gamma_1^H, \gamma_1^M \ \epsilon \ [0, 1]$:

a): $\gamma_1^H = 0$: In this case, $\gamma_1^M = 1 - \frac{1-b}{(1-\frac{c}{2})}$. This in $\gamma_1^H = b - \gamma_1^M$ yields $\gamma_1^H = (1-b) \cdot (\frac{\frac{c}{2}}{1-\frac{c}{2}}) > 0.$

This is a contradiction.

b): $\gamma_1^M = 0$: In this case, the reaction function of the H-government yields $\gamma_1^H = b$. This in the reaction function of the M-government leads to $\gamma_1^M = (1-b)(\frac{-\frac{c}{2}}{1-\frac{c}{2}}) < 0$. The non zero constraint is binding. Therefore,

$$\gamma_1^M = 0.$$

The unique Simultaneous Move Nash Equilibrium of this game is a corner solution.

It remains to be shown that γ_1^H is the first best matching rate of the H-government and therefore implements I_1^H : With γ_1^M and γ_1^H from step 4, the optimality condition of the regional government 1, $I_1^* = \frac{\rho_1(1-b)}{1-\gamma_1^M-\gamma_1^H}$, leads to

$$I_1^* = \frac{\rho_1(1-b)}{1-b} = \rho_1.$$

This is identical to the first best marginal productivity level desired by the high-level government. Thus, γ_1^M and γ_1^H determined in step 4 implement I_1^H .

Proof of Proposition 2.4:

The proof consists of four steps:

1. The maximization problem of a region of type 1 yields FOC

$$\frac{U(C_1)}{\partial I_1} = \left(\frac{\partial Y_1(I_1)}{\partial I_1} \left(1 - b\right) - \left(1 - \gamma_i^M - \gamma_i^H\right)\right) = 0.$$

This simplifies to

$$I_1^* = \frac{\rho_1(1-b)}{1 - \gamma_i^M - \gamma_i^H}.$$

2. The M-government chooses γ^M by maximizing its objective function wrt. γ^M . Its FOC writes

$$\begin{split} &\frac{\partial W^{M}}{\partial \gamma_{1}^{M}} = \frac{\partial Y_{1}(I_{1}^{*})}{\partial I_{1}^{*}} \frac{\partial I_{1}^{*}}{\partial \gamma_{1}^{M}} \cdot \left(1 + (b - c)(\frac{1}{2N} - 1) + c(\frac{1}{4N} - 1)\right) + (1 - \frac{1}{2N}) \cdot I_{1}^{*} \\ &- \left((1 - \gamma_{i}^{M} - \gamma_{i}^{H}) + \frac{\gamma_{1}^{M}}{2N} + \frac{\gamma_{1}^{H}}{4N} \frac{\partial I_{1}^{*}}{\partial \gamma_{1}^{M}}\right) \\ &+ \left(\frac{\partial Y_{1}(I_{1}^{*})}{\partial I_{1}^{*}} \frac{\partial I_{1}^{*}}{\partial \gamma_{1}^{M}} \cdot \left((b - c)(\frac{1}{2N}) + c(\frac{1}{4N})\right) - (\frac{\gamma_{1}^{M}}{2N} + \frac{\gamma_{1}^{H}}{4N}) \frac{\partial I_{1}^{*}}{\partial \gamma_{1}^{M}}\right) = 0. \end{split}$$

This yields the reaction function of the M-government

$$\gamma_1^M = 1 - \gamma_1^H - \frac{4 - 2 \cdot b - c}{4}.$$

3. The H-government chooses γ_1^H by maximizing its objective function wrt. γ_1^H . Its FOC writes

$$\begin{aligned} &\frac{\partial W^{H}}{\partial \gamma_{1}^{H}} = \frac{\partial Y_{1}(I_{1}^{*})}{\partial I_{1}^{*}} \frac{\partial I_{1}^{*}}{\partial \gamma_{1}^{H}} \cdot \left(1 + (b - c)(\frac{1}{2} - 1) + c(\frac{1}{4} - 1)\right) + (1 - \frac{1}{4}) \cdot I_{1}^{*} \\ &- \left((1 - \gamma_{i}^{M} - \gamma_{i}^{H}) + \frac{\gamma_{1}^{M}}{2} + \frac{\gamma_{1}^{H}}{4}\right) \frac{\partial I_{1}^{*}}{\partial \gamma_{1}^{H}} \\ &+ \left(\frac{\partial Y_{1}(I_{1}^{*})}{\partial I_{1}^{*}} \frac{\partial I_{1}^{*}}{\partial \gamma_{1}^{H}} \cdot \left((b - c)(\frac{1}{2}) + c \cdot (\frac{1}{4})\right) - (\frac{1}{4}) \cdot I_{1}^{*} - (\frac{\gamma_{1}^{M}}{2N} + \frac{\gamma_{1}^{H}}{4}) \frac{\partial I_{1}^{*}}{\partial \gamma_{1}^{H}}\right) \end{aligned}$$

$$+ \left(c \cdot \left(\frac{1}{4}\right) \cdot \frac{\partial Y_1(I_1^*)}{\partial I_1^*} \frac{\partial I_1^*}{\partial \gamma_1^H} - \left(\frac{1}{4}\right) \cdot I_1^* - \left(\frac{\gamma_1^H}{4}\right) \frac{\partial I_1^*}{\partial \gamma_1^H} \right) \\ + \left(c \cdot \left(\frac{1}{4}\right) \cdot \frac{\partial Y_1(I_1^*)}{\partial I_1^*} \frac{\partial I_1^*}{\partial \gamma_1^H} - \left(\frac{1}{4}\right) \cdot I_1^* - \left(\frac{\gamma_1^H}{4}\right) \frac{\partial I_1^*}{\partial \gamma_1^H} \right) = 0.$$

This yields the reaction function of the H-government

$$\gamma_1^H = b - \gamma_1^M.$$

4. To determine the Simultaneous Move Nash Equilibrium, we substitute γ_1^H in the reaction function of the M-government. This yields $\gamma_1^M = 1 - b + \gamma_1^M - \frac{1-b}{(1-\frac{c}{2})} \cdot \left(1 - \frac{b-\gamma_1^M}{2}\right)$, which simplifies to

$$\gamma_1^M = b - c > 0.$$

Substitution into the reaction function of the H-government leads directly to

$$\gamma_1^H = c.$$

This is an interior solution.

We still need to check two potential corner solutions given the parameter restrictions $\gamma_1^H, \gamma_1^M \ \epsilon \ [0, 1]$:

a): $\gamma_1^H = 0$: In this case $\gamma_1^M = 1 - \frac{1-b}{(1-\frac{c}{2})}$. This together with $\gamma_1^H = b - \gamma_1^M$ yields

$$\gamma_1^H = (1-b) \cdot \left(\frac{\frac{c}{2}}{1-\frac{c}{2}}\right) > 0.$$

This is a contradiction.

b): $\gamma_1^M = 0$: In this case $\gamma_1^H = b$. This together with $\gamma_1^M = 1 - \gamma_1^H - \frac{1-b}{(1-\frac{c}{2})} \cdot \left(1 - \frac{\gamma_1^H}{2}\right)$

yields

$$\gamma_1^M = (1-b)(1 - \frac{1 - \frac{b}{2}}{1 - \frac{c}{2}}) > 0$$

for b > c. This is again a contradiction. The unique equilibrium of this game is the interior solution characterized above.

It remains to be shown that γ_1^H and γ_1^M together implement first best investment, I_1^H . With γ_1^M and γ_1^H from step 4, the optimality condition of the regional government, $\frac{\partial Y_1(I_1)}{\partial I_1} = \frac{1-\gamma_1^M-\gamma_1^H}{1-b}$, yields

$$\frac{\partial Y_1(I_1)}{\partial I_1} = \frac{1 - b + c - c}{1 - b} = 1.$$

This is identical to the first best marginal productivity desired by the H-government. Hence, γ_1^M and γ_1^H determined in step 4 yield regional first best investment I_1^H .

In the unique Nash Equilibrium, H- and M-governments are characterized by the same investment target. This can be seen from the optimality condition of the M-government, $\frac{\partial Y_1(I_1^*)}{\partial I_1^M}(1-\frac{c}{2}) = 1 - \frac{\gamma_1^H}{2}$. Together with the matching rates from step 4, this simplifies to

$$\frac{\partial Y_1(I_1^*)}{\partial I_1^M} = 1$$

corresponding to the marginal productivity target of the H-government.

Proof of Proposition 2.5:

i) Solving the maximization problem of a rich region (type 2) yields

$$I_2^* = \frac{\rho_2(4 - 2 \cdot b - c)}{4 \cdot (1 - \gamma_2^M)}.$$

The maximization problem of the M-government for the rich region yields

$$\gamma_2^M = 1 - \frac{4 - 2 \cdot b - c}{4 - 2c}.$$

This matching rate implements the regional investment target of the M-government,

$$I_2^M = \rho_1 \cdot (1 - \frac{c}{2}) < \rho_1.$$

Investment in rich regions is below first best.

ii) Solving the maximization problem of the poor region yields

$$I_1^* = \frac{\rho_1(1 - \frac{b}{2})}{1 - \gamma_1^H - \gamma_1^M}.$$

Similar to the approach in Proposition 4.2, we derive reaction functions

$$\gamma_1^M = 1 - \gamma_1^H - \frac{1 - \frac{b}{2}}{1 - \frac{c}{2}}$$

for the M-government and

$$\gamma_1^H = 1 - \gamma_1^M - \left(1 - \frac{b}{2}\right)$$

for the H-government. Solving the simultaneous move game by plugging reaction functions into each other reveals again that there is no interior solution for this problem.

Two potential corner solutions remain to be checked given the restriction on parame-

ters, $\gamma_1^H, \gamma_1^M \ \epsilon \ [0,1]$:

a): $\gamma_1^H = 0$: In this case $\gamma_1^M = 1 - \frac{1-\frac{b}{2}}{1-\frac{c}{4}}$. This together with $\gamma_1^H = 1 - \gamma_1^M - (1-\frac{b}{2})$ yields

$$\gamma_1^H = \frac{1 - \frac{b}{2}}{1 - \frac{c}{4}} - \left(1 - \frac{b}{2}\right) > 0,$$

which is a contradiction.

b): $\gamma_1^M = 0$: In this case $\gamma_1^H = 1 - (1 - \frac{b}{2}) = \frac{b}{2}$. This in the reaction function of the M-government yields

$$\gamma_1^M = \left(1 - \frac{b}{2}\right) - \frac{1 - \frac{b}{2}}{1 - \frac{c}{2}} < 0.$$

The non zero constraint is binding. Therefore, $\gamma_1^M = 0$. This is the unique Simultaneous Move Nash Equilibrium with $\gamma_1^H = \frac{b}{2}$ and $\gamma_1^M = 0$, a corner solution. This holds for $\gamma_1^H \cdot I_1 \leq c \cdot \left(\frac{\sum_k^4 Y_k(\cdot)}{4} - Y_1(\cdot)\right)$ or

$$c > \overline{c} = \frac{\frac{b}{2}\rho_1}{\frac{\sum_k^4 Y_k}{4} - Y_1}$$

In this case $\tau_i > 0$.

Besides this unrestricted solution with large c, we also have to consider the case where c is small. This distinction allows for two additional sub cases:

i): c small: holds for $\gamma_1^H < \gamma_1^{M*}$ and $\gamma_1^H \le (1 - \frac{1-\frac{b}{2}}{1-\frac{c}{2}})$. In this case we know that $\gamma_1^M + \gamma_1^H = (1 - \frac{1-\frac{b}{2}}{1-\frac{c}{2}})$ and $I_1 = \rho_1 \cdot (1 - \frac{c}{2})$.

The M-government implements its suboptimal desired level of investment. This holds if $\gamma^H \cdot I_1 \ge (1 - \frac{1 - \frac{b}{2}}{1 - \frac{c}{2}}) \cdot \rho_1 \cdot (1 - \frac{c}{2})$. Together, this yields $\gamma^H \cdot I_1 \ge (1 - \frac{1 - \frac{b}{2}}{1 - \frac{c}{2}}) \cdot \rho_1 \cdot (1 - \frac{c}{2}) > 0$

 $c \cdot \left(\frac{\sum_{k}^{4} Y_{k}(\cdot)}{4} - Y_{1}(\cdot)\right)$ which simplifies to

$$c < \underline{c} = \frac{\frac{b}{2} \cdot \rho_1}{\frac{\sum_{k=1}^{4} Y_k(\cdot)}{4} - Y_1(\cdot) + \frac{\rho_1}{2}}.$$

ii): Finally, $I_1^M \leq I_i \leq I_1^H$ if c lies between the two margins determined above:

$$\frac{\frac{b}{2}\rho_1}{\frac{\sum_k^4 Y_k}{4} - Y_1} = \overline{c} > c > \underline{c} = \frac{\frac{b}{2}\rho_1}{\frac{\sum_k^4 Y_k}{4} - Y_1 + \frac{\rho_1}{2}}.$$

However, note that the left hand side and the right hand side are evaluated at different levels of regional investment $(I_1 = \rho_1)$, the first best on the left hand side and $I_1 = \rho_1 \cdot (1 - \frac{c}{2})$, the target of the middle-level on the right hand side). To show that this range exists, we need to show that

$$\left[\left(\frac{\sum_{k}^{4} Y_{k}}{4} - Y_{1}\right) | I_{1} = \rho_{1}\right] < \left[\left(\frac{\sum_{k}^{4} Y_{k}}{4} - Y_{1}\right) | I_{1} = \rho_{1} \cdot \left(1 - \frac{c}{2}\right)\right],$$

With the production function defined in section 2.3, this can also be written as

$$\frac{\sum_{k}^{4}\rho_{k}\cdot ln(\rho_{k})}{4} - \rho_{1}\cdot ln(\rho_{1}) < \frac{\sum_{k}^{4}\rho_{k}\cdot ln\left(\rho_{k}\cdot\left(1-\frac{c}{2}\right)\right)}{4} - \rho_{1}\cdot ln\left(\rho_{1}\cdot\left(1-\frac{c}{2}\right)\right)$$

By transforming left- and right-hand side, one can easily show that this inequality is indeed true. ■

Chapter 3

The Composition of Public Investment and Fiscal Federalism: Panel Data Evidence From Europe¹

3.1 Introduction

Public investment has received only limited academic attention as an aggregate variable, and its composition has to our knowledge received none at all, at least in the European context. This paper seeks to fill that gap at least in part by presenting an empirical analysis of what drives different types of public investment, with a special focus on the impact of fiscal federalism.

Though public investment is an important variable in economics, its nature, drivers, and impact are still not completely understood. Most notably, there is often confusion about what it is in the first place. Perhaps the most prominent example of this type of confusion is the customary synonymous use of "public investment" and "infrastructure investment" in much of economic literature. There is, however, a great deal of infrastructure investment that is not public, and there is a great deal of public investment

¹This chapter is based on Kappeler and Välilä 2008. The theory part and most of the robustness checks are extensions presented only in this thesis.

that is not infrastructure investment. While it is well-known that many roads, water and sanitation networks, and municipal swimming pools are publicly funded and provided, both economic theory and empirical analyses have hardly distinguished between them when studying what determines "public investment" or how productive "public investment" is.

We are interested in the impact of decentralization on the composition of public investment. In doing so, we refer to the regional competition literature by arguing that regions compete for private capital by increasing the provision of more productive investment relative to less productive investment. However, there are two points worthwhile noting in advance: First, while this strand of literature concentrates on the composition of public expenditure as a whole, we apply this theory to the composition of public investment. This, however, can be done without difficulty. Second, in our empirical analysis we concentrate on regional competition (and therefore regional decentralization) rather than competition among nations to explain the level of different types of public investment. Thereby we account for the fact that public investment is mostly conducted at the regional level - though often influenced by the central government, e.g. by earmarked grants. We also argue that concentrating on regional competition constitutes a lower bound case: Regional and national governments have similar investment targets if externalities arise at both levels. This holds since both tiers of government in this case are biased into the same direction. Thus, national competition renders the effect observable at the regional level weaker. Therefore, significant decentralization parameters in our analysis represent kind of a lower bound case and would support the hypotheses that regional competition in public inputs does exist in Europe.

The theory of fiscal federalism - or any other theory for that matter - does not deal explicitly with the composition of public investment. At best, it distinguishes between consumption-oriented public expenditure and public expenditure to produce "public inputs" for the production processes of private firms. In what is to come we do not consider differences between current public spending and public investment per se; rather, we consider the various types of public investment as enhancements of production potential for different public services. Thus, infrastructure investment is considered to produce more future transportation services, and redistribution investment is considered to produce, e.g., more future recreation services. This perspective allows us to link the theory of fiscal federalism with the kind of data on the composition of public investment that we have.

The traditional theory of fiscal federalism is based on the seminal contributions by Tiebout (1956), Oates (1972), and Musgrave (1959). The underlying assumptions include, most importantly, the benevolence of the policy-maker in the centre (that is, his objective is the maximization of social welfare); the existence of pure local public goods and global public goods (whose benefits accrue locally and nation-wide, respectively): benefit taxation (same incidence for the cost and benefit of public spending); factor mobility; and absence of spill-over effects of fiscal decisions horizontally (between regions) and vertically (between regions and the centre). Considering the responsiveness of public spending to local preferences and the creation of incentives for economic efficiency as policy goals, the theory derives normative conclusions about the optimal task assignment between the central and sub-national levels of government. Responsiveness to local preferences implies that decentralization and fiscal competition are preferable in the provision of local public goods whenever local preferences are heterogeneous. On the other hand, centralization is warranted in the provision of public goods whose optimal supply cannot be achieved by fiscal competition. Such goods include most notably global public goods, and it also includes the macroeconomic stabilization and income redistribution functions of the government (which may be interpreted as providing global public goods as well). Finally, public goods may also have spillover effects, with one region benefiting from a highway built by its neighboring region, for example. Fiscal competition among sub-national levels of government will result in a sub-optimally low level of provision of such goods, as regions do not consider the spillover benefits in their individual decision-making. Oates (1972) suggests that the optimal provision can be achieved by means of matching grants from the centre, which act to internalize the externality.

More recent literature on fiscal federalism has relaxed the assumption of no spillover

effects in policy-making. Focusing on horizontal policy spillovers, consider regional tax competition. With capital mobile across regions that seek to attract it, tax competition can lead to sub-optimally low tax rates ("race to the bottom") and, as a consequence, insufficient provision of public services (both public consumption goods and "infrastructure"). In this sense, also Zodrow and Mieszkowski (1986) is a standard reference. Also, Hulten and Schwab (1997) discuss the circumstances where tax competition can lead to a sub-optimally low level of public capital. Competition between regions for an industry with external scale economies is a case in point: in competing for the location of such an industry, regions may reduce their tax rates so low as to unduly suppress public investment.

Several authors have come out against the assumptions in Zodrow and Mieszkowski (1986). E.g. Noiset (1995) argues that regional competition for private capital may reduce marginal costs up to a level where it actually increases the provision of public inputs. Also Keen and Marchand (1997) emphasize that the marginal cost for providing one additional unit of a public input does not necessarily exceed its additional gross return to capital (see also Matsumoto (1998) and Sinn (2003)). This holds if private capital reacts relatively strong to an increase in pubic inputs. Thus, relaxing the assumptions imposed by Zodrow and Mieszkowski (1986) renders over-provision of public inputs more likely. Considering the impact of fiscal competition on the composition of public expenditure, Keen and Marchand (1997) also argue that uncoordinated fiscal competition induces regions to over-invest in "local public inputs" at the cost of (consumption-oriented) local public goods. Investment in public inputs increases the potential of regions to attract mobile private capital, since public inputs reduce production costs for private firms. This generates distortions in the composition of public expenditure. Decentralization leads to an over-supply of public inputs and an under-supply of local public goods.

To sum up, fiscal competition has been argued to reduce public investment across the board (tax competition), but it has also been argued to boost productive public investment, at least relative to local public goods (broader fiscal competition). In terms of the public investment types in Table 1, these results would imply that decentralization

increases investment in Infrastructure as well as Hospitals and Schools, while reducing investment in Redistribution, at least in relative terms. This contrasts, notably, with the hypotheses above based on the older fiscal federalism literature, which suggests that regional tax-competition leads to under-provision of any kind of public goods.

As regards empirical literature, different types of public investment are rarely compared. Empirical work on the composition of public spending mostly refers to productive vs. unproductive public expenditure rather than different categories of public investment (Arze del Granado et al. (2005), Gonzalez Alegre (2006a), or Hauptmeier (2007)). There is one empirical paper on Bolivian data by Faguet (2004) investigating different types of public investment. However, he concentrates on the responsiveness of regional governments to local needs by focusing on one particular reform, only. As regards input competition in Europe, Devereux and Griffith (1998) and Devereux and Freeman (1995) argue that competition for private capital indeed does occur among EU countries. They analyze FDI flows from the US to several EU countries and do find competition effects. Further, Benassy-Quere et al. (2005) also show that input competition is an issue among EU-countries.

In the following, we derive hypotheses about the link between fiscal decentralization and the composition of public investment in Section 3.2 We then decompose public investment into different types with distinctly different economic characteristics in Section 3.3. Section 3.4 seeks to articulate empirical tests of the hypotheses, and results are interpreted from an economic perspective, before concluding in Section 3.5.

3.2 Modelling the Composition of Public Spending

This section presents an analytical framework based on Keen and Marchand (1997) to investigate the partly conflicting arguments on the provision of public inputs in more detail.² Consider the following utility function of a representative regional voter:

 $^{^{2}}$ Compared to the more general treatment in Keen and Marchand (1997), we disregard the labor market and only consider capital taxation. Though less general, this framework establishes the same results, which are of interest for the empirical analysis below.

$$V = V(X, G).$$

X is a private good and G is a public consumption good. Regional governments also provide a public input P, which directly enters the production function as specified below. Public goods G and P, can be financed either by a capital tax τ , or by a lump sum tax T. T is imposed independently of where companies locate. This leads to the following budget constraint of the public sector:

$$P + G = \tau \cdot K + T$$

The production function is defined by

$$F = F(P, K(P, \tau)).$$

K depends on the level of public inputs and on tax rate τ . Private capital inflows increase in P, $\frac{\partial K}{\partial P} > 0$, implying that private capital in this region increases in the provision of public inputs. On the other hand, investors try to evade taxes. Therefore, capital inflows decrease in τ , $\frac{\partial K}{\partial \tau} < 0$. The firm's objective is to maximize profits:

$$R = F(P, K) - (r + \tau) \cdot K - T.$$

For perfectly mobile capital, the marginal product of capital equals its gross return $(r + \tau = F_K)$, which directly results from the profit-maximization objective of private firms. Thus, the private budget constraint can be written as

$$X = F(P, K) - (r + \tau) \cdot K + \rho \cdot \overline{K} - T.$$

Consumption of good X equals production minus expenses for input K plus net return ρ (with $r = \rho + \tau$) from privately owned capital in this region (\overline{K}) minus the lump sum tax. Substituting public and private budget constraints in the utility function yields the objective function of the government:

$$maxV(X,G) = V\left(F(P,K) - (r+\tau) \cdot K + \rho \cdot \overline{K} - T, \tau \cdot K(\tau) + T - P\right)$$
(3.1)

3.2.1 Two Forms of Competition for Private Capital

Maximizing Equation (3.1) with respect to T and τ , respectively, leads to the following quantitative result on the provision of public goods G and P.

Proposition 3.1: Assume that the stock of private capital invested in the region depends negatively on τ and positively on public input P. Then, compared to first best provision, financing public input P and public consumption good G through a capital tax τ leads to under-provision of G, while P may be either under- or over-provided.

Proof see Appendix 3.6. \blacksquare .

Two externalities distort the provision of P. First, the presence of a distortionary tax instrument negatively affects the level of public inputs: Regions want to attract additional capital at the cost of other regions resulting into a race to the bottom in capital taxes. Second, regions benefit from a higher level of P through its positive effect on private capital inflows. This constitutes an additional externality, which distorts the level of P in the opposite direction: Regions compete in the provision of P in order to attract private capital. The more public inputs are available in a region, the more attractive it is for private firms to invest. Since it is not clear, which effect prevails, public input provision may be either too high or too low.

As regards public services, there is only one of the two effects in place. Distortionary capital taxes generate regional tax competition and thereby an inefficiently low level of G. Since G does not affect the private capital invested in a region, no strategic competition in G arises. There is no additional positive externality that might shift the provision of public consumption goods upwards. After all, marginal utility exceeds marginal costs implying strategic under-provision of local public services.

Concerning our empirical analysis on the composition of public investment, Proposition 1 implies that we expect decentralization to reduce investment in Redistribution. On the other hand, it is not clear whether decentralization increases or decreases the level of public investment in Infrastructure.

3.2.2 Composition of Public Investment

Based on the quantitative effects examined above, Keen and Marchand (1997) argue that uncoordinated fiscal competition induces regions to over-invest in public inputs relative to public services. To see this, consider the differential of the utility function defined above:

$$dV = V_X \cdot dX + V_G \cdot dG. \tag{3.2}$$

From the private budget constraint we know that $dX = F_P \cdot dP$ and from the public budget constraint that dG = -dP. Substituting yields

$$dV = \left(F_P - \frac{V_G}{V_X}\right)V_X \cdot dP$$

With this expression, we can test how utility changes, if, everything else constant, expenditure is shifted from G towards P:

Proposition 3.2: Holding capital tax τ constant, at a symmetric non-cooperative equilibrium, welfare increases through a small shift of expenditure from P towards G. Proof see Appendix 3.6. \blacksquare .

Regions can attract private capital by increasing the supply of P relative to G. Additional supply of P can attract private capital, while this is not possible with higher levels of G. On the other hand, the financing of additional public inputs through a distortionary capital tax negatively affects the level of private capital invested in this region. Thus, there are two externalities arising in context with the provision of public inputs in a region. This line of argument has direct implications for the relative composition of public investment in terms of P and G: P is overprovided relative to G in order to attract additional private capital. Welfare is thus unambiguously increased by a revenue neutral rebalancing of expenditures from P towards G. Note that Proposition 3.2 does not refer to over- or under-provision of public inputs in absolute terms. What it does say is that we can expect regional competition for private capital to increase productive investment (e.g. in Infrastructure and Hospitals and Schools) relative to public consumption goods (e.g. investment in Redistribution).

It is also worthwhile noting that Proposition 3.2 explicitly deals with welfare. However, it is difficult to empirically verify welfare implications of a certain policy tool. Also, the following empirical analysis will concentrate on levels of different types of public investment rather than shares. We therefore focus mainly on Proposition 3.1 for the economic interpretation of our results.

3.3 Composition of Public Investment in Europe: Stylized Facts

Based on the functional classification of government expenditure in the 1993 UN System of National Accounts and in the 1995 European System of Accounts (ESA 95), Eurostat provides a breakdown of public investment for EU countries starting in the early 1990s. Complete data are available for EU15 countries from 1995 (i.e, the introduction of ESA 95) through 2005.³ However, many countries have back-dated their time series to 1990. The "public investment" variable is gross capital formation of the general government. This includes changes in inventories, which may create some undesired noise for our analysis; however, the breakdown between gross fixed capital formation and changes in inventories is not available.

The functional breakdown of public investment is presented in Table 3.1. The righthand side column shows the functional classification (Classification of Functions of Government, COFOG for short) in ESA 95. The left-hand side shows our aggregation of the 10 available "functions" into four types of public investment with economically distinct roles. This aggregation will be used in the remainder of this paper.

Types of Investment	ESA 95 COFOG
1. Infrastructure (INF)	Economic Affairs;
2. Hospitals and Schools (HS)	Health; Education;
3. Public Goods (PG)	Defence; General Public Services; Environment; Order and Safety;
4. Redistribution (RED)	Housing; Recreation; Social Protection

 Table 3.1: Functional Breakdown of Public Investment

³EU15 comprises Austria, Belgium, Denmark, Finland, France, Germany, Greece, Ireland, Italy, Luxembourg, the Netherlands, Portugal, Spain, Sweden, and the United Kingdom.

The four different types of public investment affect the economy through different channels, with varying degrees of directness, and over different time horizons. Public investment in Infrastructure, consisting of just Economic Affairs in the ESA 95 CO-FOG,⁴ seeks to measure public investment in traditional infrastructure, mainly transport. This type of public investment has the most direct economic impact by reducing firms' production and transaction costs. The economic impact of public investment in Hospitals and Schools is more long-term and less direct in character, as it facilitates the building up and maintenance of the economy's stock of human capital. Investment in Public Goods affects the economy's allocative efficiency indirectly through framework conditions for productive activity. Finally, Redistribution affects the economy's income distribution rather than allocative or productive efficiency per se.

In addition to the composition of Infrastructure investment, some other aggregates shown in Table 1 contain undesirable "noise" as no further breakdowns of the righthand side "functions" are available. For example, public investment in water supply and wastewater management are not part of Infrastructure as one would wish; instead, they are part of Redistribution (Housing) and Public Goods (Environment), respectively. Similarly, one would wish to include street lightning in Public Goods; now it is in Housing and thereby Redistribution. However, as with Infrastructure, we expect such "noise" to be of sufficiently small magnitude so as not to invalidate the empirical analysis below.

⁴Economic Affairs comprise a number of different sectors, including agriculture; fuel and energy; mining, manufacturing, and construction; transport; communication; R and D; and others. Among these sectors, transport is likely to be by far the dominant recipient of public investment. Note that investment by energy companies owned by the public sector, for example, is classified as private investment in national accounts statistics as long as such companies are commercially run.

3.4 Empirical Analysis

Based on the theoretical findings in Section 3.2, Section 3.4 empirical investigates the impact of decentralization on the level of each type of public investment. Before presenting the methodologies and results of these analyses, we specify the model to be estimated.

3.4.1 Model Specification

The hypotheses formulated in Section 3.2 are based on literature, which considers productive vs. unproductive public spending. We directly apply these findings to analyze the composition of more productive vs. less productive types of public investment, only. For this reason we specify a reduced-form model to be estimated. In so doing we seek to identify exogenous variables measuring the impact of decentralization on public investment, as well as a set of control variables that render the model empirically well-specified.

The reduced-form specification to be used is as follows:

$$\begin{split} I_{c,it} &= \alpha + \beta_1 tax_{it-1} + \beta_2 cap_{it} + \beta_3 gdp_{it-1} + \beta_4 debt_{it-1} + \beta_5 lend_{it-1} + \beta_6 pop_{it-1} + \beta_7 year_t + \gamma_i + u_{it} \end{split}$$

where u_{it} i.i.d $(0,\sigma^2)$, with subscript *i* referring to observations in the cross-section dimension (individual countries) and *t* to observations in the time dimension. The dependent variable I_c represents public investment of type c, with $c \in 1, \ldots, 4$ as shown in Table 1. In the analysis I_c is expressed relative to trend GDP,⁵ thus in theory assuming values in \Re_+ .

Our primary interest is in the share of tax revenue attributed to sub-national levels of

⁵Considering ratios to (trend) GDP improves the time series properties of the variables and facilitates the economic interpretation of the estimation results. Note that trend GDP is calculated using the Hodrick-Prescott Filter with a smoothing parameter $\lambda = 100$.

government (regional and local governments), which is denoted tax.⁶ As regards the control variables, they seek to capture the general economic, fiscal, and demographic developments of significance for the determination of public investment. We control for investment grants from the central government to sub-national levels of government (cap); in the empirical analyses it is measured in relation to trend GDP.⁷ The tax share is lagged by one period to reflect the fact that investment decisions are most often taken a year before, based on knowledge about the revenue situation at that time. In contrast, capital transfers are contemporaneous with investment, as they finance investment the same year it is undertaken ⁸ Real GDP, denoted qdp in (1), is measured in per capita terms and lagged by one period to remove any simultaneity bias. The short- and longer term fiscal environment is captured by the budget surplus of the general government (lend) and public debt (debt). Both are measured in relation to trend GDP and lagged by one period, for the reasons mentioned above. We also control for population (pop).⁹ γ_i denotes unobserved time-invariant country-specific effects that are included in the estimations. Finally, as explained below in greater detail, a linear time trend (year) is included, as some of the time series are trend stationary.

3.4.2 Sample Properties

The main sample used in the estimations consists of a panel of EU10 countries (EU15 less the Cohesion countries less Luxembourg) during the period 1990-2005. Data are from Eurostat's New Cronos and OECD statistics. Time series for FDI-stocks, fiscal

⁶See Stegarescu (2005). We also considered other measures of decentralization, including total revenue share of sub-national levels of government; expenditure share of sub-national levels of government; and the ratio of sub-national tax revenue to expenditure. However, none of these alternative measures is conceptually superior to the tax share variable used, and all of them are empirically inferior, as they risk spurious correlation by including capital transfers (total revenue share) or the dependent variable (expenditure share), or by exhibiting non-stationarity (sub-national revenue-to expenditure ratio).

⁷The interaction term of tax and cap turned out to be insignificant in most of the estimations below and is therefore not reported.

⁸See Rodden (2003) for more details.

⁹As a robustness check we also considered unemployment, birth rates, migration rates, and mortality rates as additional control variables. They turned out to be mostly insignificant and did not change the estimation results materially.

variables (budgetary surplus, public debt, and decentralization measures), and population come from the OECD. Not all countries have back-dated all relevant series to 1990, so the panel is unbalanced.

Summary statistics are provided in Table 3.14 (Appendix 3.7). Overall the standard deviation is about 1/2 to 1/3 of the mean. Although this variation in the data is relatively low compared to large sample studies, it is still in line with other studies using national data such as Stegarescu (2005). The summary statistics also reveal that the three decentralization measures (taxsh, tDec1, and tDec3) used in this study have similar statistical properties.

Unit root tests indicate that our variables are either stationary or trend stationary (Appendix 3.7, Table 3.16), thus warranting the inclusion of a time trend as another explanatory variable. We perform Levin, Lin, and Chu test (LLC), [see Levin et al. (2002)] as well as Im, Pesaran and Chin test (IPS); [see Im et al. (1997)] to verify the stationarity properties of our variables.

The dependent variables are highly autocorrelated and persistent (Appendix 3.7, Table 3.17), with first-order autocorrelation coefficients between 0.8 and 0.9 for all types of public investment.

3.4.3 Methodology

The high autocorrelation in our dependent variables suggests a dynamic specification of model (1):

$$I_{c,it} = \alpha + \beta_1 I_{c,it-1} + \beta_2 tax_{it-1} + \beta_3 cap_{it} + \beta_4 gdp_{it-1} + \beta_5 debt_{it-1} + \beta_6 lend_{it-1} + \beta_7 pop_{it-1} + \beta_8 year_t + \gamma_i + u_{it}$$

The estimation of specification (2) will have to account for the correlation between the regressors (lagged dependent) and the composite term ($\gamma_i + u_{it}$), which renders least squares estimators inconsistent even asymptotically. To circumvent this problem we employ General Method of Moments (GMM) estimation, which has become the workhorse in estimating dynamic panel data models.

The intuition underlying GMM estimation is a follows. According to Arellano and Bond (1991) and Bond (2002) the levels equation is first-differenced, which eliminates the fixed effects γ_i , as well as the time trend. A set of (internal) instrumental variables is then specified that are orthogonal to the error term. Assuming that the error term is not serially correlated and that the explanatory variables are weakly exogenous, higherorder lags of the dependent variable $(I_{c,it-2}, I_{c,it-3}, \cdots)$ constitute valid instruments.¹⁰ This results in the following set of instruments

$$Z_{i} = \begin{pmatrix} I_{c,t-2} & 0 & 0 & \cdots & 0 & \cdots & 0 \\ 0 & I_{c,it-2} & I_{c,it-3} & \cdots & 0 & \cdots & 0 \\ \vdots & \vdots & \ddots & \vdots & \ddots & \vdots & \vdots \\ 0 & \vdots & 0 & \cdots & I_{c,it-2} & \cdots & I_{c,i1} \end{pmatrix}$$

While identification requires the number of instruments to equal the number of explanatory variables, overidentification is in practice necessary, as it both allows the testing of the moment conditions, as explained below, and improves efficiency.¹¹ The orthogonality requirement is equivalent to conditions on the first moment of the sample data.

$$E(Z'_{c,i} \Delta u_i) = 0 \text{ for } i = 1, 2, ..., N$$

where $\Delta u_i = (\Delta u_{i3}, \Delta u_{i4}, ..., \Delta u_{iT})'$. The fulfillment of these moment conditions is a sufficient condition for the asymptotic consistency of GMM estimators. The derivation of the GMM estimators can be done in one or two steps (labeled 1-step and 2-step GMM estimation), which are asymptotically equivalent under homoskedasticity of the error term. The two step estimator computes consistent estimate $\widehat{\Delta u_i}'$ in the first step, which are then used in the second step to minimize

¹⁰Higher-order lags of other explanatory variables can also be used as instruments under the same assumptions.

¹¹Note that GMM estimation dominates least squares estimation with instrumental variables, as GMM can be shown to be asymptotically efficient, which is not the case for least squares estimation with instrumental variables.

$$J_N = \left(\frac{1}{N}\sum_{i=1}^N \Delta u_i' Z_{c,i}\right) W_N\left(\frac{1}{N}\sum_{i=1}^N Z_{c,i}' \Delta u_i\right)$$
(3.3)

using weighting matrix

$$W_N = \left[\frac{1}{N} \sum_{i=1}^N Z'_{c,i} \widehat{\Delta u_i} \widehat{\Delta u_i}' Z_{c,i}\right]^{-1}.$$
(3.4)

Instead, the one step estimator directly minimizes

$$W_N = \left[\frac{1}{N} \sum_{i=1}^{N} Z'_{c,i} H Z_{c,i}\right]^{-1}$$
(3.5)

where H is a (T-2) square matrix with 2's on the main diagonal, -1's on the first off-diagonals and zero elsewhere. The 2-step procedure is more efficient under heteroskedasticity.

To test the validity of the moment conditions, the number of instruments has to exceed the number of explanatory variables. With overidentification, two tests suffice to assess whether the model is well-specified. First, the Sargan test for overidentifying restrictions is conducted to test the joint orthogonality of all instruments. Second, the serial correlation properties of the error term are tested; more specifically, tests developed by Arellano and Bond (1991) allow us to assess whether first order serial correlation is present in the error term (as it should, as our regression equation is estimated in first differences) and whether second order serial correlation is absent (as it should).

3.4.4 Basic Results

Table 2 presents the results of the preferred estimation method, which is one-step difference-GMM. We present results for total expenditure, overall investment, and the four investment types (all relative to trend GDP).

One-step difference-GMM estimation is alone in passing the Sargan test for overidentifying restrictions and residual autocorrelation tests (labeled m1 for first-order and m2 for second-order autocorrelation) for all four estimated models in levels. Twostep difference-GMM estimation is associated with the absence of first-degree residual autocorrelation throughout.

As shown in Appendix 3.7, the residuals from the least squares-based estimations are not always well-behaved. The FE OLS estimation suffers from residual non-normality, as indicated by the p-value of the Jarque-Bera (JB) test statistic.

1-step GMM	GEXP	GINV	1.INF	2.HS	3.PG	4.RED
Inv(lag)	0.36889***	0.46137***	0.425862^{***}	0.51883^{***}	0.61116***	0.49203***
	(4.65)	(4.47)	(4.78)	(3.94)	(4.90)	(4.89)
tax	0.04042	0.03342^{***}	0.01967^{***}	0.00754	0.01193^{**}	0.00077
	(0.33)	(2.08)	(3.36)	(1.04)	(2.38)	(0.16)
cap	3.20978^{***}	0.37123^{***}	0.05091	0.07498^{***}	0.10902^{*}	0.02656
	(4.10)	(2.16)	(1.38)	(3.11)	(1.82)	(0.42)
gdp	-0.35307	1.27741^{***}	0.47619^{***}	0.23047	0.33452^{***}	0.20746^{**}
	(0.17)	(3.16)	(2.46)	(1.41)	(3.55)	(2.39)
lend	0.01427	-0.00424*	-0.00098	-0.00024	-0.00121	-0.00203**
	(0.77)	(1.54)	(0.99)	(0.28)	(0.98)	(2.09)
debt	0.01057	-0.00019	-0.00061	0.00007	0.001839^{**}	0.00067
	(0.40)	(0.07)	(1.06)	(0.08)	(2.36)	(0.90)
pop	15.4208***	0.34454^{***}	-0.12783	0.22029	-0.03762	0.12006
	(2.44)	(0.03)	(0.43)	(0.98)	(0.08)	(0.45)
Sargan	1.0000	0.6066	1.0000	1.0000	1.0000	1.0000
m1	0.0239	0.0617	0.0465	0.0640	0.0248	0.0349
m2	0.4547	0.3229	0.3212	0.5098	0.1196	0.8624
Obs.	121	104	104	104	102	101

Table 3.2: Results for One-step GMM by Type of Investment (per Trend GDP)

One-step GMM calculated with heteroskedasticity robust standard errors. Sargan test for robust one-step GMM cannot be performed. We refer to test results from two-step GMM. Results for the constant are omitted. Note that the time trend is dropped for GMM since this procedure is based on first differencing. Before considering results of types of investment in more detail, let us interpret parameter estimates for overall expenditure and public investment: First, differently from the "Leviathan hypothesis", we observe that general expenditure does not depend on decentralization. On the other hand, capital transfers and the lagged dependent variable are highly significant. For our sample, decentralization mainly implies a shift in revenue or spending responsibility rather than a change in overall expenditure per se. Note that GDP does not positively affect the overall level of public expenditure. This is probably due to the inclusion of capital transfers into the equation. However, for consistency reasons, we do not change the model specification across estimations.

As regards estimates for general public investment (per trend GDP), it is roughly the sum of the parameter values for investment types one to four. Parameters for the taxshare, capital transfers, and GDP are significant. However, two exceptions do exist. First, the parameter estimates of the lagged dependent variables do not add up over the four types of investment. This is because the lagged dependent variable is the only variable, which is not identical across estimations (it is the lagged value of each of the four types of public investment per GDP). Thus, parameters are not directly comparable. Second, the four parameter estimates for the debt variable do not add up the debt effect in GINV. This may be due to the low significance of this variable in most regressions.

Considering the results by type of investment in Table 3.2, we conclude that a higher sub-national tax share increases the aggregate level of investment in Infrastructure and Public Goods, but it has no statistically significant impact on the aggregate public investment in Hospitals and Schools as well as Redistribution. The parameter estimates imply that an increase in the sub-national tax share by one percentage point leads to an increase in investment in Infrastructure of about 0.02 percentage points of GDP, or 2 percent, respectively, evaluated at sample mean.

Returning to Table 3.2 and considering the coefficient estimates for capital transfers, we observe a significant positive impact on investment in Hospitals and Schools as well as Public Goods. An increase of capital transfers by 1 percent of GDP boosts these types of investment by 0.07 and 0.11 percentage points of GDP (14 percent and 15 percent, respectively, at sample mean). As regards other control variables, real per capita GDP is positive and significant in all four models, except of hospitals and schools with coefficient estimates of 0.2-0.5. The fiscal variables are mostly insignificant, except that higher budgetary surpluses reduce investment in Redistribution and that higher public debt goes hand in hand with higher investment in Public Goods.

Economic Interpretation

Our results suggest that decentralization increases economically productive public investment, notably investment in public spillover goods (Infrastructure). There is no statistically significant impact of decentralization on public investment in consumptionoriented local public goods (Redistribution). Although one might have expected decentralization to increase Hospitals and Schools - as this could be interpreted as a more indirect kind of public input, this is not confirmed by our results. Thus, public Hospitals and Schools do not function as a public input here.

We saw in Section 2 how the theory of fiscal federalism could be used to derive some hypotheses about the composition of public investment. Most notably, it suggests that decentralized tax autonomy leads to lower provision of public consumption goods, while the level of public inputs may either decrease or increase (see Proposition 3.1). The impact of decentralization on our variable Public Goods was considered ambiguous, depending on whether it is dominated by local or global public goods. In other words, while not readily reconcilable with the traditional theory of fiscal federalism, especially as regards the provision of local public goods, these findings can be interpreted in terms of the literature on fiscal competition, with not only tax rates but also the quality of public expenditure weighing in firms' location decisions. Decentralization increases the level of investment in Infrastructure. This is in line with our theoretical analysis, particularly Proposition 3.1, which claims that public inputs provision may increase with decentralization. This being the case, we do not see any evidence of decentralization being associated with tax competition that would have a detrimental impact on public investment. On the other hand, we expected Redistribution to decrease with decentralization. This parameter, however, is not significant and does therefore not support our theoretical prediction in Proposition 3.1 as regards the provision of public consumption goods. This may be due to the stickiness of public expenditure: Over time public expenditure hardly decreases. But if kept constant (as Redistribution is in our case), inflation and more pronounced expenditure in other sectors, may lead to lower spending for this type of investment.

Note that the estimation results for total expenditure and GINV together suggest that our tax competition argument in terms of public inputs also holds in a more general context: total public expenditure as such is not affected by decentralization, while public investment does increase. This implies that decentralization leads to a shift of public expenditure towards public investment (at the cost of other, more consumption oriented expenditure). This is also in line with other studies such as De Granado (2006) and Arze et al. (2005).

3.4.5 Measuring Decentralization

To measure decentralization, Stegarescu (2005) also accounts for the autonomy of subnational governments to determine their tax base and/or tax rate. He argues that federal systems are more decentralized if sub-national governments rather than central governments can determine the tax rate or tax base - even if sub-national governments receive the same tax share. According to this argument, the OECD (1999) proposes a classification of taxes by decreasing order of regional control over tax revenues. This is shown in Table 3.3.

Based on this classification of tax autonomy, Stegarescu (2005) proposed three decentralization measures relative to the tax revenue of sub-national governments. We are interested in two of them, TDec1 and TDec3, as defined in Equations 3.6 and 3.7

 Table 3.3: Classification of Taxes (in Decreasing Order of Control over Revenue Sources)

(a)	sub-central government (SCG) determines tax rate and tax base
(b)	SCG determines tax rate only
(c)	SCG determines tax base only
(d)	tax sharing:
(d.1)	SCG determines revenue-split
(d.2)	revenue-split only changed with consent of SCG
(d.3)	revenue-split unilaterally changed by centr. gov. (CG) (legislation)
(d.4)	revenue-split unilaterally changed by CG (annual budget)
(e)	CG determines tax rate and tax base

Source: OECD (1999).

$$TDec1 = \frac{SCG \ own \ tax \ rev. \ (a) \ to \ (c)}{GG \ total \ tax \ rev.}$$
(3.6)

This index only considers regional tax revenues if the tax base and/or tax rate are independently determined by the regional government. Stegarescu also proposes a more general form of this index to measure decentralization:

$$TDec3 = \frac{SCG \text{ own tax rev. (a) to (e)}}{GG \text{ total tax rev.}}$$
(3.7)

This measure is closer to the taxshare used so far in this paper, since it relaxes the condition on local tax revenue to be taken into consideration. TDec3 accounts for all seven categories of tax revenue.

The decentralization measure, which we do not consider includes tax revenues of type a to c as well as d2 (This is labeled TDec2 in Stegarescu (2005)). Summary statistics in Table A1.1 show that the statistical properties of decentralization measures TDec1 and TDec3 are very similar for our sample. A further distinction among categories as suggested for TDec2 does not lead to further insights in our case.

Re-estimating our model by using TDec1 instead of the taxshare yields the following

results:

1-step GMM	1.INF	2.HS	3.PG	4.RED
Inv(lag)	0.45033^{***}	0.51164^{***}	0.64227^{***}	0.51713^{***}
	(4.37)	(4.10)	(6.21)	(4.82)
TDec1	0.01944^{**}	0.01093	0.01598^{***}	0.00264
	(2.35)	(1.44)	(2.90)	(0.49)
cap	0.05568^{*}	0.07261^{***}	0.08878^{*}	0.01402
	(1.75)	(2.87)	(1.87)	(0.23)
gdp	0.51547^{**}	0.33063^{**}	0.45739^{***}	0.21523^{**}
	(2.52)	(2.12)	(5.22)	(2.06)
lend	-0.00076	-0.00006	-0.00119	-0.00189**
	(0.77)	(0.07)	(1.01)	(1.99)
debt	-0.00008	0.00047	0.00214^{**}	0.00060
	(0.13)	(0.43)	(2.03)	(0.82)
pop	0.11729	0.31289	0.10848	0.14465
	(0.37)	(1.49)	(0.24)	(0.50)
Sargan	1.0000	1.0000	1.0000	1.0000
m1	0.0511	0.0588	0.0227	0.0287
m2	0.3053	0.7376	0.11074	0.9733
Obs.	105	105	103	102

 Table 3.4:
 Results Using Decentralization Measure tDec1

One-step GMM calculated with heteroskedasticity robust standard errors. Sargan test for robust one-step GMM cannot be performed. Test results from two-step GMM. Results for constant are omitted.

Substituting TDec1 by TDec3 yields the results in Table 3.5.

It turns out that the significance in general (and for decentralization in particular) follows the same pattern as in Table 3.2, though slightly decreasing for TDec3. TDec1 as well as TDec3 is significant for investment types 1 and 3, whereas parameter estimates for investment types 2 and 4 are not. Thus, the measures of decentralization as proposed by Stegarescu (2005) also confirm that decentralization leads to more investment in public infrastructure, which is the most evident public input among the four categories. This is in line with the theoretical findings in Section 3.2 that decentralization leads to strategic competition for private capital through increased public input provision.

1-step GMM	1.INF	2.HS	3.PG	4.RED
Inv(lag)	0.47685^{***}	0.49618^{***}	0.62314^{***}	0.51602^{***}
	(4.85)	(4.10)	(5.56)	(4.83)
TDec3	0.01332^{*}	0.01050	0.01468^{***}	0.00308
	(1.73)	(1.52)	(3.08)	(0.59)
cap	0.06357^{**}	0.07201^{***}	0.09452^{**}	0.01300
	(2.03)	(2.91)	(2.03)	(0.22)
gdp	0.38175^{**}	0.29501^{**}	0.39640^{***}	0.22499^{***}
	(2.05)	(2.40)	(4.97)	(2.62)
lend	-0.00058	-0.00014	-0.00129	-0.00200**
	(0.66)	(0.16)	(1.02)	(2.03)
debt	-0.00043	0.00013	0.00171^{*}	0.00066
	(0.65)	(0.14)	(1.86)	(0.89)
pop	0.00420	0.31694^{*}	0.71817	0.13854
	(0.01)	(1.65)	(0.16)	(0.52)
Sargan	1.0000	1.0000	1.0000	1.0000
m1	0.0538	0.0577	0.0271	0.02715
m2	0.2688	0.6030	0.1073	0.8852
Obs.	105	105	103	102

 Table 3.5:
 Results Using Decentralization Measure tDec3

One-step GMM calculated with heteroskedasticity robust standard errors. Sargan test for robust one-step GMM cannot be performed. Test results from two-step GMM. Results for constant are omitted.

3.4.6 Identifying Regional Autonomy

There is a further problem with our definition of decentralization. On may argue that decentralization also leads to tax competition among regions. This reasoning implies that tax revenues decline with more decentralization - and as a result also overall public investment (including Infrastructure). Our estimation would still yield a significantly positive decentralization parameter, although the actual effect was reversed.

Indeed, it is local autonomy, in what we are interested. If regions are more autonomous, we argue that they will use this autonomy to compete for private capital by increasing investment in Infrastructure. One way to evade the problem with tax competition is therefore to only concentrate on additional regional funds, which are controlled by the national government. If, e.g. tax revenue is levied by the central government and transferred to regions, they have no means to compete in tax rates. If these centrally levied funds still increase public infrastructure, we confirm that public input competition is the driving force in our setting (as suggested). Therefore, concentrating on spending autonomy allows identifying the competition aspect related to public inputs. Given the definition of decentralization as in Table 3.3, we therefore re-estimate our model by measuring decentralization through centrally controlled taxes, only. This comprises items (d) to (e) in Table 3.3.

1-step GMM	1.INF	2.HS	3.PG	4.RED
Inv(lag)	0.59324***	0.56097^{***}	0.69880***	0.52785***
	(4.99)	(5.06)	(7.82)	(5.34)
autonomy	0.03021***	-0.00370	0.00524	-0.00886
	(2.98)	(0.31)	(0.78)	(1.27)
cap	0.08652^{***}	0.08807^{***}	0.09638^{**}	0.02155
	(2.81)	(3.17)	(2.20)	(0.39)
gdp	0.26896^{*}	0.12164	0.17302^{*}	0.17956
	(1.81)	(0.86)	(1.81)	(1.57)
lend	0.00047	-0.00005	-0.00059	-0.00207**
	(0.33)	(0.07)	(0.56)	(2.16)
debt	0.00058	0.00000	0.00132	-0.00055
	(0.61)	(0.01)	(1.55)	(0.67)
pop	-0.23167	0.09492	-0.15547	0.05104
	(1.39)	(0.51)	(0.36)	(0.22)
Sargan	1.0000	1.0000	1.0000	1.0000
m1	0.0418	0.0572	0.0298	0.0212
m2	0.2411	0.3739	0.0868	0.7293
Obs.	105	105	103	102

Table 3.6: Results for Regional Autonomy (Categories (d) to (e))

As before, the decentralization parameter is significant for Infrastructure investment. This suggests that if more funds are transferred from the central government towards its regions (which may be interpreted as higher spending autonomy at the regional level), regions use these additional funds to increase their public infrastructure spending. This finding supports the view of Keen and Marchand (1997): They argue that regional competition (higher autonomy) leads to increased investment into public inputs relative to public consumption. Interestingly, the new decentralization measure has no longer an positive impact on Public Goods. Thus, this additional robustness check is even more in favor of our hypothesis in Proposition 3.1, which concerns Infrastructure and Recreation, only.

One-step GMM calculated with heteroskedasticity robust standard errors. Sargan test for robust onestep GMM cannot be performed. We refer to test results from two-step GMM. Results for constant are omitted.
3.4.7 Results for Aggregated Types of Investment

In order to directly refer to the theoretical findings in Section 3.2, we re-estimate the model by aggregating types of investment: Investment of type one and two is aggregated and labeled public input. Similarly we label the sum of public investment of type 3 and 4 as public consumption. Estimating the 1-step GMM model for theses two types of investment leads to the following results:

1-step GMM	INPUT	CONSUMPTION
Inv(lag)	0.38411***	0.47162^{***}
	(5.67)	(3.66)
tax	0.03324***	0.0069
	(4.50)	(1.23)
cap	0.12204*	0.19768^{***}
	(1.81)	(2.26)
gdp	0.81818***	0.56898^{***}
	(3.45)	(5.38)
lend	-0.00196	-0.00348*
	(1.15)	(1.71)
debt	-0.00144	0.00068
	(1.18)	(0.67)
pop	0.54318*	0.11267
	(1.74)	(0.17)
Sargan	1.0000	1.0000
m1	0.0462	0.0241
m2	0.3398	0.7701
Obs.	104	104

Table 3.7: Results, Public Inputs vs. Public Consumption Goods

One-step GMM calculated with heteroskedasticity robust standard errors. Sargan test for robust one-step GMM cannot be performed. We refer to test results from two-step GMM. Results for the constant are omitted.

Again, the results confirm the theoretical findings in Section 3.2 as regards public input provision. The tax share variable is significant at the 1 percent level only for public inputs, whereas it is insignificant for public consumption goods. This result directly replicates our theoretical finding from Proposition 3.1.¹² Decentralization leads to more competition among regions. In order to attract private capital, they invest more into public inputs. This raises the overall level of Infrastructure Investment in the

¹²The only difference is again that we distinguish among different types of investment, while the standard literature refers to public expenditure.

economy. On the other hand, public investment classified as Public Consumption remains unchanged.

3.4.8 External vs. Internal Competition

So far, we assumed that regional competition for private capital leads to higher overall investment in public inputs. However, given the structure of our data, it is not clear, whether this effect does not also depend on competition at the national level (among EU-countries rather than regions). E.g. Devereux and Griffith (1998) and Devereux and Freeman (1995) argue that competition for private capital indeed does occur among EU countries. Benassy-Quere et all (2005) also show that in this context input competition arises. So far we did not consider this aspect in our analysis. Our results might therefore suffer from an omitted variable bias. In the following we introduce a measure to account for inter-state competition. Thereby we account for the fact that some countries within the EU may be subject to fiercer external competition than others. This, of course, may also have an impact on the composition of public investment. In particular, Foreign Direct Investment is a good measure for the interaction of one economy with other countries. Therefore, we add to our analysis the sum of inward and outward FDI-stocks per GDP as additional explanatory variable. For better comparison, we refer again to the tax share as decentralization measure.

FDI stocks are mainly a measure for the long-term interaction among countries. However, one might argue that short-term variations are more likely to affect the composition of public investment. In order to capture also these short run variations, we also present results for an FDI index including inflows and outflows per GDP in Table 3.9.

1-step GMM	1.INF	2.HS	3.PG	4.RED
Inv(lag)	0.48364***	0.57731^{***}	0.52782^{***}	0.42669^{***}
	(5.67)	(4.79)	(4.90)	(3.16)
tax	0.01440***	0.00546	0.00729^{*}	0.00073
	(2.72)	(1.11)	(1.91)	(0.20)
cap	0.08402*	0.07898^{**}	0.12913^{***}	0.04128
	(1.95)	(2.55)	(2.89)	(0.60)
gdp	0.40320*	0.23424^{*}	0.23918^{**}	0.28992
	(1.86)	(1.66)	(2.41)	(1.63)
lend	0.00007	0.00147^{**}	0.00050	-0.00236***
	(0.07)	(1.24)	(0.55)	(3.90)
debt	-0.00033	-0.00000	-0.00074	-0.00051
	(0.29)	(0.00)	(0.71)	(0.32)
pop	-0.06791	0.24900	-0.20590	0.20738
	(0.24)	(1.18)	(0.51)	(0.75)
FDI- $stock$	-0.04437**	0.0561	0.08142^{***}	0.01085
	(2.22)	(0.72)	(4.07)	(0.25)
Sargan	1.0000	1.0000	1.0000	1.0000
m1	0.0646	0.1055	0.0352	0.0449
m2	0.2901	0.4209	0.2612	0.9588
Obs.	86	86	84	83

Table 3.8: Results Including FDI-Stocks (Stock of Inflows and Outflows, EU15)

One-step GMM calculated with heteroskedasticity robust standard errors. Sargan test for robust one-step GMM cannot be performed. We refer to test results from two-step GMM. Results for the constant are omitted.

It turns out that the basic findings regarding the impact of decentralization on the composition of public investment do not change by considering the openness of a country (apart from lower significance of Public Goods). Therefore, we conclude that our results are also robust towards external competition effects. However, against our intuition, the FDI-stock has a negative impact on Infrastructure and a positive effect on Public Goods, while FDI-flows are not significant for any type of public investment. These results suggest that competition among nations has no clear-cut effect on the composition of public investment. One possible explanation is that at this level public input competition and tax competition occur at the same time. In this case an unambiguous conclusion about which effect is significant is not possible - at least in our specification. However, this interpretation also confirms our approach to concentrate on regional competition, only, in order to identify public input competition (see also Section 3.4.6)

1-step GMM	1.INF	2.HS	3.PG	4.RED
Inv(lag)	0.43129***	0.65296^{***}	0.54930^{***}	0.40647^{***}
	(7.28)	(4.82)	(5.98)	(2.74)
tax	0.01558***	0.00430	0.00555	0.00448
	(2.72)	(1.05)	(1.28)	(1.50)
cap	0.03187	0.06014^{*}	0.15233^{**}	0.00931
	(0.70)	(1.68)	(2.41)	(0.20)
gdp	0.17770	0.18584	0.10854	0.35270^{***}
	(0.72)	(1.47)	(0.95)	(2.61)
lend	0.00019	-0.00005	-0.00183***	-0.00125**
	(0.13)	(0.06)	(2.61)	(1.99)
debt	0.00037	0.00022	-0.00041	-0.00102
	(0.36)	(0.15)	(0.50)	(0.62)
pop	-0.18475	0.27381	0.13970	0.34492
	(0.47)	(1.56)	(0.65)	(1.13)
FDI-flow	0.00002	0.00000	0.00000	0.00003
	(0.56)	(0.27)	(0.45)	(1.36)
Sargan	1.0000	1.0000	1.0000	1.0000
m1	0.0946	0.0823	0.0672	0.1046
m2	0.1478	0.7339	0.1294	0.3181
Obs.	83	83	81	80

Table 3.9: Results Including FDI-Index (FDI Flows, In and Out, per GDP, EU15)

One-step GMM calculated with heteroskedasticity robust standard errors. Sargan test for robust one-step GMM cannot be performed. We refer to test results from two-step GMM. Results for the constant are omitted.

Finally, there is also a theoretical argument why omitting international competition does not weaken our results: Assume that also national governments compete for private capital. In this case also central governments have incentives to increase investment in Infrastructure. In line with Kappeler (2007), the national (or middle level) and regional targets would be subject to a bias in the same direction to exploit the supranational government. Regional decentralization would then exert a limited additional effect on the level of public inputs. Thus, if competition at the national level is an issue, our results would constitute kind of a lower bound case: Though we only account for local competition, our decentralization parameter for infrastructure is significant. We may expect regional competition effect to be even larger if we separately capture an existing competition effect at the national level.

3.4.9 Methodological Robustness-Checks

As Bond et all (2001) suggest, for large T and small N the reliability of standard GMM-methods may be limited due to the large number of instruments generated in this case. To test, whether this problem is of relevance for our sample, we re-estimate our basic model with two alternative approaches: Besides Corrected Least Squares Dummy Variable estimates (LSDVC) as proposed by Bruno (2005), we also refer to 2SLS with only the second lag being used to instrument the lagged dependent variable.

Bond (2002) argues that GMM is generally more suitable for large N and small T. In our case, however, T (16) is large relative to N (10). Similar to OLS, one-step GMM in this case might be biased since the number of instruments becomes relatively large (in our case up to 125). 2SLS allows estimating the model with a smaller number of instruments. Thereby, we can correct for a large part of the bias (though probably not all), while the number of instruments remains low. Therefore, we also estimate the basic model with 2SLS with only the second lag as instrument for the lagged dependent variable allowing for more degrees of freedom.

To properly deal with a small sample bias in dynamic econometric models, a Corrected Least Squares Dummy Variable estimator is recently discussed in the econometric literature. Bruno (2005) proves efficiency of this estimator in case of unbalanced data with large T and small N such as ours. The estimator can be performed in two ways - the Anderson-Hsiao (AH) and Arrelano Bond (AB) based approaches. Judson and Owen (1996) show in a Monte Carlo Study that for T > N and a sample size similar to ours, the Anderson-Hsiao approach is the preferred one. Therefore, we expect the AH-approach to be superior compared to LSDVC AB in our case. For comparison reasons, we report results for both specifications together with 2SLS in Tables 3.10 to 3.13.

1.INV (level)	2SLS FE	LSDVC AB	LSDVC AH
Inv1(lag)	-0.787149	0.54962^{***}	0.47006***
	(0.48)	(6.09)	(4.46)
tax	0.023841***	0.01222^{**}	0.01407
	(3.17)	(2.06)	(0.07)
cap	0.08717	0.06166	0.06283
	(1.63)	(1.19)	(0.04)
gdp	0.67749^{***}	0.370^{**}	0.38334
	(2.94)	(1.99)	(0.06)
lend	-0.00470**	0.0003	0.00005
	(2.02)	(0.20)	(0.00)
debt	-0.00323**	0.00019	-0.00027
	(2.03)	(0.16)	(0.01)
pop	0.07607	-0.08348	-0.02325
	(0.21)	(0.25)	(0.00)
R^2 - Adj	0.3333		
Sargan		0.2861	
m1		0.0389	
m2		0.2021	
Obs.	111	104	104

Table 3.10: Results in Levels, Investment in Infrastructure per Trend GDP

GLS cross section weights used. Instrument used for 2SLS is the second lag of the dependent variable. For LSDVC, bias correction up to order O(1/T); boostrap variance-covariance matrix with 100 iterations.

2.HS (level)	2SLS FE	LSDVC AB	LSDVC AH
Inv2(lag)	0.22737*	0.63470***	0.69218***
	(1.67)	(7.74)	(7.46)
tax	0.00983^{*}	0.00482	0.00352
	(1.89)	(1.03)	(0.52)
cap	0.09906***	0.07822^{*}	0.07039
	(2.74)	(1.93)	(1.38)
gdp	0.27753*	0.23013	0.24561
	(1.84)	(1.58)	(1.27)
lend	0.00004	-0.00055	-0.00072
	(0.03)	(0.53)	(0.54)
debt	0.00021	0.00074	0.00055
	(0.24)	(0.84)	(0.48)
pop	0.62487**	0.20319	0.16069
	(2.25)	(0.71)	(0.44)
R^2 -Adj	0.4655		
Sargan		0.9542	
m1		0.0052	
m2		0.5355	
Obs.	111	104	104

Table 3.11: Results in Levels, Investment in Hospitals and Schools per Trend GDP

GLS cross section weights used. Instrument used for 2SLS is the second lag of the dependent variable. For LSDVC, bias correction up to order O(1/T); boostrap variance-covariance matrix with 100 iterations.

3.PG (level)	2SLS FE	LSDVC AB	LSDVC AH
Inv3(lag)	0.38665***	0.65266^{***}	0.56929^{***}
	(3.07)	(5.58)	(4.94)
tax	0.00817	0.00628	0.0107
	(1.44)	(0.97)	(1.07)
cap	0.15717***	0.10543^{**}	0.09517
	(3.01)	(2.07)	(1.31)
gdp	0.22465	0.21320	0.26253
	(1.34)	(1.28)	(1.05)
lend	0.00022	-0.00045	-0.00021
	(0.14)	(0.29)	(0.09)
debt	0.00122	0.00091	0.00092
	(1.20)	(0.80)	(0.57)
pop	0.17937	0.04649	-0.08657
	(0.64)	(0.13)	(0.17)
R^2 -Adj	0.5770		
Sargan		0.3576	
m1		0.7438	
m2		0.8014	
Obs.	109	102	102

Table 3.12: Results in Levels, Investment in Public Goods per Trend GDP

GLS cross section weights used. Instrument for 2SLS is second lag of the dependent variable. For LSDVC, bias correction up to order O(1/T); boostrap variance-covariance matrix with 100 iterations.

4.RED (level)	2SLS FE	LSDVC AB	LSDVC AH
Inv4(lag)	0.43572***	.54735***	.46214***
	(3.27)	(5.71)	(4.03)
tax	-0.00013	0.00084	0.00123
	(0.03)	(0.16)	(0.14)
cap	0.05590	0.05183	0.05221
	(1.43)	(1.45)	(0.85)
gdp	0.22701	0.23844^{*}	0.25135
	(1.59)	(1.71)	(1.02)
lend	-0.00168	-0.00157	-0.00179
	(1.32)	(1.39)	(0.91)
debt	-0.00031	-0.00000	-0.00035
	(0.38)	(0.01)	(0.23)
pop	0.10678	0.12188	0.14232
	(0.48)	(0.49)	(0.32)
R^2 -Adj	0.4943		
Sargan		0.7139	
m1		0.0010	
m2		0.8878	
Obs.	108	101	101

Table 3.13: Results in Levels, Investment in Redistribution per Trend GDP

It turns out that 2SLS supports our basic estimations in Section 3.4.4: As before, the tax variable is significant for Infrastructure, while Redistribution is unaffected by a change in the tax share. Moreover, Hospitals and Schools are also positively affected by decentralization, while Public Goods are not. Thus, 2SLS speaks even more in favor of our theoretical hypothesis on input competition than 1-step GMM does. The results suggest that the potential bias of 1-step GMM for our sample is likely to be small.

In order to further verify the reliability of 1-step GMM with our data, we also use LSDVC. However, the different specifications of LSDVC do not provide a clear picture: The Sargan test and Arrelano-Bond autocorrelation tests for first-stage regressions do not reject LSDVC AB. Therefore, we conclude that this estimator is consistent. Indeed, the results are in line with our findings in Section 3.4.4 in terms of significance and parameter values - particularly for our decentralization variable. The only difference occurs for investment of type 3: Its decentralization parameter is not significant. This supports our robustness checks in Sections 3.4.5 to 3.4.8. Thus, overall AB LSDVC is

GLS cross section weights used. Instrument for 2SLS is second lag of the dependent variable. For LSDVC, bias correction up to order O(1/T); boostrap variance-covariance matrix with 100 iterations.

in line with our hypothesis that regions compete in public inputs for private capital and suggests that the bias underlying the GMM and FE OLS results does not drive our results.

As mentioned, in theory the LSDVC approach based on Anderson-Hsiao (AH) outperforms the GMM based LSDVC estimator (AB) for N and T as in our sample (again, see Judson and Owen (1996)). Therefore, we also present AH-LSDVC results in Tables 3.10 to 3.13. It becomes evident that all variables apart from the lagged dependent are insignificant. This puts doubts on whether these results should be interpreted in our case. In particular, it seems puzzling that neither GDP nor conditional capital transfers do affect any of the four types of public investment. We would expect at least some of them to be positively correlated with either GDP or conditional transfers. The reason for the weak performance of AH LSDVC is likely to be due to the high level of aggregation of our data and the structural inbalancedness of our sample (there are few years missing for most countries, whereas for other years the sample is complete). Given the discrepancy between the empirical superiority of LSDVC AB (which is accepted by statistical tests) and the theoretical superiority of AH (but its overall insignificance), LSDVC serves as a robustness check rather than our basic regression methodology.

3.5 Conclusion

The analysis of the relationship between fiscal decentralization and the composition of public investment is first-of-a-kind, at least in the European context. It yields some interesting insights, most notably that fiscal decentralization seems to boost economically productive public investment and to curb the relative share of economically less productive public investment.

While not readily reconcilable with the traditional theory of fiscal federalism, especially as regards the provision of local public goods, these findings can be interpreted in terms of the literature on fiscal competition, with not only tax rates but also the quality of public expenditure weighing in firms' location decisions. The finding that decentralization reduces the relative share of Redistribution investment can also signal over-investment in more centralized system with competition for a common pool of resources.

Several robustness checks are conducted to verify model specification and econometric approach: Redefining our measure of decentralization did not affect our result that public inputs increase in decentralization. Also, regrouping types of public investment as well as the inclusion of external competition in the analysis confirm our findings. In terms of methodological robustness checks, Two Stage Least Squares and Corrected Least Squares Dummy Variable estimators suggest that the potential bias of 1-step GMM estimators with macro data is unlikely to drive our results.

Clearly, this is but a first step in the analysis of the composition of public investment. There is plenty of scope for future research to tackle issues that our analysis leaves open. The theoretical foundations for studying the composition of public investment remain thin, especially as regards the articulation of an explicit link between fiscal federalism and different types of investment. Empirical examination of different types of public investment could usefully focus on differences in their productivity, as well as on a more nuanced examination of what drives the different types of investment, including but not limited to fiscal federalism.

3.6 Appendix

Proof of Proposition 3.1:

Proof by comparing first best provision with the non-cooperative equilibrium:

First Best $(T > 0 \text{ and } \tau = 0)$:

FOC wrt. T:

 $-V_X + V_G = 0.$

Thus, $\frac{V_G}{V_X} = 1$, this implies first best provision of G.

FOC wrt. P:

$$V_X \cdot (F_P + F_K \cdot K_P - (r+\tau) \cdot K_P) + V_G \cdot (\tau \cdot K_P - 1) = 0.$$

Thus, $F_P = \frac{V_G}{V_X} (1 - \tau \cdot K_P).$

This implies $F_P = 1$ since $\tau = 0$ and T > 0 in the first best and $-V_X + V_G = 0$ by FOC wrt. T.

Non-cooperative equilibrium $(T = 0 \text{ and } \tau > 0)$:

FOC wrt. τ :

$$V_X \cdot (F_K \cdot K_\tau - K - (r + \tau) \cdot K_\tau) + V_G \cdot (K + \tau \cdot K_\tau) = -V_X \cdot K + V_G \cdot (K + \tau \cdot K_\tau) = -V_X \cdot K + V_G \cdot K(1 + \frac{\tau \cdot K_\tau}{K}) = 0 \text{ (using } r + \tau = F_K).$$

Thus, $\frac{V_G}{V_X} = \frac{K}{K\left(1 + \frac{\tau \cdot K_T}{K}\right)} > 1$. This implies under-provision of G compared to the first best.

FOC wrt. P:

$$V_X \cdot (F_P + F_K \cdot K_P - (r + \tau) \cdot K_P) + V_G \cdot (\tau \cdot K'_P - 1) = 0.$$

From this,

$$F_P = \frac{V_G \cdot (1 - \tau \cdot K_P)}{V_X} = \frac{(1 - \tau \cdot K_P)}{1 + \frac{\tau \cdot K_T}{K}}.$$

Two effects are responsible for the deviation from first best $(F_P = 1)$: $-\tau \cdot K_P$ is the public input effect, which positively affects the provision of P. $\frac{\tau \cdot K_{\tau}}{K}$ is the tax competition effect, which has a negative impact on the provision of P. It is no longer clear, whether P is over- or under-provided relative to the first best.

Proof of Proposition 3.2:

The proof consists of two steps. First determine F_P and $\frac{V_G}{V_X}$ for the non-cooperative equilibrium. Second, substitute these results in Equation (3.2). Thereby, one can show that at the optimum $\frac{dV}{dP} < 0$:

1. Determine F_P and $\frac{V_G}{V_X}$ for the non-cooperative equilibrium:

$$L = V \left(F(P, K) - (r + \tau) \cdot K + r\overline{K} - T, (\tau \cdot K(\tau, P) - P) \right)$$

FOC wrt. P:

 $V_X \cdot (F_P + F_K \cdot K_P - (r + \tau) \cdot K_P) + V_G \cdot (\tau \cdot K_P - 1) = V_X \cdot F_P + V_G \cdot (\tau \cdot K_P - 1) = 0$ (using $r + \tau = F_K$)

FOC wrt. τ :

$$-V_X \cdot K + V_G \cdot (K + \tau \cdot K_\tau) = 0 \text{ (using } r + \tau = F_K)$$

From FOC wrt. P:

$$F_P = (1 - \tau \cdot K_P) \cdot \frac{V_G}{V_X}$$
. From FOC wrt. τ : $\frac{V_G}{V_X} = (1 + \frac{\tau \cdot K_\tau}{K})^{-1} > 1$

for K_{τ} smaller zero. The marginal productivity of G is larger than that of the private good. G is underprovided.

2. Plug F_P and $\frac{V_G}{V_X}$ into Equation (3.2):

$$dV = \left(F_P - \frac{V_G}{V_X}\right) V_X \cdot dP = \left(\left(1 - \tau \cdot K_P\right) \cdot \frac{V_G}{V_X} - \frac{V_G}{V_X}\right) V_X \cdot dP$$
$$dV = \left(\left(1 - \tau \cdot K_P\right) - 1\right) \left(1 + \frac{\tau \cdot K_\tau}{K}\right)^{-1} V_X \cdot dP$$
$$\frac{dV}{dP} = -\left(\frac{\tau \cdot K \cdot K_P}{K + \tau \cdot K_\tau}\right) V_X < 0.$$

Since we know that $\frac{V_G}{V_X} = \frac{K}{K\left(1 + \frac{\tau \cdot K_T}{K}\right)} > 1$, it directly follows that $K + \tau \cdot K_\tau > 0$.

Thus, $\frac{dV}{dP} < 0$ implies that rebalancing expenditure from P towards G strictly increases utility.

3.7 Appendix

Variable	Obs	Mean	Std. Dev.	Min	Max
inv1	134	0.00766	0.00351	0.00270	0.01849
inv2	134	0.00511	0.00235	0.00127	0.01345
inv3	132	0.00596	0.00196	0.00231	0.00944
inv4	133	0.00456	0.00246	0.00142	0.01215
tax	158	0.18972	0.11003	0.02061	0.35691
tdec1	160	0.12134	0.09319	0.01739	0.30666
tdec3	160	0.16275	0.09616	0.01739	0.31339
captr	151	0.00735	0.00476	0.00130	0.02004
gdp	160	0.02173	0.00403	0.01373	0.03075
lend	157	-0.03798	0.14311	-0.90478	0.45339
debt	159	0.72395	0.28398	0.14797	1.44307
pop	160	30953.0	28077.3	4964.00	82520.0

Table 3.14: Summary Statistics

 Table 3.15:
 Multi-co-linearity

	tax	cap	gdp	lend	debt	pop
tax	1					
cap	-0.1664	1				
gdp	0.6460	-0.1708	1			
lend	-0.1163	0.0847	0.2037	1		
debt	0.0762	0.0305	-0.2045	-0.1421	1	
pop	-0.4374	0.1374	-0.3317	0.1210	0.1928	1

	Effect	Test	Statistic	P-value		Effect	Test	Statistic	P-value	
1.INF (level)	ft	LLC	-2.8633	0.002	***	f	LLC	-0.6358	0.263	
		IPS	0.0706	0.528			IPS	0.16646	0.566	
$2.HS \ (level)$	$_{\rm ft}$	LLC	-4.472	0.000	***	f	LLC	-4.66191	0	***
		IPS	-0.8755	0.191			IPS	-2.11709	0.017	**
$3.PG \ (level)$	$_{\rm ft}$	LLC	-3.7699	0.000	***	f	LLC	-3.68242	0.000	***
		IPS	0.0221	0.509			IPS	-2.28425	0.011	**
4.RED (level)	$_{\rm ft}$	LLC	-1.6212	0.053	*	f	LLC	-3.16266	0.001	***
		IPS	-0.5546	0.290			IPS	-1.0984	0.136	
Lend	$_{\rm ft}$	LLC	-2.0155	0.022	*	f	LLC	-3.98932	0.000	***
		IPS	-1.2798	0.100			IPS	-1.04089	0.149	
Debt	ft	LLC	-10.593	0.000	***	f	LLC	-0.70526	0.240	
		IPS	-2.6409	0.004	***		IPS	0.58686	0.721	
Cap	$_{\rm ft}$	LLC	-6.4297	0.000	***	f	LLC	-2.80209	0.003	***
		IPS	-2.6135	0.005	***		IPS	-1.94034	0.026	**
Gdp	$_{\rm ft}$	LLC	-8.8095	0.000	***	f	LLC	2.14455	0.984	
		IPS	-4.586	0.000	***		IPS	5.18343	1	
O taxshl	$_{\rm ft}$	LLC	-2.2478	0.012	**	f	LLC	-0.39658	0.346	
		IPS	-1.8406	0.033	**		IPS	-0.03522	0.486	
Pop	ft	LLC	4.96377	0.000	***	f	LLC	0.42430	0.664	
		IPS	3.25404	0.001	***		IPS	4.39171	1.000	

Table 3.16: Unit Root Tests

Tests conducted according to the lag length indicated by Akaike (AIC) information criteria. Asterisks ***, **, and * denote significance at 1 percent, 5 percent and 10 percent level, respectively. f denotes fixed effects and individual effects, denotes time trends.

Table 3.17: Q-test of Autocorrelation for Dependent Variable

	AC	Q-Stat	Prob
1.INF	0.886	107.56	0.000
2.HS	0.844	97.511	0.000
3.PG	0.810	88.663	0.000
4.RED	0.854	99.156	0.000

Chapter 4

Privatization and Public Social Spending

4.1 Introduction

Privatization is often assumed to affect public revenue. However, there is only little literature analyzing the link between privatization and the expenditure of the government. Nevertheless, Sheshinski et all (2003) argue that one objective of privatization programs is to free resources for allocation in other government activities - e.g. related to social policy. It is the objective of this paper to better understand this link between privatization and the composition of public expenditure.

In particular, we are interested in the link between privatization and the level of public social spending. By privatization we mean a reallocation of control rights over firms from a public towards a private decision maker. Thereby, it may also be interpreted as transfer of risk from the public sector towards individuals being exposed to a higher probability of becoming unemployed. Moreover, private firms may provide less social security services than public firms do. In order to outbalance these additional risks, individuals will adjust their demand for social protection by the government. The suggestion that voters require the government to provide insurance is intuitive, given that it is the government's privatization policy that causes the higher risk. Thus, there

are good arguments why privatization should generate higher public social spending. Our theoretical model explains how privatization affects social public spending. Based on Kanbur (1981) and Haskel and Szymanski (1993) we assume two sectors, a private and a public with the risk of becoming unemployed being higher in the private sector. An exogenously given increase of privatization then leads to a higher expected rate of unemployment, and at the same time higher productivity. We investigate how privatization affects the per capita transfer as well as overall redistribution if transfers are financed only through the profits of the public firms or through public profits and a lump-sum tax. Our results suggest that overall redistribution is likely to increase with privatization while per capita transfers decrease if redistribution is financed only through profits of public firms. On the other hand, if lump-sum taxes are available to finance redistribution, overall as well as per capita redistribution increase with privatization. Finally, if it is costly to raise public funds - a commonly accepted characteristic in the literature on public finance - additional distortions arise: The higher need for redistribution (and thereby higher excess burden) leads to additional inefficiencies, which partly offset the productivity gains from privatization.

Although, an empirical analysis of the positive link between privatization and social public expenditure goes beyond the scope of this paper, we provide some data to emphasize our reasoning. We present data on the 31 provinces of China - one of the most dynamic countries in terms of privatization in recent years. Figure 4.1 shows the development of privatization, measured as the share of self- and privately employed individuals over the total number of employees for large cities¹, for the rather developed Eastern Provinces, and for the mostly rural Western Provinces:

The share of employees in the private sector steadily increased from 1996 to 2006. This tendency is most pronounced in urban areas, where privatization increased from 9% to 42%. Although, the development is less pronounced in Eastern (10% to 17%) and Western provinces (5% to 12%) they still follow a clear upward trend. As expected, the level of privatization in the more developed Eastern Provinces is constantly above the level of the more rural Western Provinces. Although huge variation exists among

¹Cities with provincial status in China include Beijing, Shanghai, and Tianjin

Figure 4.1: Private Sector Employees and Self-Employed as Share of Total, by Type of Province, 1996-2006



Source: Statistical Office of China, own calculations.

provinces in China, the data suggests that the privatization process is rather dynamic throughout the country.

In contrast, Figure 4.2 shows the development of public social expenditure by category as share of overall public expenditure. Figure 4.2 presents social public expenditure by category as defined by the OECD (2007) upon availability of data.²





Source: Statistical Office of China, OECD(2007), own calculations.

It becomes evident that also the composition of public expenditure throughout this

 $^{^{2}}$ The OECD categorizes social spending into 9 fields. Date on some of them, such as "family" or "other social policy areas", are not available for China

period is characterized by notable changes: There is a clear upwards trend in the development of the three social categories of public expenditure in Figure 4.2. The share of Social Expenditure and Retirement rose from 2% to 6% and from 4% to 7.5%, respectively. Although, the development of Welfare Expenditure (including pensions for disabled and civil affairs) is less pronounced, it still increased by about 50% from 2 percentage points to 3 percentage points. After all, the data suggests that China was characterized by a very dynamic development of privatization as well as an over-proportional increase of social expenditure in the last decade.

The evidence from China is also backed by data from other countries: E.g. Erdmann (1998) investigates the development of social spending relative to GDP in Poland, Hungary and the Slovak Republic throughout the transformation period in the 1990s. Indeed she finds that although GDP decreased in the years after the system transformation, social spending as share of GDP increased for all three countries. Note as that in the case of China, these findings are unable to prove a direct link between privatization and social public spending. Though, the data does provide evidence to motivate the theoretical arguments that we develop below.

The paper is organized as follows. After a short literature review in Section 4.2, Section 4.3 presents a theoretical model to identify the positive link between privatization and social expenditure by the government. In Section 4.4 we analyze the link between privatization and redistribution if transfers are only financed through profits of public firms. In Section 4.5 we extend this setting by introducing lump-sum taxes on workers to finance redistribution. Section 4.6 also accounts for the potential costs of taxation in order to finance redistribution. Finally, Section 4.7 concludes.

4.2 Literature Review

While some macro-implications of privatization are still obscure as outlined by Sheshinski et al. (2003), other aspects have already been discussed in more detail: E.g. several authors looked at the link between privatization and overall economic growth: Barnett (2000) finds a significantly positive link, in particular for non-transition countries. He also shows that privatization goes hand in hand with fiscal consolidation. This is also confirmed by Jeronimo et al. (2000) testing this relationship with data on four southern states in the US from 1990 to 1997.

However, there is hardly any literature on the link of privatization and the composition of public expenditure. The starting point for our theoretical model is an hypothesis developed by Grossman and Hart (1983) and Laffont and Tirole (1993). They show how the relationship between principals and agents changes with privatization due to a risk transfer from the public towards the private sector. In this context, Kanbur (1981) proposes a model with two sectors, employees and entrepreneurs, where the latter face the risk of failing with their business. This allows him to determine the optimal share of workers in each sector. In a broader sense this setting could be interpreted in terms of privatization, with a higher risk of becoming unemployed in the private sector. However, with this interpretation, the basic idea of Kanbur (1981) that individuals in one sector (entrepreneurs) employ workers from the other sector (employees) is problematic. This brings us closer to Haskel and Szymanski (1993). They assume a private and a public production sector, with privatization changing the objective of a firm away from social aspects towards more profit orientation. This implies that employment declines with privatization. They also analyze the implications of privatization for wages; and identify a negative link. Their theoretical findings are also supported by the empirical evidence they provide for a sample of 14 firms in the UK.

The negative effect between privatization and employment is not without controversy. E.g. Brown et al. (2005) find that there is a positive - though small- employment effect of privatization due to its potential positive effect on efficiency. Aghion and Blanchard (1993) show that depending on the level of unemployment privatization may lead to more or less jobs. Balla et al. (2004) extend this setting by distinguishing between skilled and unskilled labor. For our approach, this latter finding implies that a higher need for redistribution can be motivated by arguments going beyond the overall unemployment rate. Indeed, unemployment among unskilled workers is likely to increase, even if privatization leads to notable efficiency gains. Balla et al. (2004) propose a subsidy towards unskilled workers to smooth this effect. However, their inter-temporal setting does not concentrate on public expenditure in the first place as we do - but mainly focus on the labor market implications of privatization.

The idea that a certain policy measure implies additional risks for individuals and hence should be accompanied by social measures has already been discussed in other contexts: E.g. Rodrik (1998) claims that the size of government is larger in open economies, because the associated risk for individuals becomes larger through more interactions with other economies. Kreider (2003) shows how income uncertainty - which may be related to the risk of becoming unemployed - affects redistribution. He shows that the degree of risk aversion is an important variable to explain the relationship between uncertainty and redistribution.

4.3 Model

Based on Haskel and Szymanski (1993) and Kanbur (1981) we set up a two sector model - with a private and a public sector. While we assume full employment in the public sector, the deterministic probability of becoming unemployed in the private sector is strictly larger zero. This is a simplified representation of the higher propensity of the state for full employment.³ Assuming a deterministic rate of unemployment in order to simplify the labor market and thereby receive explicit solutions for other variables of interest (in our case redistribution) is a common feature in public economic theory. Good examples are Kreider (2003) or Boycko et al. (1996).

We assume a productivity parameter of labor being larger in the private than in the public sector, $\gamma_{priv} > \gamma_{pub}$. This assumption is in line with the common belief that profit oriented, private firms are generally more efficient and productive. Higher productivity in the private sector is the justification, why the state should be interested in

³For instance, Haskel and Szymanski (1993) introduce a deterministic propensity of the public sector for high employment by introducing a variable A, which is strictly larger in the public sector. Our approach is similar in the sense that we assume a higher preference for employment in the public sector. However, for simplicity we don't introduce a complete labor market, but instead assume the unemployment rate in the private sector to be deterministic.

privatization in our model. The overall labor force in the economy is normalized to 1 with ρ being the share of labor employed by private firms. The government maximizes utilitarian welfare and engages in redistribution from workers towards unemployed individuals.

4.3.1 Firms

Firms only use labor as input in the production process. For simplicity we assume a linear production technology. With prices normalized to one, the profit of a private sector firm writes

$$\pi_{priv} = \rho \cdot (\gamma_{priv} - w) = 0 \tag{4.1}$$

w is the wage rate in the private sector. Given the linear production function, profits in the private sector are defined as the difference between marginal productivity and wage rate times the number of employees. However, in line with the assumptions for perfect competition (large number of firms) in the product market, $w = \gamma_{priv}$ throughout the paper. Wages in the private sector correspond to the marginal productivity of this sector; private profits are consequently equal to zero. In order to simplify our calculations, the labor market in our model is incomplete. We assume a deterministic unemployment rate in the private sector as explained above. This is also the reason, why wages can not adjust to eliminate unemployment. The idea is to capture in a simple manner the assumption that privatization leads to higher unemployment.⁴ Similarly, profits in the public sector are defined as

$$\pi_{pub} = (1 - \rho) \cdot (\gamma_{pub} - v) \tag{4.2}$$

with v as the wage rate in the public sector. The profit function of the public sector resembles that of the private sector. There is, however, one important difference: For

 $^{^4.\}mathrm{e.g.}$ see Haskel and Szymanski (1993) for a theoretical analysis as well as an empirical verification of this hypothesis.

the public sector, we do not impose any restrictions on the value of v a priori: The public sector can, but must not pay wages equal to the marginal productivity of labor - even under perfect competition. This holds since the government can outbalance positive or negative profits of public firms through its budget as it is also proposed in Shleifer and Vishny (1994). Indeed, it is often argued that public firms make losses due to excess employment or higher public wages, which then enter the budget constraint of the government. This is also the case in our setting, as we will see below.

4.3.2 Households

We distinguish among households assigned to the public and private sector, respectively. Public employees do not face any risk of becoming unemployed. On the contrary, households in the private sector are exposed to unemployment with probability (1 - p). Correspondingly, p is the probability of being employed once an individual is assigned to the private sector. Together with our assumptions from above, our model implies a deterministic unemployment rate of $\rho \cdot (1 - p)$.

We define preferences for each individual as a function V, with V' > 0 and V'' < 0, depending on net-income, only. The concavity of the utility function introduces a preference of the government for equity.⁵ Preferences in the public sector write

$$V(C_{pub}) = V(v)$$

Correspondingly, the expected utility in the private sector is

$$V(C_{priv}) = p \cdot V(w) + (1-p) \cdot V(b)$$

b is the per capita unemployment benefit provided by the government towards unemployed individuals. We assume that $p \cdot w = p \cdot \gamma_{priv} > \gamma_{pub}$. Thus, the expected wage of an individual in the private sector is larger than its deterministic efficiency in the

⁵For the moment no income tax does exist. This assumption will be relaxed later on.

public sector. Or, to put it differently, privatization leads to an overall increase in expected income since the expected productivity of an employee in the private sector exceeds the productivity in the public sector.⁶

4.3.3 State Sector

The state maximizes utilitarian welfare with respect to the individual unemployment benefit, b, subject to its budget constraint. In the following we assume a logarithmic utility function for all individuals - a functional form satisfying the concavity conditions from above. The utilitarian welfare function reads

$$W = (1 - \rho) \cdot \ln(v) + \rho \cdot (p \cdot \ln(w) + (1 - p) \cdot \ln(b))$$
(4.3)

 ρ may be interpreted as a measure for privatization. For the moment, the state does not levy a lump-sum tax to finance individual unemployment benefits, b. Instead, profits of public firms are the only source of income to finance redistribution. Consequently, the government has no tax income to finance redistribution. This assumption is critical for two reasons: First in most countries, redistribution is at least partly financed by taxes on wages. Second, while the number of unemployed individuals steadily increases with ρ , the public sector and thereby the number of contributors to the redistribution systems shrinks. In a completely privatized economy the number of persons financing redistribution converges to zero requiring other means to finance redistribution. Usually, unemployment benefits are at least partly financed through income taxes. For the moment, we concentrate on public profits in order to isolate the link between the public budget and privatization and how it affects the utility of individuals in the public and private sector. Section 4.5 will relax this assumption and introduce taxes to co-finance unemployment benefits. In the present setting, the

⁶This assumption is a lower bound for $\gamma_{priv} > \gamma_{pub}$.

budget constraint of the state writes:

$$(1-\rho) \cdot (\gamma_{pub} - v) = \rho \cdot (1-p) \cdot b \tag{4.4}$$

Thus, the profits in the public sector equal its social expenditure for redistribution. This implies that in the case of positive redistribution transfers b, $\gamma_{pub} > v$ must hold. Since $w = \gamma_{priv} > \gamma_{pub}$, the wage paid in the public sector must therefore be well below the wage in the private sector.⁷

The public budget constraint unambiguously determines the wage rate in the public sector:

$$v = \gamma_{pub} - \frac{\rho}{1-\rho} \cdot (1-p) \cdot b.$$
(4.5)

It increases in γ_{pub} and p, and decreases in b and ρ : If the marginal productivity in the public sector γ_{pub} increases, we can see from Equation (4.2) that the public firm's balance increases and therefore the public sector has more funds available. This allows for higher public wages. On the other hand, a higher level of per capita redistribution b requires more public spending. The only way to finance b is to reduce wages in the public sector. Moreover, v decreases in ρ : More privatization yields higher expected unemployment and therefore a higher need of public funds, which can only be generated through a reduction in v. Note also that a higher probability of employment p in the private sector reduces the need for redistribution. The production of public firms can then be used to a larger extent for compensating its employees; v increases.

⁷This result contrasts the findings in Haskel and Szymanski (1993). They determine a wage rate being higher in the public than in the private sector. However, their finding does not account for higher efficiency in the private sector or the financing of redistribution through public profits as we do. Instead, what drives their results is the distinction between different welfare weights for unions and consumer surplus, respectively.

4.4 Result without Lump-Sum Tax

The government redistributes from workers towards unemployed individuals, which is welfare improving due to its utilitarian welfare objective and the concavity of consumers' preferences. To asses the optimal per capita unemployment transfer, the government maximizes Equation (4.3) wrt. b subject to its budget constraint in Equation (4.4). This yields the following first order condition:

$$\frac{\partial W}{\partial b} = (1-\rho) \cdot \frac{1}{v} \cdot \frac{\partial v}{\partial b} + \rho \cdot (1-p) \cdot \frac{1}{b} = 0$$

It states that the negative impact of a higher b on public wages (first term) needs to be exactly outbalanced by the benefits of a larger b for the unemployed (second term). Since we deal with utilitarian welfare, the two opposing effects are weighted by the marginal utility times the number of persons concerned. Solving for b yields Proposition 4.1.

Proposition 4.1:

i) The optimal redistribution rate is characterized by $b = \gamma_{pub} \cdot \frac{1-\rho}{1-\rho \cdot p}$.

ii) Higher privatization implies a decrease in the optimal redistribution rate b and an increase in overall redistribution $(b \cdot \rho \cdot (1-p))$ if and only if $p > \frac{2 \cdot \rho - 1}{\rho^2}$.

iii) Redistribution from public workers to unemployed individuals is perfect, in particular w > v = b.

Proof see Appendix 4.8. ■.

The optimal redistribution rate depends on ρ , γ_{pub} , and p: As expected, b increases in γ_{pub} . Higher productivity in the public sector makes more public funds available, which can be distributed among public workers and unemployed. Intuitively, b also increases in p: A higher probability of being employed in the private sector requires less overall redistribution. The reduced overall requirement for public spending is then transferred partly to workers in the public sector through higher wages and partly to unemployed

through higher per-capita transfers in order to equalize marginal utilities.

The second part of Proposition 4.1 claims that privatization leads to lower per-capita redistribution b. If public firms are privatized, more people face the exogenously given risk of becoming unemployed, (1-p). Further, privatization leads - by definition - to a decrease in the number of public workers (and in our case the number of contributors to the redistribution system). If the number of expected transfer receivers increases at the cost of the number of contributors, individual unemployment benefits need to decrease. Otherwise, the tax burden for contributors would be too high to restore the optimality condition for redistribution of equal marginal utilities.

Furthermore, overall redistribution, defined as per capita redistribution times the expected number of unemployed, increases with privatization if $p > \frac{2 \cdot \rho - 1}{\rho^2}$. Privatization increases the expected number of unemployed persons, implying a higher need for redistribution (direct effect). On the other hand, we have seen that per capita redistribution transfers decrease in ρ (indirect effect): Due to the assumed utilitarian welfare objective and the functional form of individual preferences, the government wants to distribute the cost of additional unemployment evenly throughout the society. This has a negative, indirect impact on overall redistribution. Which of these two effects prevails depends on the parameter values of p and ρ determining the share of unemployed individuals in the economy. To see this, Figure 4.3 displays the graph of $p = \frac{2 \cdot \rho - 1}{\rho^2}$, that is the condition determining the sign of the derivative of overall redistribution with respect to privatization.

In an economy with high expected unemployment (ρ high and p low) overall redistribution decreases in privatization, meaning that the indirect effect prevails: With a large number of unemployed, the reduction in b for each transfer receiver has a larger aggregate effect on redistribution than the marginal increase in transfer receivers caused by privatization. On the other hand, if the expected unemployment rate is low (ρ low and p high), the direct effect prevails and overall redistribution increases in privatization: The number of public workers is large relative to the number of unemployed. Therefore, most of the cost of additional unemployment will be financed through lower public wages rather than the reduction in b. Privatization leads then to an increase in

Figure 4.3: Sign of Derivative of Overall Redistribution wrt. ρ



Source: own calculations.

overall redistribution. Loosely speaking, if unemployment is low, the government can generously increase overall redistribution, since this cost is born by a relatively large number of public workers. This is not possible if unemployment is high.

Part three of Proposition 4.1 says that while the income of private employees exceed the income of public workers and unemployed, redistribution between public workers and unemployed is perfect: Since the financing of redistribution through public profits does not imply any loss whatsoever, the government can perfectly redistribute among public workers and unemployed individuals. Thereby, it achieves its first best objective of identical marginal utilities for these two groups. On the other hand, absent taxes, there are no feedback effects of redistribution on private workers - they are simply not involved in redistribution. Since private wages correspond to the higher productivity in this sector, the utility of private workers exceeds the utility of both public workers and unemployed individuals. Thus, the instrument to finance redistribution is incomplete because it does not involve private workers. The government can therefore achieve its objective to equalize marginal utilities only for two out of three consumer groups.

4.5 Result with Lump-Sum Tax

In Section 4.4 public profits have been the only means to finance redistribution. As we have seen this assumption is critical. In this section we therefore introduce an additional tool to raise public funds: lump-sum taxes on public and private workers. This enables the government to also involve private workers in financing unemployment benefits.

With lump sum taxes on workers, the objective function of the government writes.

$$max_{\tau,b}W = (1-\rho) \cdot ln (v-\tau) + \rho \cdot (p \cdot ln (w-\tau) + (1-p) \cdot ln (b))$$
(4.6)

Correspondingly, its budget constraint reads

$$(1 - \rho) \cdot (\gamma_{pub} - v) + (1 - \rho + \rho \cdot p) \cdot \tau = \rho \cdot (1 - p) \cdot b$$

Besides public profits, the lump sum tax on both public workers $(1 - \rho)$ and private workers $(\rho \cdot p)$ is available to cover social expenditure. The budget constraint unambiguously determines the wage rate in the public sector:

$$v = \gamma - \frac{1}{1 - \rho} \cdot \left((1 - p) \cdot \rho \cdot b - (1 - \rho + \rho \cdot p) \cdot \tau \right)$$

As before, the public wage rate decreases in b and ρ , and increases in p and in the productivity of the public sector. New is the positive impact of τ on v: τ is an additional instrument to finance redistribution, which (differently from public profits) also includes private workers. Thus, while τ does put a burden on workers in the public sector, it also implies a reduced need to generate public profits in order to cover unemployment benefits. This latter effect has a positive impact on v, since τ assures that the cost of redistribution is no longer covered by public workers, only. The linear budget constraint assures that public wages increase faster than the tax rate does, implying an overall increase in the net income of public workers. Using the budget constraint and deriving Equation (4.6) with respect to τ and b yields the following first

order conditions:

$$\frac{\partial W}{\partial \tau} = (1-\rho) \cdot \frac{1}{v-\tau} \cdot \left(\frac{\partial v}{\partial \tau} - 1\right) + \rho \cdot p \cdot \frac{1}{w-\tau} \cdot (-1) = 0$$

and

$$\frac{\partial W}{\partial b} = (1-\rho) \cdot \frac{1}{v-\tau} \cdot \frac{\partial v}{\partial b} + \rho \cdot (1-p) \cdot \frac{1}{b} = 0$$

The first order condition wrt. b is similar to the one in Section 4.4 with the only

difference that now incomes net of lump-sum taxes matter. The optimality condition for τ requires that at the margin the positive net effect of higher lump-sum taxes for public workers equals the negative income effect for private employees. This is again weighted by the size and marginal utilities of the two groups. Solving for τ and bresults in

Proposition 4.2:

i) The optimal redistribution rate is characterized by $b = \rho \cdot p \cdot w + (1 - \rho) \cdot \gamma_{pub}$, the optimal lump-sum tax by $\tau = w - \rho \cdot p \cdot w - (1 - \rho) \cdot \gamma_{pub}$.

- ii) Privatization implies an increase in b and in overall redistribution $(b \cdot \rho \cdot (1-p))$.
- iii) Redistribution is perfect, in particular: $w \tau = v \tau = b$.

Proof see Appendix 4.8. \blacksquare .

The optimal redistribution transfer corresponds to the weighted sum of the productivity levels in the two sectors, with the weights corresponding to the share of workers in each sector. Since the number of individuals is normalized to one, this is equivalent to the per-capita productivity in the economy. γ_{pub} and now also $\gamma_{priv} = w$ have a positive impact on the optimal choice of b.

The characteristics of the optimal τ yield some interesting insights in the functioning of the model. First, there is a positive relation between τ and w: With a higher productivity in the private sector, it is welfare improving to shift some cost of redistribution towards this sector: The only way to reach this goal is to increase τ . Second, higher productivity in the public sector has a negative impact on τ . The intuition behind this somewhat surprising result is that higher productivity ceteris paribus increases public profits. Thus, less funds need to be acquired through taxation to finance a given level of redistribution. Moreover, the derivative of the optimal tax rate with respect to privatization, $\frac{\partial \tau}{\partial \rho} = -p \cdot w + \gamma_{pub}$, is negative: Since (differently from Section 4.4) public profits are negative, redistribution occurs also from private workers towards public workers. In a highly privatized economy, the most productive group in the economy is relatively large. Thus, the per capita burden of redistribution for this group can be relaxed corresponding to a reduction in τ .

The impact of privatization on per capita redistribution is now positive contrasting the result from Section 4.4. The interpretation is as follows: By assumption, the expected productivity in the private sector exceeds the productivity in the public sector. This implies that an increase in ρ leads to a higher average productivity in the economy allowing for higher per capita transfers towards the unemployed. In Section 4.4 this mechanism did not work. There was no means to tax workers in the more productive sector. The number of contributors decreased one to one with privatization. This also explains why our result from above that overall redistribution increases with privatization is more clear-cut in this section: Privatization leads to a higher expected productivity in the economy and at the same time higher unemployment - both speaking in favor of higher overall redistribution.

The government perfectly insures individuals against the risk of becoming unemployed in the private sector by equalizing marginal utilities beyond all three states (public and private employment as well as unemployment). Indeed, given the risk aversion and identical utility functions of individuals in each state, the best a government can do is to equalize the income of individuals beyond states. The government disposes over two instruments to reach this goal (public profits and lump sum taxes). None of them entails any additional cost. Thus the government can equalize marginal utilities without introducing additional distortions. This is realized in two steps: First, it sets the redistribution transfer b such that it equals net income in the private sector. Second, with public profits the government has a second tool to assure that the net income in the private and public sector are equal. Since v < w and taxes are imposed in lump-sum form on all workers, public profits must be negative. Thus, the public wage exceeds the productivity of workers in this sector. This result contrasts the result from Section 4.4 where public profits have been positive in order to finance redistribution.

The results in Sections 4.4 and 4.5 can be interpreted in a more policy relevant context. First of all, privatization appears to be an important factor to understand what determines the composition of public spending. Our model suggests that privatization leads to additional risks and thereby a higher demand for social spending of the government. It is therefore not enough to evaluate the efficiency gains of a privatization reform in order to assess its overall performance. Rather, it is necessary to also account for potential shifts in the demand for public goods. While privatization implies less state control in one respect, it leads to higher state intervention in other fields undermining one of the most important aspects of privatization reform, it is imperative to also understand its implications for the level and composition of public spending. This finding is particularly important for transition economies characterized by huge privatization efforts.

4.6 Excess Burden of Taxation

As we have seen, privatization is likely to require higher social public spending and therefore public funds. If these funds are financed through taxation, another undesired effect may arise: Excess burden of taxation. Indeed, it is often argued that raising one unit of public funds generates a tax burden larger than one (Laffont and Tirole (1993)). The most immediate interpretation of this finding is a distortion through taxation. With lump-sum taxes - as assumed throughout this chapter - such distortions do not arise. However, with the approach by Laffont and Tirole (1993) they can be easily implemented without loosing the mathematical simplicity that lump-sum taxes offer. Thus, our discussion from above is incomplete because we disregard the fact that privatization generates an additional negative effect through its higher need for social spending. For this purpose, we extend the model from Section 4.5 by introducing an excess burden of taxation.

Compared to Section 4.5 the welfare function of the government remains unchanged (see Equation (4.6)). On the other hand, the budget constraint of the government rewrites

$$(1-\rho)\cdot(\gamma-v) + (1-\rho+\rho\cdot p)\cdot\tau\cdot(1-\lambda) = \rho\cdot(1-p)\cdot b$$

with λ representing the excess burden. A share λ of each unit of tax-funds raised is omitted from the equation due to the excess burden of taxation.

Maximizing the welfare function of the government wrt. b and τ and considering its new budget constraint yields

$$b = \frac{\gamma \cdot (1 - \rho) + (\rho \cdot p - (1 - \rho + \rho \cdot p) \cdot \lambda) \cdot \tau}{1 - \rho \cdot p}$$

and

$$\tau = \frac{w \cdot \frac{\rho \cdot p - (1 - \rho + \rho \cdot p) \cdot \lambda}{p \cdot \rho} - \gamma + \frac{1}{1 - \rho} \cdot (1 - p) \cdot \rho \cdot b}{\frac{\rho \cdot p - (1 - \rho + \rho \cdot p) \cdot \lambda}{p \cdot \rho} + \frac{1 - \lambda}{1 - \rho} \cdot (1 - \rho + \rho \cdot p) - 1}.$$

Merging these two expressions leads to the explicit solution of the model and Proposition 4.3:

Proposition 4.3:

- i) Excess burden λ leads to a decrease in b and an increase in τ .
- ii) Redistribution is no longer perfect, in particular: $w \tau > v \tau = b$, if $\gamma \cdot (1 \rho) + w \cdot \rho \cdot p > (1 \rho + \rho \cdot p) \cdot \lambda$.

Proof see Appendix 4.8. ■.

Introducing λ notably complicates the explicit solution for b and τ . However, we do fall back to results in Section 4.5 for $\lambda = 0$. As comparative statics show, λ implies a decrease in b and an increase in τ : If the cost of raising one unit of additional tax income increases, it is no longer optimal to maintain the unemployment benefits from Section 4.5: Exploiting the advantage of lump-sum taxes (redistributing among three instead of two consumer groups) now entails an excess burden. The government wants to evenly distribute this cost among all three groups of consumers. The only way to reach this goal is to increase the tax rate for tax payers (that is the workers in the private and the public sector) and at the same time reduce the redistribution rate b. The reduction in b assures that also unemployed individuals participate in financing the additional cost of taxation.

This has considerable consequences for the overall level of utility of each of the three groups of consumers. Evidently, the utility must decreases for all three groups if τ increases and b decreases as described above. However, a-priori it is not clear, whether utility levels are still equal as in Section 4.5 or not. To address this issue, note that the marginal cost of raising one unit of redistributions funds is larger than one due to λ . Therefore, the government will no longer redistribute until complete equity is restored. At one point the difference in marginal utilities will be lower than the cost of redistribution, λ . The government will choose redistribution below first best implying a net income for private workers above the transfer towards unemployed: $w - \tau > b$. Moreover, financing redistribution through public profits allows evading the additional cost of taxation. At the margin it is profitable to reduce public wages slightly in order to forgo λ . This implies that differently from Section 4.5, v < w.⁸ Note that public profits introduce another form of inequality since it excludes private workers from financing unemployment benefits. Therefore, a τ strictly larger zero is still desirable. Since this does not affect the optimal redistribution between unemployed individuals and public

⁸Note that v < w holds for $\gamma \cdot (1 - \rho) + w \cdot \rho \cdot p > (1 - \rho + \rho \cdot p) \cdot w \cdot \lambda$. This implies that the excess burden must be small enough. In particular this condition says that the actual production needs to exceed the maximal possible excess burden in the economy. Indeed, we have seen in Section 4.5 that v increases in τ and we know that a large λ requires high taxes.
workers, the government implements identical utility levels for both unemployed and public workers similar to Section 4.4. In consequence, net wages in the public sector equal redistribution transfers. Thus, overall, $w - \tau > v - \tau = b$. This result is due to the fact that more general instrument to finance redistribution (τ) implies additional costs, whereas the instrument to redistribute between public workers and unemployed (public profits) does not.

The findings in this section allow for interesting policy insights. Privatization may lead to further, indirect distortions: If privatizations increases the need for public social funds, the well known excess burden of taxation may increase and at least partly offset the efficiency gains generated by privatization: Although privatization may lead to higher overall productivity in the economy, it also requires additional redistribution as this paper shows. Additional redistribution, however, may be costly and offset some of the gains from privatization. Thus, in order to correctly evaluate a privatization policy, potential indirect (and undesired) effects need to be taken into account. Thinking further, one could also imagine that the higher need for social spending we found in this paper could imply lower public spending in other sectors (e.g. public investment) rather than higher taxes. If, for instance, the government reduces public investment to finance social spending, the productivity of the private sector may go down.⁹ This may outbalance most of the efficiency gains from privatization. After all, this section suggests that for a thorough understanding of privatization, its potential indirect distortions matter.

4.7 Conclusion

Most of the literature on privatization focuses on its micro-economic implications. On the other hand, only few studies exist, which examine its link to macro-variables. In particular it is still unclear, how privatization affects the composition of public

⁹There is a huge literature on the functioning of public investment as a public input. See e.g. Keen and Marchand (1997) for more details.

spending. It is the aim of this paper to contribute to a better understanding of this link.

This paper investigates the impact of privatization on the composition of public expenditure. If the likelihood of becoming unemployed is higher in the private sector, privatization can be interpreted as transfer of risk from the public sector towards individuals. Consequently, individuals demand additional social protection (redistribution) by the government to insure against this risk. Our model shows that this holds under rather mild restrictions, namely that the number of workers in the economy is large enough relative to the unemployed. Further, per capita unemployment benefits do decrease if redistribution is financed by public profits, only: With privatization the number of contributors (public workers) steadily decreases and at the same time the number of transfer receivers (unemployed) increases. This conflict can only be resolved through a reduction in per capita unemployment benefits. On the other hand, results change if the government disposes over lump sum taxes for private and public workers as additional tool to finance redistribution: Per capita transfers then increase with privatization since the higher productivity of an increasing number of private workers is taken into account to determine the optimal redistribution rate. Finally, we also consider excess burden of taxation: If it is costly to raise money for public funds, privatization entails undesired distortions through a higher need of public social spending. In this case, the government is no longer willing to implement optimal redistribution.

The results in this paper allow for several policy insights: Privatization may lead to additional risks, in particular a higher probability of becoming unemployed. This in turn is likely to affect the demand and therefore also level and composition of public expenditure. E.g. while privatization implies less state control in one respect it leads to higher state intervention in other fields undermining one of the most important aspects of privatization efforts. Moreover, we have seen that privatization may lead to additional inefficiencies in the economy: E.g. if more public funds are required to finance the additional social expenditure, the commonly known excess burden of taxation e.g. administrative costs - may arise and at least partly offset the productivity gains generated by privatization. Thus, in order to properly evaluate a specific privatization reform, one needs to account for its potential indirect distortions in other respects.

Our analysis shows that future research in this direction is highly valuable, in particular, as regards the empirically verification of the link between privatization and redistribution. Further, it might also be interesting to endogenize privatization in order to better understand the tradeoff between the potential efficiency gains through privatization and the potential distortions in other sectors. Finally, a thorough analysis of the link between privatization and different types of public spending could be highly valuable: For instance, higher public social spending may lead to a reduction in other types of public expenditures such as public investment. If public investment serves as public input factor for private firms, this adjustment in the composition of public spending would partly offset the potential productivity gains from privatization.

4.8 Appendix

Proof of Proposition 4.1:

Proof of i):

With the logarithmic utility function, the first order condition wrt. b writes

 $\tfrac{\partial W}{\partial b} = (1-\rho) \cdot \tfrac{1}{v} \cdot \tfrac{\partial v}{\partial b} + \rho \cdot (1-p) \cdot \tfrac{1}{b} = 0.$

Using the budget constraint, this simplifies to

$$b = \gamma_{pub} \cdot \frac{1-\rho}{1-\rho \cdot p}$$

Proof of ii):

In order to determine the effect of privatization on redistribution transfers, derive b wrt. ρ :

$$\frac{\partial b}{\partial \rho} = \gamma_{pub} \cdot \frac{-(1-\rho \cdot p) - (1-\rho) \cdot (-p)}{(1-\rho \cdot p)^2} = \gamma_{pub} \cdot \frac{-1+\rho \cdot p + p + \rho \cdot (-p)}{(1-\rho \cdot p)^2} = \gamma_{pub} \cdot \frac{-(1-p)}{(1-\rho \cdot p)^2} < 0$$

Overall redistribution is defined by $b \cdot \rho \cdot (1 - p)$ (per capita transfer times number of unemployed). Using the explicit solution of b and deriving wrt. ρ yields

$$\frac{\partial (b \cdot \rho \cdot (1-p))}{\partial \rho} = b \cdot (1-p) + \frac{\partial b}{\partial \rho} \cdot \rho \cdot (1-p) = \gamma_{pub} \cdot \frac{1-\rho}{1-\rho \cdot p} \cdot (1-p) + \gamma_{pub} \cdot \frac{-(1-p)}{(1-\rho \cdot p)^2} \cdot \rho \cdot (1-p) = \gamma_{pub} \cdot \frac{1}{1-\rho \cdot p} \cdot (1-p) \cdot \left((1-\rho) + \frac{-(1-p)}{(1-\rho \cdot p)} \cdot \rho \right)$$

In order to prove that overall redistribution unambiguously increases with privatization, this expression needs to be larger than 0:

$$\gamma_{pub} \cdot \frac{1}{1-\rho \cdot p} \cdot (1-p) \cdot \left((1-\rho) + \frac{-(1-p)}{(1-\rho \cdot p)} \cdot \rho \right) > 0$$

or

 $1 - 2 \cdot \rho + \rho^2 \cdot p > 0.$

This yields

 $p> \tfrac{2\cdot\rho-1}{\rho^2}.$

The graph of this function is displayed in Figure (4.3).

Proof of iii):

Show that v = b:

$$v = \gamma_{pub} - \frac{\rho}{1-\rho} \cdot (1-p) \cdot b = b$$

or

$$\gamma_{pub} - \frac{\rho}{1-\rho} \cdot (1-p) \cdot \gamma_{pub} \cdot \frac{1-\rho}{1-\rho \cdot p} = \gamma_{pub} \cdot \frac{1-\rho}{1-\rho \cdot p}.$$

This simplifies to

$$1 - \rho = 1 - \rho$$

qed.

Show that w > v:

$$w > v = \gamma_{pub} - \frac{\rho}{1-\rho} \cdot (1-p) \cdot b$$

or

$$w > \gamma_{pub} - \frac{\rho}{1-\rho} \cdot (1-p) \cdot \gamma_{pub} \cdot \frac{1-\rho}{1-\rho \cdot p}$$

This simplifies to

$$w > \gamma_{pub} \cdot \left(1 - \frac{\rho \cdot (1-p)}{1-\rho \cdot p}\right).$$

Since by definition $w > \gamma_{pub}$ and $1 > \frac{\rho \cdot (1-p)}{1-\rho \cdot p} > 0$ this is always true.

Proof of Proposition 4.2:

Proof of i):

With the logarithmic utility function, the first order condition wrt. τ reads

$$(1-\rho) \cdot \frac{1}{\gamma_{pub} - \frac{1}{1-\rho} \cdot ((1-p) \cdot \rho \cdot b - (1-\rho+\rho \cdot p) \cdot \tau) - \tau} \cdot \left(\frac{1}{1-\rho} \cdot (1-\rho+\rho \cdot p) - 1\right) - \rho \cdot p \cdot \frac{1}{w-\tau} = 0.$$

This can be transformed into

$$\tau = \frac{(1-\rho)\cdot(w-\gamma_{pub})+\rho\cdot(1-p)\cdot b}{1-\rho+\rho\cdot p}.$$

The first order condition wrt. b reads

$$(1-\rho) \cdot \frac{1}{\gamma_{pub} - \frac{1}{1-\rho} \cdot ((1-p) \cdot \rho \cdot b - (1-\rho+\rho \cdot p) \cdot \tau) - \tau} \cdot \frac{\rho}{1-\rho} \cdot (1-p) - \rho \cdot (1-p) \cdot \frac{1}{b} = 0,$$

which yields

$$b = \frac{\gamma_{pub} \cdot (1-\rho) + \rho \cdot p \cdot \tau}{1 - \rho \cdot p}.$$

Substituting b and τ respectively yields the result:

$$\tau = w - \rho \cdot p \cdot w - (1 - \rho) \cdot \gamma_{pub}$$

and

 $b = \rho \cdot p \cdot w + (1 - \rho) \cdot \gamma_{pub}.$



Proof of ii):

Deriving the per capita redistribution transfer b wrt. ρ yields

$$\frac{\partial b}{\partial \rho} = (p-1) \cdot \gamma_{pub} + p \cdot (w - \gamma_{pub}) = p \cdot w - \gamma_{pub} > 0 \text{ (by definition)}.$$

Deriving overall redistribution $b \cdot \rho \cdot (1-p)$ wrt. ρ yields

$$\frac{\partial (b \cdot \rho \cdot (1-p))}{\partial \rho} = b \cdot (1-p) + \frac{\partial b}{\partial \rho} \cdot \rho \cdot (1-p) =$$
$$= (1-p) \cdot ((1-\rho+\rho \cdot p) \cdot \gamma_{pub} + \rho \cdot p \cdot (w-\gamma_{pub}) + (p \cdot w - \gamma_{pub}) \cdot \rho) > 0.$$

Proof of iii):

Rewriting τ yields:

$$\tau = w - \rho \cdot p \cdot w - (1 - \rho) \cdot \gamma_{pub} = w - b$$
. Thus, $w - \tau = b$.

Since the lump sum tax is equal for workers in the public and private sector, it still needs to be shown that v = w: Plugging b and τ into the budget constraint of the government yields

$$v = \gamma_{pub} - \frac{1}{1-\rho} \cdot \left((1-p) \cdot \rho \cdot b - (1-\rho+\rho \cdot p) \cdot \tau \right)$$

With the explicit solutions for b and τ , this yields

 $v = \gamma_{pub} - \frac{\rho \cdot p \cdot w + (1-\rho) \cdot \gamma_{pub} - (1-\rho+\rho \cdot p) \cdot w}{1-\rho}$

or

v = w

Proof of Proposition 4.3:

Proof of i):

With the logarithmic utility function, the first order condition wrt. τ reads

$$\frac{\partial W}{\partial \tau} = (1-\rho) \cdot \frac{1}{v-\tau} \cdot \left(\frac{\partial v}{\partial \tau} - 1\right) + \rho \cdot p \cdot \frac{1}{w-\tau} \cdot (-1) = 0$$

or

$$\frac{1}{\gamma_{pub} - \frac{1}{1-\rho} \cdot ((1-p) \cdot \rho \cdot b - (1-\rho+\rho \cdot p) \cdot \tau \cdot (1-\lambda)) - \tau} \cdot \left(\left(1 - \rho + \rho \cdot p\right) \cdot \left(1 - \lambda\right) - \left(1 - \rho\right) \right) = p \cdot \rho \cdot \frac{1}{w - \tau}.$$

Solving for τ yields

$$\tau = \frac{w \cdot \frac{\rho \cdot p - (1 - \rho + \rho \cdot p) \cdot \lambda}{p \cdot \rho} - \gamma_{pub} + \frac{1}{1 - \rho} \cdot (1 - p) \cdot \rho \cdot b}{(1 - \rho + \rho \cdot p) \cdot \left(\frac{1 - \lambda}{1 - \rho} - \frac{\lambda}{p \cdot \rho}\right)}.$$

The first order condition wrt. b reads

$$F = \frac{\partial W}{\partial b} = (1 - \rho) \cdot \frac{1}{v - \tau} \cdot \frac{\partial v}{\partial b} + \rho \cdot (1 - p) \cdot \frac{1}{b} = 0$$

or

$$(1-\rho)\cdot\frac{1}{\gamma_{pub}-\frac{1}{1-\rho}\cdot((1-p)\cdot\rho\cdot b-(1-\rho+\rho\cdot p)\cdot\tau\cdot(1-\lambda))-\tau}\cdot\frac{\rho}{1-\rho}\cdot(1-p)=\rho\cdot(1-p)\cdot\frac{1}{b}.$$

Solving for b yields

$$b = \frac{\gamma_{pub} \cdot (1-\rho) + (\rho \cdot p - (1-\rho+\rho \cdot p) \cdot \lambda) \cdot \tau}{1-\rho \cdot p}.$$

Substituting b and τ respectively yields the result:

$$\tau = \frac{w \cdot \frac{\rho \cdot p - (1 - \rho + \rho \cdot p) \cdot \lambda}{p \cdot \rho} - \gamma_{pub} + (1 - p) \cdot \rho \cdot \frac{\gamma_{pub}}{1 - \rho \cdot p}}{(1 - \rho + \rho \cdot p) \cdot \left(\frac{1 - \lambda}{1 - \rho} - \frac{\lambda}{p \cdot \rho}\right) - \left(\frac{1}{1 - \rho} \cdot (1 - p) \cdot \rho \cdot \frac{(\rho \cdot p - (1 - \rho + \rho \cdot p) \cdot \lambda)}{1 - \rho \cdot p}\right)}{and}$$

$$b = \frac{\left(\frac{1-\lambda}{1-\rho} - \frac{\lambda}{p\cdot\rho}\right) \cdot \gamma_{pub} \cdot (1-\rho) + \left(\frac{\rho \cdot p}{1-\rho+\rho \cdot p} - \lambda\right) \cdot \left(w \cdot \left(1 - \frac{(1-\rho+\rho \cdot p) \cdot \lambda}{p\cdot\rho}\right) - \gamma_{pub}\right)}{(1-\rho \cdot p) \cdot \left(\frac{1-\lambda}{1-\rho} - \frac{\lambda}{p\cdot\rho}\right) - \left(\frac{(1-p) \cdot \rho}{1-\rho} \cdot \left(\frac{\rho \cdot p}{1-\rho+\rho \cdot p} - \lambda\right)\right)}$$

It still needs to be shown that the derivative of the per capita transfer b wrt. λ is smaller than zero, thus:

$$\frac{\partial b}{\partial \lambda} = \frac{\left((1-\rho \cdot p) \cdot \left(\frac{1-\lambda}{1-\rho} - \frac{\lambda}{p \cdot \rho}\right) - \left(\frac{(1-p) \cdot \rho}{1-\rho} \cdot \left(\frac{\rho \cdot p}{1-\rho+\rho \cdot p} - \lambda\right)\right)\right) \cdot \left(-\gamma_{pub} - \frac{(1-\rho) \cdot \gamma_{pub}}{\rho \cdot p} - w \cdot \left(1 - \frac{(1-\rho+\rho \cdot p) \cdot \lambda}{p \cdot \rho}\right) + \gamma_{pub} - \left(w \cdot \left(1 - \frac{(1-\rho+\rho \cdot p) \cdot \lambda}{p \cdot \rho}\right)\right)\right)}{\left((1-\rho \cdot p) \cdot \left(\frac{1-\lambda}{1-\rho} - \frac{\lambda}{p \cdot \rho}\right) - \left(\frac{(1-p) \cdot \rho}{1-\rho} \cdot \left(\frac{\rho \cdot p}{1-\rho+\rho \cdot p} - \lambda\right)\right)\right)^2}{\left((1-\rho \cdot p) \cdot \left(\frac{1-\lambda}{1-\rho} - \frac{\lambda}{p \cdot \rho}\right) - \left(\frac{(1-p) \cdot \rho}{1-\rho+\rho \cdot p} - \lambda\right) \cdot \left(w \cdot \left(1 - \frac{(1-\rho+\rho \cdot p) \cdot \lambda}{p \cdot \rho}\right) - \gamma_{pub}\right)\right)}{\left((1-\rho \cdot p) \cdot \left(\frac{1-\lambda}{1-\rho} - \frac{\lambda}{p \cdot \rho}\right) - \left(\frac{(1-p) \cdot \rho}{1-\rho} \cdot \left(\frac{\rho \cdot p}{1-\rho+\rho \cdot p} - \lambda\right)\right)\right)^2} < 0.$$

This can be transformed into

$$\begin{split} & \left((1-\rho \cdot p) \left(-\frac{1}{1-\rho} - \frac{1}{\rho \cdot p} \right) + \frac{(1-p) \cdot \rho}{1-\rho} \right) \cdot \\ & \cdot \left(\frac{\rho \cdot p - (1-\rho+\rho \cdot p) \cdot \lambda}{(1-\rho) \cdot \rho \cdot p} \cdot \gamma_{pub} \cdot (1-\rho) + \left(\frac{\rho \cdot p - (1-\rho+\rho \cdot p) \cdot \lambda}{1-\rho+\rho \cdot p} \right) \cdot \left(w \cdot \left(1 - \frac{(1-\rho+\rho \cdot p) \cdot \lambda}{p \cdot \rho} \right) - \gamma_{pub} \right) \right) - \\ & - \left((1-\rho \cdot p) \cdot \frac{\rho \cdot p - (1-\rho+\rho \cdot p) \cdot \lambda}{(1-\rho) \cdot \rho \cdot p} - \left(\frac{(1-p) \cdot \rho}{1-\rho} \cdot \left(\frac{\rho \cdot p - (1-\rho+\rho \cdot p) \cdot \lambda}{1-\rho+\rho \cdot p} \right) \right) \right) \cdot \\ & \cdot \left(- \frac{(1-\rho) \cdot \gamma_{pub}}{\rho \cdot p} - 2 \cdot w \cdot \left(1 - \frac{(1-\rho+\rho \cdot p) \cdot \lambda}{p \cdot \rho} \right) \right) > 0, \end{split}$$

which finally yields

$$0 < w \cdot (p \cdot \rho - (1 - \rho + \rho \cdot p) \cdot \lambda) \text{ for } p \cdot \rho - (1 - \rho + \rho \cdot p) \cdot \lambda > 0 \text{ and}$$
$$0 > w \cdot (p \cdot \rho - (1 - \rho + \rho \cdot p) \cdot \lambda) \text{ for } p \cdot \rho - (1 - \rho + \rho \cdot p) \cdot \lambda < 0.$$

This is always true.

Further, it needs to be shown that the derivative of the lump-sum tax τ wrt. λ is larger than zero:

$$\frac{\partial \tau}{\partial \lambda} = \frac{\left(-(1-\rho+\rho\cdot p)\cdot\left(\frac{1}{1-\rho}+\frac{1}{\rho\cdot p}\right) + \left(\frac{(1-p)\cdot\rho}{1-\rho}\cdot\frac{(1-\rho+\rho\cdot p)}{1-\rho\cdot p}\right)\right)\cdot\left(w\cdot\frac{\rho\cdot p-(1-\rho+\rho\cdot p)\cdot\lambda}{p\cdot \rho} - \gamma_{pub} + \frac{(1-p)\cdot\rho\cdot\gamma_{pub}}{1-\rho\cdot p}\right)}{\left((1-\rho+\rho\cdot p)\cdot\left(\frac{1-\lambda}{1-\rho}-\frac{\lambda}{p\cdot\rho}\right) - \left(\frac{(1-p)\cdot\rho}{1-\rho}\cdot\frac{(\rho\cdot p-(1-\rho+\rho\cdot p)\cdot\lambda)}{1-\rho\cdot p}\right)\right)^2} - \frac{(1-\rho+\rho\cdot p)\cdot\left(\frac{1-\lambda}{1-\rho}-\frac{\lambda}{p\cdot\rho}\right)}{\left((1-\rho+\rho\cdot p)\cdot\left(\frac{1-\lambda}{1-\rho}-\frac{\lambda}{p\cdot\rho}\right) - \left(\frac{(1-\rho+\rho\cdot p)\cdot\lambda}{1-\rho\cdot p}\right)\right)^2}$$

$$-\frac{\left((1-\rho+\rho\cdot p)\cdot\left(\frac{1-\lambda}{1-\rho}-\frac{\lambda}{p\cdot\rho}\right)-\left(\frac{(1-p)\cdot\rho}{1-\rho}\cdot\frac{(\rho\cdot p-(1-\rho+\rho\cdot p)\cdot\lambda)}{1-\rho\cdot p}\right)\right)\cdot\left(-w\cdot\frac{(1-\rho+\rho\cdot p)}{\rho\cdot p}\right)}{\left((1-\rho+\rho\cdot p)\cdot\left(\frac{1-\lambda}{1-\rho}-\frac{\lambda}{p\cdot\rho}\right)-\left(\frac{(1-p)\cdot\rho}{1-\rho}\cdot\frac{(\rho\cdot p-(1-\rho+\rho\cdot p)\cdot\lambda)}{1-\rho\cdot p}\right)\right)^2}>0.$$

This can be simplified to

$$\begin{array}{l} \left(\left(1-\rho+\rho\cdot p\right) \cdot \left(1-\rho\cdot p\right) - \left(\left(1-p\right)\cdot\rho\right)\cdot\rho\cdot p\right) \cdot \frac{p\cdot\rho-(1-\rho+\rho\cdot p)\lambda}{\rho\cdot p\cdot(1-\rho\cdot p)}\cdot w > \\ > \frac{1-\rho}{(1-\rho\cdot p)} \cdot \left(w \cdot \frac{\rho\cdot p-(1-\rho+\rho\cdot p)\cdot\lambda}{p\cdot \rho} - \frac{1-\rho}{1-\rho\cdot p}\cdot\gamma_{pub} \right) \end{array}$$

or

$$0 > -\frac{1-\rho}{1-\rho \cdot p} \cdot \gamma_{pub},$$

which yields the desired result.

Proof of ii):

To show that redistribution is no longer complete, we need to prove that $w - \tau > v - \tau$ and v < w:

1. $w - \tau > v - \tau$.

This rewrites

$$w > v = \gamma_{pub} - \frac{1}{1-\rho} \cdot \left((1-p) \cdot \rho \cdot b - (1-\rho+\rho \cdot p) \cdot \tau \cdot (1-\lambda) \right)$$

or

$$w > \gamma_{pub} - \frac{1}{1-\rho} \cdot \left(1-p\right) \cdot \rho \cdot \frac{\left(\frac{1-\lambda}{1-\rho} - \frac{\lambda}{p\cdot\rho}\right) \cdot \gamma_{pub} \cdot (1-\rho) + \left(\frac{\rho\cdot p}{1-\rho+\rho\cdot p} - \lambda\right) \cdot \left(w \cdot \left(1 - \frac{(1-\rho+\rho\cdot p)\cdot \lambda}{p\cdot\rho}\right) - \gamma_{pub}\right)}{(1-\rho\cdot p) \cdot \left(\frac{1-\lambda}{1-\rho} - \frac{\lambda}{p\cdot\rho}\right) - \left(\frac{(1-p)\cdot \rho}{1-\rho} \cdot \left(\frac{\rho\cdot p}{1-\rho+\rho\cdot p} - \lambda\right)\right)} + \left(1-\rho + \frac{w \cdot \frac{\rho\cdot p - (1-\rho+\rho\cdot p)\cdot \lambda}{p\cdot\rho} - \gamma_{pub} + (1-p)\cdot \rho \cdot \frac{\gamma_{pub}}{1-\rho\cdot p}}{(1-\rho+\rho\cdot p) \cdot \left(\frac{1-\lambda}{1-\rho} - \frac{\lambda}{p\cdot\rho}\right) - \left(\frac{1}{1-\rho} \cdot (1-p)\cdot \rho \cdot \frac{(\rho\cdot p - (1-\rho+\rho\cdot p)\cdot \lambda)}{1-\rho\cdot p}\right)}{(1-\rho+\rho\cdot p) \cdot \left(\frac{1-\lambda}{1-\rho} - \frac{\lambda}{p\cdot\rho}\right) - \left(\frac{1}{1-\rho} \cdot (1-p)\cdot \rho \cdot \frac{(\rho\cdot p - (1-\rho+\rho\cdot p)\cdot \lambda)}{1-\rho\cdot p}\right)}{(1-\rho+\rho) \cdot \left(\frac{1-\lambda}{1-\rho} - \frac{\lambda}{p\cdot\rho}\right) - \left(\frac{1}{1-\rho} \cdot (1-p)\cdot \rho \cdot \frac{(\rho\cdot p - (1-\rho+\rho)\cdot \lambda)}{1-\rho\cdot p}\right)}{(1-\rho+\rho) \cdot \left(\frac{1-\lambda}{1-\rho} - \frac{\lambda}{p\cdot\rho}\right) - \left(\frac{1-\rho}{1-\rho} \cdot (1-\rho)\cdot \rho \cdot \frac{(\rho\cdot p - (1-\rho+\rho)\cdot \lambda)}{1-\rho\cdot p}\right)}{(1-\rho+\rho) \cdot \left(\frac{1-\lambda}{1-\rho} - \frac{\lambda}{p\cdot\rho}\right) - \left(\frac{1-\lambda}{1-\rho} \cdot \frac{$$

After some transformations this yields

$$p) \cdot \left(w \cdot \frac{\rho \cdot p - (1 - \rho + \rho \cdot p) \cdot \lambda}{p \cdot \rho} - \frac{1 - \rho}{1 - \rho \cdot p} \gamma_{pub}\right)$$

or

$$\gamma_{pub} \cdot (1-\rho) + w \cdot \rho \cdot p > w \cdot (1-\rho+\rho \cdot p) \cdot \lambda,$$

which yields the desired result.

2.
$$v - \tau = b$$
:
 $\gamma_{pub} - \frac{1}{1-\rho} \cdot \left((1-p) \cdot \rho \cdot b - (1-\rho+\rho \cdot p) \cdot \tau \cdot (1-\lambda) \right) - \tau = b$.

This rewrites

$$(1-\rho)\cdot\gamma_{pub}-((1-\rho+\rho\cdot p)\cdot\lambda-\rho\cdot p)\cdot\tau=(1-p\cdot\rho)\cdot b$$

or

$$(1-\rho) \cdot \gamma_{pub} - \left(\left(1-\rho+\rho \cdot p\right) \cdot \lambda - \rho \cdot p \right) \cdot \frac{w \cdot \frac{\rho \cdot p - (1-\rho+\rho \cdot p) \cdot \lambda}{p \cdot \rho} - \gamma_{pub} + (1-p) \cdot \rho \cdot \frac{\gamma_{pub}}{1-\rho \cdot p}}{(1-\rho+\rho \cdot p) \cdot \left(\frac{1-\lambda}{1-\rho} - \frac{\lambda}{p \cdot \rho}\right) - \left(\frac{1}{1-\rho} \cdot (1-p) \cdot \rho \cdot \frac{(\rho \cdot p - (1-\rho+\rho \cdot p) \cdot \lambda)}{1-\rho \cdot p}\right)}{(1-\rho \cdot p) \cdot \left(\frac{1-\lambda}{1-\rho} - \frac{\lambda}{p \cdot \rho}\right) - \left(\frac{(1-\rho) \cdot \rho \cdot (1-\rho+\rho \cdot p) \cdot \lambda}{p \cdot \rho} - \lambda\right)}{(1-\rho \cdot p) \cdot \left(\frac{1-\lambda}{1-\rho} - \frac{\lambda}{p \cdot \rho}\right) - \left(\frac{(1-p) \cdot \rho \cdot (1-\rho+\rho \cdot p) \cdot \lambda}{1-\rho + \rho \cdot p} - \lambda\right)}{(1-\rho \cdot p) \cdot \left(\frac{1-\lambda}{1-\rho} - \frac{\lambda}{p \cdot \rho}\right) - \left(\frac{(1-p) \cdot \rho \cdot (1-\rho+\rho \cdot p) \cdot \lambda}{1-\rho + \rho \cdot p} - \lambda\right)}{(1-\rho \cdot p) \cdot \left(\frac{1-\lambda}{1-\rho} - \frac{\lambda}{p \cdot \rho}\right) - \left(\frac{(1-p) \cdot \rho \cdot (1-\rho+\rho \cdot p) \cdot \lambda}{1-\rho + \rho \cdot p} - \lambda\right)} \right)$$

After some transformations this yields

$$(1-\rho) \cdot \gamma_{pub} \cdot \left((1-\rho+\rho \cdot p) \cdot \frac{\rho \cdot p - (1-\rho+\rho \cdot p) \cdot \lambda}{(1-\rho) \cdot p \cdot \rho} - \left(\frac{(1-p) \cdot \rho}{1-\rho} \cdot \frac{\rho \cdot p - (1-\rho+\rho \cdot p) \cdot \lambda}{1-\rho \cdot p} \right) \right) - ((1-\rho+\rho \cdot p) \cdot \lambda - \rho \cdot p) \cdot \left(w \cdot \frac{\rho \cdot p - (1-\rho+\rho \cdot p) \cdot \lambda}{p \cdot \rho} - \frac{(1-\rho) \cdot \gamma_{pub}}{1-\rho \cdot p} \right) =$$

$$= \left(1 - \rho + \rho \cdot p\right) \cdot \left(\frac{\rho \cdot p - (1 - \rho + \rho \cdot p) \cdot \lambda}{p \cdot \rho} \cdot \gamma_{pub} + \frac{\rho \cdot p - (1 - \rho + \rho \cdot p) \cdot \lambda}{1 - \rho + \rho \cdot p} \cdot \left(w \cdot \frac{\rho \cdot p - (1 - \rho + \rho \cdot p) \cdot \lambda}{p \cdot \rho} - \gamma_{pub}\right)\right)$$

or

$$\gamma_{pub} \cdot (1-\rho) \cdot \left(\frac{1-\rho \cdot p - 1 + \rho \cdot p}{(1-\rho \cdot p) \cdot p \cdot \rho}\right) = 0.$$

This is identical to

$$0 = 0,$$

which is always true.



Chapter 5

Bibliography

- Aghion P., Blanchard, O.J., 1994. On the speed of transition in Central Europe. Working Paper 4736, National Bureau of Economic Research, Cambridge.
- Arrelano M., Bond S., 1991. Some tests of specification for panel data: Monte Carlo evidence and an application to employment equations. Review of Economics Studies 58, 277-297.
- Arze del Granado F.J., Martinez-Vazquez J., McNab R., 2005. Fiscal decentralization and the functional composition of public expenditures. Working Paper 05-01, Andrew Young School of Policy Studies, Georgia State University, Atlanta.
- Balla K., Kertesi G., Köllö J., Simonovits A., 2004. Privatization, unemployment and subsidy for low-skilled labor. Mimeo, Hungarian Academy of Science.
- Baretti C., Huber B., Lichtblau K., 2002. A tax on tax revenue: The incentive effects of equalizing transfers: Evidence from Germany. International Tax and Public Finance 9, 631-649.
- Barnet S., 2000. Evidence on the fiscal and macroeconomic impact of pivatization. Working Paper 00/130, International Monetary Fund, Washington.
- Benassy-Quere A., Gobalraja N., Trannoy A., 2005. Tax competition and public input. European Network of Economic Policy Research Institutes, Working Paper

No 40/2005.

- Bergvall D., Charbit C., Kraan D.J., Merk O., (2006). Intergovernmental transfers and decentralised public spending. Working Paper 2006-3, Public Governance and Territorial Development Directorate, OECD, Paris.
- Besley T., Ghatak M., 2001. Government versus private ownership of public goods. Quarterly Journal of Economics 116, 1343-1372.
- Bond S., 2002. Dynamic panel data models: A guide to micro data methods and practice. Working paper CWP 09-02, Economic and Social Research Council, Swindon.
- Bond S., Hoeffler A., Temple J., 2001. GMM estimation of empirical growth models. Mimeo, University of Oxford, University of Bristol.
- Bordignon M., Manasse P., Tabellini G., 2001. Optimal regional redistribution under asymmetric information. American Economic Review 91, 709-723.
- Boycko M., Shleifer A., Vishny R.W., 1996. A theory of privatisation. The Economic Journal 106, 309-319.
- Brown J.D., Earle J., Telegdy A., 2005. Does privatization hurt workers? Lessons from comprehensive manufacturing firm panel data in Hungary, Romania, Russia, and Ukraine. Discussion Paper 2005/09, Centre for Economic Reform and Transformation, Heriot-Watt University, Edinburgh.
- Bruno G.S.F., 2005. Estimation and inference in dynamic unbalanced panel data models with a small number of individuals. Working Paper 165, Centre of Research on Innovation and Internationalization, Milan.
- Cremer H., Pestieau P., 1996. Distributive implications of european integration. European Economic Review 40, 747-757.

- Cremer H., Pestieau P., 1997. Income redistribution in an economic union: The trade off between inter- and intra-national redistribution. International Tax and Public Finance 4, 325-335.
- Dahlby B., Wilson S., 2003. Vertical fiscal externalities in a federation. Journal of Public Economics 87, 917-930.
- Dahlby B., 1996. Fiscal externalities and the design of intergovernmental grants. International Tax and Public Finance 3, 397-412.
- Devereux M.P., Griffith R., 1998. Taxes and the location of production: Evidence from a panel of US multinationals. Journal of Public Economics 68, 335-67.
- Devereux, M.P., Freeman H., 1995. The impact of tax on foreign direct investment: Empirical evidence and the implications for tax integration schemes. International Tax and Public Finance 2, 85-106.
- Eaton J., Rosen H.S., 1980. Optimal redistributive taxation and uncertainty. The Quarterly Journal of Ecnomics 95, 357-364.
- Erdmann Y., 1998. Development of social benefits and social policy in Poland, Hungary and the Slovak Republic since the system transformation. East European Quarterly 32, 301-314.
- Fenge R., Wrede M., 2004. EU regional policy: Vertical fiscal externalities and matching grants. Mimeo, ifo Institute Munich.
- Ford A., 1999. Local government grant distribution an international comparative study. Mimeo, Pricewaterhouse Coopers, London.
- Greene W.H., 2003. Econometric Analysis, 5th edition. International Edition, Prentice Hall, New York.
- Grossman S.J., Hart O.D., 1983. An analysis of the principal-agent problem. Econometrica 51, 7–46.

- Haskel J.J., Szymanski S., 1993. Privatization, liberalization, employment and wages: Theory and evidence for the UK. Economica 60, 161-182.
- Hauptmeier S., 2007. Intergovernmental grants and public input provision: Theory and evidence from Germany. Discussion Paper 07-006, Zentrum f
 ür Europ
 äische Wirtschaftsforschung, Mannheim.
- Hulten C.R., Schwab R.M., 1997. A fiscal federalism approach to infrastructure policy. Regional Science and Urban Economics 27, 139-159.
- Im K.S., Pesaran M.H., Shin Y., 1997. Testing for Unit Roots in Heterogeneous Panels. Mimeo, Department of Applied Economics, University of Cambridge.
- Jefferson, G.H., Jian S., 2006. Privatization and restructuring in China: Evidence from shareholding ownership, 1995-2001. Journal of Comparative Economics 34, 146-166.
- Jeronimo V., Pagan J.A., Soydemir G., 2000. Privatization and European Economic and Monetary Union. Eastern Economic Journal 26, 321-333.
- Judson R.A., Owen A.L., 1996. Estimating dynamic panel data models: A Practical Guide for Macroeconomists. Mimeo, Federal Reserve Board of Governors.
- Kanbur S.M., 1981. Risk taking and taxation An alternative perspective. Journal of Public Economics 15, 163-184.
- Kreider B., 2003. Income uncertainty and optimal redistribution. Southern Economic Journal 69, 718-725.
- Kappeler A., 2007. Fiscal externalities in a three-tier structure of government. Mimeo, Munich University (LMU).
- Kappeler A., Välilä T., 2008. Fiscal federalism and the composition of public investment in Europe. European Journal of Political Economy 24, 562–570.
- Keen M., Marchand M., 1997. Fiscal competition and the pattern of public spending. Journal of Public Economics 66, 33-53.

- Laffont, J.J., Tirole, J., 1993. A Theory of Incentives in Procurement and Regulation. MIT Press, Cambridge, MA.
- Levin A., Lin C.-F., Chu C-S.J., 2002. Unit root tests in panel data: asymptotic and finite-sample properties. Journal of Econometrics 108, 1-24.
- Martínez-López D., 2005. Fiscal federalism and public inputs provision vertical externalities matter. Working Paper Series 484, European Central Bank, Frankfurt.
- Matheson, T., 2005. Does fiscal redistribution discourage local public investment? Evidence from transitional Russia. Economics of Transition 13, 139-162.
- Matsumoto M., 1998. A note on tax competition and public input provision. Regional Science and Urban Economics 28, 456-473.
- McKenzie G. A., (1998). The macroeconomic impact of privatization. IMF Staff Papers, 45, 363-73.
- Musgrave R.A., 1959. The Theory of Public Finance, McGraw-Hill, New York.
- Oates W.E., 1972. Fiscal Federalism, Harcourut Brace Jovanovich, New York.
- OECD, 1999. Taxing powers of state and local government. OECD Tax Policy Studies 1, OECD, Paris.
- Persson T., Tabellini G., 1996. Federal fiscal constitutions: Risk sharing and redistribution. Journal of Political Economy 104, 979-1009.
- Rodrik D., 1998. Why do more open economies have bigger governments? The Journal of Political Economy 106, 997-1032.
- Seitz, H., Fester T., Kappeler A., Thum M., 2005. Öffentliche infrastruktur und kommunale finanzen. IFO-Dresden Studien 37, Ifo-Institute Dresden.
- Sheshinski E., López-Calva L.F., 2003. Privatization and its benefits: Theory and evidence. CESifo Economic Studies 49, 429-459.

- Shleifer A., Vishny R.W., 1994. Politicians and firms. The Quarterly Journal of Economics 109, 995-1025.
- Sinn H.W., 2003. The New Systems Competition, Blackwell, Oxford.
- Stegarescu D., 2005. Public sector decentralization: Measurement concepts and recent international trends. Fiscal Studies 26, 301–333.
- Tiebout C.M., 1956. A pure theory of local public expenditures. Journal of Political Economy 64, 416-424.
- Wildasin, D.E., 1997. Externalities and bailouts: Hard and soft budget constraints in intergovernmental fiscal relations. Policy Research Working Paper Series 1843, The World Bank, Washington.
- Wooldridge J.M., 2002. Econometric Analysis of Cross Section and Panel Data. MIT Press, Cambridge MA.
- Yannis K., Likoyanni E., 2002. Fiscal and other macroeconomic effects of privatization. Fondazione Eni Enrico Mattei.
- Zodrow G., Mieszkowski P., 1986. Pigou, Tiebout, property taxation, and the underprovision of local public goods. Journal of Urban Economics 19, 356-370.

Curriculum Vitae

09.2005 - 09.2008	PhD-Studies.
	Munich Graduate School of Economics, Munich, Germany.
02.2007 - 05.2007	Internship.
	European Investment Bank, Luxembourg.
03.2006	Visiting Researcher.
	Fudan University, School of Economics Shanghai, China.
09.2004 - 07.2005	MSc. in Economic Theory and Econometrics.
	Midi Pyrénées School of Economics, Toulouse, France.
09.2000 - 06.2004	Diploma in Economics.
	European-University Viadrina, Frankfurt-Oder, Germany.
01.2004 - 09.2004	Research Assistant.
	German Institute for Economic Research, Berlin, Germany.
10.2002 - 04.2003	Student exchange.
	Università degli Studi di Trento, Trento, Italy.
31.10.1978	Born in Oberstdorf, Germany.